UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 10-Q

(Mark One)

[X]

[]

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE

SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended March 31, 2010

or

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE

SECURITIES EXCHANGE ACT OF 1934

For the transition period from Commission File Number: 0-19989



Stratus Properties Inc.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

72-1211572

(I.R.S. Employer Identification No.)

98 San Jacinto Blvd., Suite 220
Austin, Texas
(Address of principal executive offices)

78701

(Zip Code)

(512) 478-5788

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. R Yes \Box No

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\S 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). \Box o Yes \Box o No

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer o□

reporting company R

Accelerated filer o□

Non-accelerated filer o□

Smaller

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). \Box o Yes R No

On April 30, 2010, there were issued and outstanding 7,462,617 shares of the registrant's common stock, par value \$0.01 per share.

STRATUS PROPERTIES INC. TABLE OF CONTENTS

Part I. Financial Information	Page 2
Item 1. Financial Statements:	
Consolidated Balance Sheets (Unaudited)	2
Consolidated Statements of Operations (Unaudited)	3
Consolidated Statements of Cash Flows (Unaudited)	4
Consolidated Statements of Equity (Unaudited)	5
Notes to Consolidated Financial Statements (Unaudited)	6
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	12
Item 4. Controls and Procedures	21
Part II. Other Information	21
Item 2. Unregistered Sales of Equity Securities and Use of Proceeds	21
Item 6. Exhibits	21
Signature	22
Exhibit Index	E-1

STRATUS PROPERTIES INC.

PART I. FINANCIAL INFORMATION

Item 1. Financial Statements.

STRATUS PROPERTIES INC. CONSOLIDATED BALANCE SHEETS (Unaudited) (In Thousands)

	March 31, 2010		De	cember 31, 2009
ASSETS				
Cash and cash equivalents	\$	5,908	\$	15,398
Real estate held for sale – developed or under development		137,242		124,801
Real estate held for sale – undeveloped		57,485		57,201
Real estate held for use, net		115,543		101,863
Investment in unconsolidated affiliate		3,315		3,391
Deferred tax assets		9,038		8,296
Other assets		21,133		17,640
Total assets	\$	349,664	\$	328,590
LIABILITIES AND EQUITY				
Accounts payable and accrued liabilities	\$	24.712	\$	16,247
Accrued interest and property taxes	Ψ	1,318	Ψ	3,401
Deposits		7,838		7,700
Debt		87,737		81,105
Other liabilities		1,705		2,224
Total liabilities		123,310		110,677
Commitments and contingencies				
Equity:				
Stratus stockholders' equity:				
Preferred stock		-		-
Common stock		83		83
Capital in excess of par value of common stock		197,277		197,333
Accumulated deficit		(37,708)		(35,999)
Common stock held in treasury		(17,972)		(17,941)
Total Stratus stockholders' equity		141,680		143,476
Noncontrolling interest in subsidiary		84,674		74,437
Total equity		226,354		217,913
Total liabilities and equity	\$	349,664	\$	328,590

The accompanying notes are an integral part of these consolidated financial statements.

STRATUS PROPERTIES INC. CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited) (In Thousands, Except Per Share Amounts)

Three Months Ended March 31,

		1,	
		2010	2009
Revenues:			<u> </u>
Real estate	\$	874	\$ 191
Rental income		1,297	1,173
Commissions, management fees and other		129	168
Total revenues		2,300	1,532
Cost of sales:			
Real estate, net		2,106	1,061
Rental		745	831
Depreciation		420	440
Total cost of sales		3,271	2,332
General and administrative expenses		1,832	2,079
Total costs and expenses		5,103	4,411
Operating loss		(2,803)	(2,879)
Interest income		14	246
Other income		228	=
Loss on interest rate cap agreement		(24)	(33)
Loss before income taxes and equity in unconsolidated affiliate's loss		(2,585)	(2,666)
Equity in unconsolidated affiliate's loss		(76)	(74)
Benefit from income taxes		881	897
Net loss		(1,780)	(1,843)
Net loss attributable to noncontrolling interest in subsidiary		71	106
Net loss attributable to Stratus common stock	\$	(1,709)	\$ (1,737)
Net loss per share attributable to Stratus common stock:			
Basic and diluted	\$	(0.23)	\$ (0.23)
Weighted average shares of common stock outstanding:			
Basic and diluted		7,456	7,446

The accompanying notes are an integral part of these consolidated financial statements.

STRATUS PROPERTIES INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (In Thousands)

Cash flow from operating activities: 2010 2009 Net loss \$ (1,780) \$ (1,843) Adjustments to reconcile net loss to net cash used in operating activities: Temperation 420 440 Loss on interest rate cap agreement 24 33 Cost of real estate sold 563 126 Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180) Net cash used in operating activities (15,886) (9,907)
Net loss \$ (1,780) \$ (1,843) Adjustments to reconcile net loss to net cash used in operating activities: Depreciation 420 440 Loss on interest rate cap agreement 24 33 Cost of real estate sold 563 126 Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Net loss \$ (1,780) \$ (1,843) Adjustments to reconcile net loss to net cash used in operating activities: Depreciation 420 440 Loss on interest rate cap agreement 24 33 Cost of real estate sold 563 126 Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
used in operating activities: 420 440 Loss on interest rate cap agreement 24 33 Cost of real estate sold 563 126 Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Depreciation 420 440 Loss on interest rate cap agreement 24 33 Cost of real estate sold 563 126 Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Loss on interest rate cap agreement 24 33 Cost of real estate sold 563 126 Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Cost of real estate sold 563 126 Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Deferred income taxes (898) (913) Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Stock-based compensation 146 171 Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Equity in unconsolidated affiliate's loss 76 74 Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Deposits (2,133) (22) Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Purchases and development of real estate properties (10,394) (8,454) Municipal utility district reimbursements - 3,387 Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Municipal utility district reimbursements-3,387Decrease in other assets878274Decrease in accounts payable, accrued liabilities and other(2,788)(3,180)
Decrease in other assets 878 274 Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Decrease in accounts payable, accrued liabilities and other (2,788) (3,180)
Net cash used in operating activities (15,886) (9,907)
Cash flow from investing activities:
Development of commercial leasing properties (10,305) (6,153)
Proceeds from matured U.S. treasury securities - 497
Investment in unconsolidated affiliate - (260)
Other - 21
Net cash used in investing activities (10,305) (5,895)
Cash flow from financing activities:
Borrowings from revolving credit facility 5,012 -
Payments on revolving credit facility (781)
Borrowings from project and term loans 4,500 4,735
Payments on project and term loans (2,099) (68)
Noncontrolling interest contributions 10,308 11,000
Net payments for stock-based awards (76) (101)
Purchases of Stratus common shares - (404)
Financing costs (163) -
Net cash provided by financing activities 16,701 15,162
Net decrease in cash and cash equivalents (9,490) (640)
Cash and cash equivalents at beginning of year 17,097
Cash and cash equivalents at end of period \$\\ 5,908\$ \$\\ 16,457\$

1e accompanying notes are an integral part of these consolidated financial statements.

STRATUS PROPERTIES INC. CONSOLIDATED STATEMENTS OF EQUITY (Unaudited) (In Thousands)

Stratus Stockholders' Equity

	Common Number of Shares	Stock At Par Value	Capital in Excess of Par Value	Accum- ulated Deficit	Commo Held in 7 Number of Shares	on Stock Treasury At Cost	Total Stratus Stockholders' Equity	Noncontrolling Interest in Subsidiary	Total Equity
Balance at December 31, 2009	8,315	\$ 83	\$ 197,333	\$ (35,999)	873	\$ (17,941)	\$ 143,476	\$ 74,437	\$217,913
Exercised and issued stock-based awards									
and other	25	-	(202)	-	-	-	(202)	-	(202)
Stock-based compensation	-	-	146	-	-	-	146	-	146
Tender of shares for stock-based awards	-	-	-	-	4	(31)	(31)	-	(31)
Noncontrolling interest contributions	-	-	-	-	-	-	-	10,308	10,308
Comprehensive income (loss):									
Net loss	-	-	-	(1,709)	-	-	(1,709)	(71)	(1,780)
Other comprehensive income	-	-	-		-	-			
Total comprehensive income (loss)				(1,709)			(1,709)	(71)	(1,780)
Balance at March 31, 2010	8,340	\$ 83	\$ 197,277	\$ (37,708)	877	\$ (17,972)	\$ 141,680	\$ 84,674	\$226,354

The accompanying notes are an integral part of these consolidated financial statements.

STRATUS PROPERTIES INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED)

1. GENERAL

The accompanying unaudited consolidated financial statements should be read in conjunction with the consolidated financial statements and notes thereto for the year ended December 31, 2009, included in Stratus Properties Inc.'s (Stratus) Annual Report on Form 10-K (Stratus 2009 Form 10-K) filed with the Securities and Exchange Commission (SEC). In the opinion of management, the accompanying consolidated financial statements reflect all adjustments (consisting only of normal recurring items) considered necessary for a fair statement of the financial position of Stratus at March 31, 2010, and the results of operations for the three-month periods ended March 31, 2010 and 2009, and cash flows for the three-month periods ended March 31, 2010 and 2009. Operating results for the three-month period ended March 31, 2010, are not necessarily indicative of the results that may be expected for the year ending December 31, 2010.

2. EARNINGS PER SHARE

Stratus' basic and diluted net loss per share of common stock was calculated by dividing the loss by the weighted average number of common shares outstanding during the period.

Stock options and restricted stock units representing approximately 133,400 shares for the first quarter of 2010 and approximately 156,800 shares for the first quarter of 2009 were excluded from weighted average common shares outstanding for purposes of calculating diluted net loss per share because they were anti-dilutive.

3. JOINT VENTURE WITH CANYON-JOHNSON URBAN FUND II, L.P.

Effective May 1, 2008, Stratus entered into a joint venture with Canyon-Johnson Urban Fund II, L.P. (Canyon-Johnson) for the development of a 36-story mixed-use development in downtown Austin, Texas, anchored by a W Hotel & Residences (the W Austin Hotel & Residences project). Stratus' initial capital contributions to the joint venture totaled \$31.8 million, which consisted of Stratus' purchase of a 1.76 acre tract of land located across the street from Austin City Hall, the related property and development agreements for the land and other project costs incurred by Stratus before May 1, 2008.

Stratus currently accounts for this joint venture as a variable interest entity (VIE) of which Stratus is the primary beneficiary. As a result, the assets, liabilities and results of operations of the joint venture are included in Stratus' consolidated financial statements.

Stratus is the manager of, and has an approximate 40 percent interest in, the joint venture. Canyon-Johnson has an approximate 60 percent interest in the joint venture. Decisions for the joint venture are made by unanimous vote of the partners. In the aggregate, Canyon-Johnson will contribute approximately 60 percent of the joint venture's required capital and Stratus will contribute approximately 40 percent. As of March 31, 2010, capital contributions totaled \$63.4 million for Stratus and \$85.5 million for Canyon-Johnson. The joint venture has a construction loan to finance project costs after the required capital contributions are made (see below).

On October 21, 2009, the joint venture obtained construction financing from Beal Bank Nevada (Beal Bank) (Beal Bank loan agreement). Pursuant to the Beal Bank loan agreement, the joint venture may borrow up to an aggregate of \$120 million to fund the construction, development and marketing costs of the W Austin Hotel & Residences project. An initial advance under the Beal Bank loan agreement of \$3.4 million was made at closing. The next advance is expected to occur in mid-2010 and thereafter advances are expected to be made monthly until the loan is fully funded. As a condition to further funding from the Beal Bank loan agreement, the joint venture must invest at least \$180 million in the project.

On April 6, 2010, Stratus and Canyon-Johnson amended the operating agreement for the W Austin Hotel & Residences project joint venture to provide that each partner will make an additional \$0.8 million contribution to the joint venture to fund construction costs of the W Austin Hotel & Residences project. These contributions were made by the partners in March 2010, after which capital contributions totaled \$63.4 million for Stratus and \$85.5 million for Canyon-Johnson. No further contributions are required to be made by Stratus or Canyon-Johnson. See Note 9 for further discussion of financing arrangements entered into subsequent to March 31, 2010.

On August 1, 2008, the joint venture paid \$0.7 million to enter into an agreement to cap the floating London Interbank Offered Rate (LIBOR) on the W Austin Hotel & Residences project construction loan at 4.5 percent (see Note 4). The LIBOR cap notional amount varies based on originally projected loan balances throughout the term of the loan. The agreement terminates on July 1, 2011.

A Stratus subsidiary has been designated as the developer of the W Austin Hotel & Residences project and will be paid a \$6.0 million developer's fee over the term of construction. Stratus received development fees totaling \$0.5 million in the first quarters of 2010 and 2009, which have been eliminated in consolidation.

Upon formation of the joint venture, Stratus performed an initial evaluation and concluded that the joint venture was a VIE and that Stratus was the primary beneficiary. Stratus reevaluated the VIE status and primary beneficiary of the joint venture as of the amendment to the operating agreement (March 31, 2010), and concluded that the joint venture is still a VIE, and Stratus is still the primary beneficiary as it is the developer of the project and contributed the land and development to the joint venture at formation. Stratus also reevaluated the primary beneficiary of the joint venture upon adoption of new consolidation guidance, effective January 1, 2010, (see Note 8) and concluded that Stratus is still the primary beneficiary. Accordingly, the W Austin Hotel & Residences project has been consolidated in Stratus' financial statements. Stratus will continue to periodically evaluate the primary beneficiary of this joint venture in accordance with the new consolidations guidance.

At March 31, 2010, Stratus' consolidated balance sheet includes \$187.6 million in total assets and \$34.7 million in total liabilities associated with the W Austin Hotel & Residences project. The assets associated with the W Austin Hotel & Residences project can only be used to settle obligations of the joint venture. The \$187.6 million of total assets included \$1.2 million of cash and cash equivalents, \$83.7 million of real estate held for sale – developed or under development, \$87.1 million of real estate held for use and \$15.6 million of other assets. The \$34.7 million of total liabilities included \$23.7 million of accounts payable and accrued liabilities, \$0.1 million of accrued interest and property taxes, \$7.5 million of deposits and \$3.4 million of debt.

Profits and losses between partners in a real estate venture should be allocated based on how changes in net assets of the venture would affect cash payments to the investors over the life of the venture and on its liquidation. The amount of the ultimate profits earned by the W Austin Hotel & Residences project will affect the ultimate profit sharing ratios because of provisions in the joint venture agreement which would require Stratus to return certain previously received distributions to Canyon-Johnson under certain circumstances. Accordingly, the W Austin Hotel & Residences project's cumulative profits or losses are allocated based on a hypothetical liquidation of the venture's net assets as of each balance sheet date because of the uncertainty of the ultimate profits and, therefore, profit-sharing ratios. At March 31, 2010, the cumulative losses for the W Austin Hotel & Residences project were allocated based on 43 percent for Stratus and 57 percent for Canyon-Johnson.

4. FAIR VALUE MEASUREMENTS

Fair value accounting guidance includes a fair value hierarchy that is intended to increase consistency and comparability in fair value measurements and related disclosures. The fair value hierarchy is based on inputs to valuation techniques that are used to measure fair value that are either observable or unobservable. Observable inputs reflect assumptions market participants would use in pricing an asset or liability based on market data obtained from independent sources while unobservable inputs reflect a reporting entity's pricing based upon their own market assumptions.

The fair value hierarchy consists of the following three levels:

Level 1 – Inputs are quoted prices in active markets for identical assets or liabilities.

Level 2 – Inputs are quoted prices for similar assets or liabilities in an active market, quoted prices for identical or similar assets or liabilities in markets that are not active, inputs other than quoted prices that are observable and market-corroborated inputs which are derived principally from or corroborated by observable market data.

Level 3 – Inputs are derived from valuation techniques in which one or more significant inputs or value drivers are unobservable.

The following table sets forth Stratus' financial assets measured at fair value on a recurring basis as of March 31, 2010, by level within the fair value hierarchy (in thousands):

	Total Fair Value Measurement March 31, 2010	Quoted Prices in Active Markets for Identical Assets (Level 1)	 Significant Other Observable Inputs (Level 2)	 Significant Unobservable Inputs (Level 3)
Cash equivalents	\$ 175	\$ 175	\$ -	\$ -
Interest rate cap				
agreement	1	 	 1	
	\$ 176	\$ 175	\$ 1	\$ _

Valuation Techniques

Cash Equivalents. Stratus has certificates of deposits with maturity dates less than 90 days, which are considered cash equivalents. Stratus' cash equivalent instruments are classified within Level 1 of the fair value hierarchy because they are valued using quoted market prices in active markets.

Interest Rate Cap Agreement. On August 1, 2008, Stratus' joint venture with Canyon-Johnson entered into an agreement to cap the floating LIBOR rate on its W Austin Hotel & Residences project construction loan at 4.5 percent through July 1, 2011, to manage interest rate risk (see Note 3). Stratus uses an interest rate pricing model that relies on market observable inputs such as LIBOR to measure the fair value of the interest rate cap agreement. Stratus also evaluated the counterparty credit risk associated with the interest rate cap agreement, which is considered a Level 3 input, but did not consider such risk to be significant. Therefore, the interest rate cap agreement is classified within Level 2 of the fair value hierarchy. Stratus recorded non-cash charges totaling \$24 thousand in the first quarter of 2010 and \$33 thousand in the first quarter of 2009 related to decreases in fair value of the interest rate cap agreement.

Summarized below are the carrying values and estimated fair values of financial assets and liabilities (in thousands).

	 March 31, 2010				December 31, 2009		
	arrying Value		Fair Value		Carrying Value		Fair Value
Cash and cash equivalents ^a	\$ 5,908	\$	5,908	\$	15,398	\$	15,398
Accounts and notes receivable ^a	706		706		1,734		1,734
Interest rate cap agreementb	1		1		25		25
Accounts payable, accrued liabilities, accrued interest and							
property taxesa	26,030		26,030		19,648		19,648
Debt ^c	87,737		86,789		81,105		78,571

- a. Fair value approximates the carrying amounts because of the short-term nature of these instruments.
- b. Recorded at fair value. Observable inputs, such as LIBOR, are used to determine fair value (see above).
- c. Generally recorded at cost. Fair value of substantially all of Stratus' debt is estimated based on discounted future expected cash flows at estimated current interest rates. The fair value of debt does not represent the amounts that will ultimately be paid upon the maturities of the loans.

5. INTEREST CAPITALIZATION

Stratus capitalized all of its interest costs totaling \$1.6 million in the first quarter of 2010 and \$1.3 million in the first quarter of 2009.

6. INCOME TAXES

Companies are required to determine an estimated annual effective tax rate to apply to their interim pre-tax income or loss, and the estimated annual effective rate is required to be revised, if necessary, to reflect the company's best current estimate as of the end of each successive interim period during the year. If a reliable estimate cannot be made, the actual effective tax rate for the year-to-date period may be the best estimate of the annual effective tax rate.

During 2008, Stratus concluded that estimating a consistent annual effective tax rate was increasingly difficult because of the uncertainty in forecasting its taxable income or loss since such amounts are primarily dependent upon asset sales which are difficult to predict with reasonable certainty and may vary significantly from period to period. Additionally, the ability to forecast is increasingly difficult in light of the current economic environment. Stratus believes that such uncertainty goes beyond normal market variations and forecasting an annual effective rate would not provide a meaningful estimate. As such, Stratus believes that the actual year-to-date effective tax rate is the best estimate of the annual tax rate. Stratus' benefit from income taxes has been calculated utilizing its actual effective tax rate for the three-month period ended March 31, 2010.

The difference between Stratus' consolidated effective income tax rates for the first quarter of 2010 and 2009 and the U.S. federal statutory rate of 35 percent was primarily attributable to state income tax expense and other permanent items.

In its ongoing assessment of the realizability of deferred tax assets, management considers whether it is more likely than not that some portion or all of the deferred tax assets will not be realized. Based on the expectation of future taxable income and that deductible temporary differences will offset existing taxable temporary differences, management believes it is more likely than not that the benefits of these deductible differences, net of the existing valuation allowances, are realizable at March 31, 2010. Such determination may change in the future based on numerous factors, including the impact of the overall economic environment on Stratus' financial results.

7. BUSINESS SEGMENTS

Stratus has two operating segments, "Real Estate Operations" and "Commercial Leasing." The Real Estate Operations segment is comprised of all Stratus' real estate held for sale (developed, under development and undeveloped) in Austin, Texas, which consist of its properties in the Barton Creek community, the Circle C community and Lantana, and certain portions of the W Austin Hotel & Residences project. For definitions of these property classifications, see "Overview" located in Items 1 and 2 "Business and Properties" of the Stratus 2009 Form 10-K.

The Commercial Leasing segment primarily includes the two office buildings at 7500 Rialto Boulevard. In addition, the commercial leasing segment includes a retail building and a bank building in Barton Creek Village, two retail buildings in the Circle C Community, and certain portions of the W Austin Hotel & Residences project.

Stratus uses operating income or loss to measure the performance of each segment. Stratus allocates general and administrative expenses between the segments based on projected annual revenues for each segment. Stratus also allocates the W Austin Hotel & Residences capital expenditures and assets between the segments based on projected cost of construction for each segment. Accordingly, the following segment information reflects management's determinations that may not be indicative of what actual financial performance of each segment would be if it were an independent entity.

Segment data presented below were prepared on the same basis as Stratus' consolidated financial statements.

_	Real Estate Operations ^a	Commercial Leasing		Other	Total
		(In Tho	usands	5)	
Three Months Ended March 31, 2010					
Revenues \$	1,003 \$	1,297	\$	-	\$ 2,300
Cost of sales, excluding depreciation	(2,106)	(745)		-	(2,851)
Depreciation	(55)	(365)		-	(420)
General and administrative expenses	(1,040)	(792)		<u>-</u>	 (1,832)
Operating loss \$\square\$	(2,198) \$	(605)	\$		\$ (2,803)
Capital expenditures \$	10,394 \$	10,305	\$	-	\$ 20,669
Total assets at March 31, 2010 \$	209,701 \$	130,925	\$	9,038ь	\$ 349,664
Three Months Ended March 31, 2009					
Revenues \$	359 \$	1,173	\$	-	\$ 1,532
Cost of sales, excluding depreciation	(1,061)	(831)		-	(1,892)
Depreciation	(67)	(373)		-	(440)
General and administrative expenses	(1,296)	(783)			 (2,079)
Operating loss \$	(2,065) \$	(814)	\$		\$ (2,879)
Capital expenditures \$	8,454 \$	6,153	\$	-	\$ 14,607
Total assets at March 31, 2009 \$	175,301 \$	79,635	\$	8,634ь	\$ 263,570

- a. Includes sales commissions, management fees and other revenues together with related expenses.
- b. Primarily includes deferred tax assets.

8. NEW ACCOUNTING STANDARD

Consolidations. In May 2009, the Financial Accounting Standards Board (FASB) issued accounting guidance to replace the quantitative-based risks and rewards calculation for determining which enterprise, if any, has a controlling financial interest in a VIE with an approach focused on identifying which enterprise has the power to direct the activities of a VIE that most significantly impact the entity's economic performance and (1) the obligation to absorb expected losses of the entity or (2) the right to receive expected residual returns from the entity. It also requires ongoing assessments of whether an enterprise is the primary beneficiary of a VIE. Additionally, this guidance amends the consideration of related party relationships in the determination of the primary beneficiary of a VIE by providing, among other things, an exception with respect to de facto agency relationships in certain circumstances. This guidance is effective for fiscal years and interim periods beginning after November 15, 2009. Stratus' adoption of this guidance effective January 1, 2010, did not have a significant impact on its financial reporting and disclosures.

9. SUBSEQUENT EVENTS

On April 6, 2010, Stratus and Canyon-Johnson amended the operating agreement for the W Austin Hotel & Residences project joint venture to provide that each partner will make an additional \$0.8 million contribution to the joint venture to fund construction costs of the W Austin Hotel & Residences project. These contributions were made by the partners in March 2010, after which capital contributions totaled \$63.4 million for Stratus and \$85.5 million for Canyon-Johnson. No further contributions are required to be made by Stratus or Canyon-Johnson.

On April 6, 2010, Stratus and Canyon-Johnson entered into a \$30 million loan agreement with Hunter's Glen/Ford Investments I LLC (the Ford loan agreement) effective as of March 31, 2010, secured by a second lien on the W Austin Hotel & Residences project assets. Amounts borrowed under the Ford loan agreement will bear interest at an annual rate equal to 17.5 percent. Interest will accrue and can either be paid annually or added to the principal. The outstanding principal and accrued unpaid interest are due at maturity on March 31, 2012. Optional prepayments made after the first anniversary are not subject to prepayment premiums or fees. In addition, after one year from the first borrowing, the lender, with permission from Beal Bank, may require prepayment, but solely from the proceeds from the sale of W Austin Hotel & Residences residential units. Stratus has guaranteed payment of principal and interest

under the loan and completion of the project in connection with this loan agreement. In addition, the Ford loan agreement contains a covenant requiring that Stratus maintain a minimum total stockholders' equity balance of \$120 million.

Additionally, the Ford loan agreement provides for a profits interest in the joint venture. The profits interest provides that Ford will receive 95 percent of the operating cash flow and net proceeds from capital events of the joint venture up to a maximum payment of \$750,000 if paid on the first anniversary date of the Ford loan agreement and increased each full or partial month thereafter by \$62,500 until the Ford loan is paid in full.

An initial advance under the Ford loan agreement of \$10 million was made at closing, and an additional advance of \$11 million was made on April 19, 2010. Loan proceeds will continue to be disbursed to pay costs of the W Austin Hotel & Residences project before any additional loan proceeds from Beal Bank will be advanced

On April 7, 2010, Stratus extended and modified its credit facility with Comerica, effective as of March 31, 2010, such that the existing \$45 million facility was replaced with a \$35 million revolving loan and a \$10 million term loan. Any amounts repaid under the \$10 million term loan are not available for future advance to Stratus. The applicable interest rate for the revolving loan is LIBOR plus 4 percent, with a minimum rate of 6 percent, and the applicable interest rate for the term loan is LIBOR plus 5 percent, with a minimum rate of 7 percent. The outstanding principal from both loans are due at maturity on May 30, 2012.

Interest payments are due monthly on amounts outstanding under the \$35 million revolving loan. The \$10 million term loan will require monthly interest only payments for the first year, and quarterly principal payments of \$0.5 million beginning on June 1, 2011, in addition to the monthly interest payments. In addition, any distributions received by Stratus from its investment in the W Austin Hotel & Residences project shall, after repayment of any amounts due under the Beal Bank and Ford loan agreements, be paid to Comerica and applied against the \$10 million term loan to the extent of any outstanding amounts. The revised Comerica revolving loan and term loan agreements also increase Stratus' minimum net worth covenant from \$80 million to \$120 million.

On April 7, 2010, Stratus extended and modified its seven unsecured term loans with First American Asset Management (FAAM) effective as of March 31, 2010. Stratus repaid \$2.0 million on March 19, 2010, and the remaining maturities are \$2.0 million in June 2010, \$9.0 million in December 2011, \$3.5 million in December 2012, \$15.0 million in December 2013 and \$8.5 million in December 2014. The applicable interest rate for all seven unsecured term loans is 8.75 percent. In addition, the debt service coverage ratio covenant contained in the loan agreements was modified such that Stratus will remain compliant with the covenant so long as Stratus maintains total stockholders' equity of no less than \$120 million. The modified loan agreements prohibit common stock repurchases while any of the loans are outstanding. The modified loan agreements for three of the loans totaling \$18.5 million maturing in 2012 and 2013 prohibit prepayment before December 31, 2010. From January 1, 2011, until one year prior to the maturity dates, Stratus may prepay the loans, subject to applicable prepayment penalties. Beginning one year prior to the maturity dates, Stratus may prepay the loans with no prepayment penalties. The modified loan agreements for two of the loans totaling \$8.5 million maturing in 2014 prohibit prepayment before December 31, 2011. From January 1, 2012 to December 31, 2013, Stratus may prepay the loans, subject to applicable prepayment penalties. Beginning January 1, 2014, Stratus may prepay the loans with no prepayment penalties.

Stratus evaluated events after March 31, 2010, and through the date the financial statements were issued, and determined that any events or transactions occurring during this period that would require recognition or disclosure are appropriately reflected in Stratus' financial statements and the notes thereto.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations.

OVERVIEW

Management's discussion and analysis presented below should be read in conjunction with our discussion and analysis of financial results contained in our 2009 Annual Report on Form 10-K (2009 Form 10-K) filed with the Securities and Exchange Commission (SEC). The operating results summarized in this report are not necessarily indicative of our future operating results. All subsequent references to "Notes" refer to Notes to Consolidated Financial Statements (unaudited), unless otherwise stated.

We are engaged in the acquisition, development, management, operation and sale of commercial, multi-family and residential real estate properties located primarily in the Austin, Texas area. We primarily generate revenues from sales of developed properties and through rental income from our commercial properties. Developed property sales can include an individual tract of land that has been developed and permitted for residential use or a developed lot with a home already built on it. We may, on occasion, sell properties under development or undeveloped properties, if opportunities arise that we believe will maximize overall asset values.

Our principal real estate holdings are in southwest Austin, Texas. The number of developed lots, developed or under development acreage and undeveloped acreage as of March 31, 2010, that comprise our principal development projects are presented in the following table.

		Acreage									
	•	D	eveloped or	Under Development			Undeveloped				
	Developed Lots	Single Family	Multi- Family	Commercial	Total	Single Family	Commercial	Total	Total Acreage		
<u>Austin</u>											
Barton Creek	122	119	249	368	736	749	28	777	1,513		
Lantana	-	-	-	-	-	-	223	223	223		
Circle C	29	-	-	35	35	148	352	500	535		
W Austin Hotel & Residences	-	-	-	2ª	2	-	-	-	2		
San Antonio											
Camino Real					<u>-</u>		2	2	2		
Total	151	119	249	405	773	897	605	1,502	2,275		

a. Represents a city block in downtown Austin planned for a mixture of hotel, residential, retail, office and entertainment uses.

Our other Austin holdings at March 31, 2010, consisted of two 75,000-square-foot office buildings at 7500 Rialto Boulevard (7500 Rialto) located in our Lantana development, a 22,000-square-foot retail complex representing phase one of Barton Creek Village and two retail buildings totaling 21,000 square feet at the 5700 Slaughter project in the Circle C community.

The continued softness in the real estate market, among other factors, significantly impacted our consolidated financial results. In the first quarter of 2010, our revenues totaled \$2.3 million and our net loss attributable to common stock totaled \$1.7 million, compared with revenues of \$1.5 million and a net loss attributable to common stock of \$1.7 million for the first quarter of 2009. Our financial condition and results of operations are highly dependent upon market conditions for real estate activity in Austin, Texas. Our future operating cash flows and, ultimately, our ability to develop our properties and expand our business will be largely dependent on the level of our real estate sales. In turn, these sales will be significantly affected by future real estate market conditions in Austin, Texas, including development costs, interest rate levels, the availability of credit to finance real estate transactions, demand for residential and commercial real estate, and regulatory factors including our land use and development entitlements.

Current economic conditions have also resulted in a general decline in leasing activity across the United States (U.S.), and have caused vacancy rates to increase in most markets, including Austin, Texas. Investment sales activity in the U.S. declined sharply during 2008 because of, among other factors, limited availability and increased cost of financing, especially the absence of securitized debt, which was the source of heightened investment activity, and the resulting gap between buyer and seller expectations of value.

Periods of economic slowdown or recession, rising interest rates, tightening of the credit markets, declining demand for or increased supply of real estate, or the public perception that any of these events may occur can adversely affect our business. These conditions could result in a general decline in rents, which in turn would reduce revenue from leases. In addition, these conditions could lead to a decline in property values as well as a decline in funds invested in commercial real estate and related assets, which in turn may reduce revenues from leases and development fees.

U.S. credit markets have yet to fully recover, and this continuing issue is impacting the broader U.S. economy. Commercial real estate lenders have substantially tightened underwriting standards or have withdrawn from the lending market, materially impacting liquidity in the real estate debt markets, making financing terms for owners of retail properties less attractive, and in certain cases resulting in the unavailability of certain types of debt financing. Tighter lending standards and higher borrowing costs have exerted downward pressure on the value and liquidity of real estate assets which will impact the values we could obtain from the sale of our properties. These factors may make it more difficult for us to sell properties or may adversely affect the price we receive for properties that we do sell, as prospective buyers may experience increased costs of debt financing or difficulties in obtaining such financing. Our future performance will, in part, be dependent upon the recovery of the credit markets and the underlying strength of the U.S. economy. Given the current business climate in which we are operating and the numerous uncertainties related to our business, including the rate of sales, sales prices, and mortgage constraints, it is difficult to project operating and financial results for 2010 and later years.

BUSINESS STRATEGY

We continue to focus on our near-term goal of developing our properties and projects in a difficult economic climate and our long-term goal of maximizing the value of our development communities. We believe that Austin, Texas, continues to be a desirable market and many of our developments are in locations that are unique and where approvals and entitlements, which we have already obtained, are increasingly difficult to secure. Real estate development in southwest Austin historically has been constrained as a result of various restrictions imposed by the City of Austin (the City) and several special interest groups have also traditionally opposed development in the area where most of our property is located. We believe that many of our developments have inherent value given their unique nature and location and that this value should be sustainable in the future.

Our long-term success will depend on our ability to maximize the value of our real estate through obtaining additional required approvals that permit us to develop and sell our properties in a timely manner at a reasonable cost. In addition, we continue to pursue additional development opportunities, and currently believe we can obtain financing necessary for developing our properties, although our ability to obtain financing in the future, as well as the cost of such financing, may be negatively impacted by current U.S. economic conditions. See "Risk Factors" located in Item 1A of our 2009 Form 10-K.

DEVELOPMENT AND OTHER ACTIVITIES

W Austin Hotel & Residences. In 2005, the City selected our proposal to develop a mixed-use project in downtown Austin immediately north of the new City Hall complex. The W Austin Hotel & Residences project includes an entire city block and is planned for a mixture of hotel, residential, retail, office and entertainment uses. In 2006, we acquired the property for \$15.1 million. We have executed agreements with Starwood Hotels & Resorts Worldwide, Inc. for the development of a W Hotel & Residences on the site. Effective May 1, 2008, we entered into a joint venture with Canyon-Johnson Urban Fund II, L.P. (Canyon-Johnson) for the development of the W Austin Hotel & Residences project (see Note 3). Construction of the \$300 million project commenced in the second quarter of 2008 and is proceeding as scheduled. We anticipate the hotel will open in the fourth quarter of 2010. Condominium residences will be completed on a floor-by-floor basis. We anticipate that completion of the condominium residences will

begin in the fourth quarter of 2010 and continue through mid-2011. As of May 4, 2010, we had 79 of the 159 condominium residences under contract. The sales contracts are generally secured with buyer deposits of 10 percent of the purchase price. We currently consolidate the joint venture with Canyon-Johnson because the project is considered a variable interest entity (VIE) and we are considered the primary beneficiary. If it is determined that the W Austin Hotel & Residences is no longer a VIE or that we are no longer the primary beneficiary of the joint venture, the project will be deconsolidated from our financial statements.

For a discussion of the financing structure for the W Austin Hotel & Residences project see Note 3.

Crestview Station. In 2005, we formed a joint venture with Trammell Crow to acquire an approximate 74-acre tract at the intersection of Airport Boulevard and Lamar Boulevard in Austin, Texas, for \$7.7 million. The property, known as Crestview Station, is a single-family, multi-family, retail and office development, which is located on the site of a commuter rail line. With Trammell Crow, we have completed environmental remediation, which the State of Texas certified as complete in 2007, and permitting of the property. The initial phase of utility and roadway infrastructure is complete. Crestview Station sold substantially all of its multi-family and commercial properties in 2007 and one commercial site in the first quarter of 2008. The joint venture retained the single-family component of Crestview Station and one commercial site. The joint venture is currently processing permits to develop Crestview Station as a 450-unit transit-oriented neighborhood. At March 31, 2010, our investment in the Crestview Station project totaled \$3.3 million and the joint venture partnership had \$8.2 million of outstanding debt, of which we guarantee \$1.4 million. A reserve for interest and property taxes through May 2011 has been established with the lender. Scheduled principal payments begin in June 2011, and the loan matures in May 2012. We account for our 50 percent interest in the Crestview Station joint venture under the equity method.

Residential. As of March 31, 2010, the number of our residential developed lots, lots under development and potential development by area are shown below (excluding lots and units associated with our Canyon-Johnson and Crestview Station joint ventures):

	Residential Lots							
	Developed	Under Development	Potential Development a	Total				
Barton Creek:								
Calera:								
Calera Court Courtyard Homes	2	-	-	2				
Calera Drive	8	-	-	8				
Verano Drive	67	-	-	67				
Amarra Drive:								
Phase I Lots	7	-	-	7				
Phase II Lots	35	-	-	35				
Townhomes	-	-	221	221				
Phase III	-	89	-	89				
Mirador Estate	2	-	-	2				
Wimberly Lane Phase II	1	-	-	1_				
Section N Multi-family	-	-	1,860	1,860				
Other Barton Creek Sections	-	-	154	154				
Circle C:								
Meridian	29		57	86				
Total Residential Lots	151	89	2,292	2,532				

a. Our development of the properties identified under the heading "Potential Development" is dependent upon the approval of our development plans and permits by governmental agencies, including the City. Those governmental agencies may either not approve one or more development plans and permit applications related to such properties or require us to modify our development plans. Accordingly, our development strategy with respect to those properties may change in the future. While we may be proceeding with approved infrastructure projects on some of these properties, they are not considered to be "under development" for disclosure in this table unless other development activities necessary to fully realize the properties' intended final use are in progress or scheduled to commence in the near term.

Calera. Calera is a residential subdivision with plat approval for 155 lots. During 2004, we began construction of 16 courtyard homes at Calera Court, the 16-acre initial phase of the Calera subdivision. The second phase of Calera, Calera Drive, consisting of 53 single-family lots, many of which adjoin the Fazio Canyons Golf Course, received final plat and construction permit approval in 2005. Construction of

the final phase, known as Verano Drive, was completed in July 2008 and includes 71 single-family lots. As of March 31, 2010, two courtyard homes at Calera Court, eight lots at Calera Drive and 67 lots at Verano Drive remained unsold. In April 2010, we sold one of the courtyard homes at Calera Court.

Amarra Drive. Amarra Drive Phase I, which is the initial phase of the Amarra Drive subdivision, was completed in 2007 and includes eight lots with sizes ranging from approximately one to four acres, some of which are course-side lots on the Fazio Canyons Golf Course and others are secluded lots adjacent to the Nature Conservancy of Texas. As of March 31, 2010, seven Amarra Drive Phase I lots remain unsold. In 2008, we commenced development of Amarra Drive Phase II, which consists of 35 lots on 51 acres. Development was substantially completed in October 2008, but no sales have occurred.

Mirador Estate. The Mirador subdivision consists of 34 estate lots, with each lot averaging approximately 3.5 acres in size. As of March 31, 2010, two Mirador estate lots remained unsold

Wimberly Lane. Wimberly Lane included two phases, with phase one consisting of 75 residential lots and phase two consisting of 47 residential lots. We entered into a contract with a national homebuilder to sell 41 lots within the Wimberly Lane Phase II subdivision. We sold the last homebuilder lot in January 2008, and have one Wimberly Lane lot remaining for sale.

<u>Circle C.</u> We are developing the Circle C community based on the entitlements secured in our Circle C settlement with the City. Our Circle C settlement, as amended in 2004, permits development of 1.16 million square feet of commercial space, 504 multi-family units and 830 single family residential lots. Meridian is an 800-lot residential development at the Circle C community. Development of Meridian included our contracts with three national homebuilders to complete the construction and sales of 494 lots. We sold the final 13 lots for \$0.9 million in the first quarter of 2010.

In 2006, we signed another contract with a national homebuilder for 42 additional lots. Development of those lots was substantially completed in April 2008. In June 2009, the contract was terminated by the homebuilder. As of the date the contract was terminated, there were 30 remaining lots. In connection with the termination, the homebuilder forfeited a deposit of \$0.6 million, which we recorded as other income in the second quarter of 2009. We are currently pursuing contracts with other homebuilders for the sale of the remaining lots. One lot was sold in August 2009 for \$0.1 million and 29 lots remained unsold as of March 31, 2010. The final phase of Meridian is expected to consist of 57 one-acre lots.

Commercial. As of March 31, 2010, the number of square feet of our commercial property developed, under development and our potential development are shown below (excluding property associated with our Canyon-Johnson and Crestview Station joint ventures):

	Developed	Under Development	Potential Development a	Total
Barton Creek:				
Barton Creek Village Phase I	22,000	-	-	22,000
Barton Creek Village Phase II	-	-	18,000	18,000
Entry Corner	-	-	5,000	5,000
Amarra Retail/Office	-	-	90,000	90,000
Section N	-	-	1,500,000	1,500,000
Circle C:				
Chase Ground Lease	4,000	-	-	4,000
5700 Slaughter	21,000	_	_	21,000
Parkside Village	-	80,000	-	80,000
Tract 110	-	_	760,000	760,000
Tract 101	-	-	90,000	90,000
Tract 102	-	-	25,000	25,000
Tract 114	-	-	5,000	5,000
Lantana:				
7500 Rialto	150,000	-	-	150,000
Advanced Micro Devices				
Option Tract	-	-	100,000	100,000
Tract G06	-	-	400,000	400,000
Tract GR1	-	-	325,000	325,000
Tract G05	-	-	260,000	260,000
Tract G07	-	-	210,000	210,000
Tract CS5	-	-	175,000	175,000
Tract CS1-CS3	-	-	150,000	150,000
Tract LR1	-	-	75,000	75,000
Tract L04	-	-	70,000	70,000
Austin 290 Tract	-	-	20,000	20,000
Total Square Feet	197,000	80,000	4,278,000	4,555,000

a. Our development of the properties identified under the heading "Potential Development" is dependent upon the approval of our development plans and permits by governmental agencies, including the City. Those governmental agencies may either not approve one or more development plans and permit applications related to such properties or require us to modify our development plans. Accordingly, our development strategy with respect to those properties may change in the future. While we may be proceeding with approved infrastructure projects on some of these properties, they are not considered to be "under development" for disclosure in this table unless other development activities necessary to fully realize the properties' intended final use are in progress or scheduled to commence in the near term.

Barton Creek. The first phase of the Barton Creek Village includes a 22,000-square-foot retail complex and a 3,300-square-foot bank building within this retail complex. As of March 31, 2010, the retail complex was 81 percent leased and the bank building is leased through January 2023.

<u>Circle C.</u> During the third quarter of 2008, we completed the construction of two retail buildings, totaling 21,000 square feet, at the 5700 Slaughter project. This retail project also includes a 4,000-square-foot bank building on an existing ground lease. As of March 31, 2010, occupancy was approximately 91 percent for the two retail buildings.

The Circle C community also includes Parkside Village, an 80,000-square-foot planned retail project. The project will be developed in two phases. The first phase will consist of a 34,000-square-foot building to accommodate a full-service restaurant and theater. The second phase will consist of three tilt-wall retail buildings at 14,775 square feet, 8,075 square feet and 7,600 square feet, and two pads available for ground leases. We are pursuing final permits and entitlements to position the project for commencement of construction when appropriate.

Lantana. Lantana is a partially developed, mixed-use real-estate development project. Lantana includes two 75,000-square-foot office buildings at 7500 Rialto. As of March 31, 2010, occupancy was 97 percent for the original office building and 94 percent for the second office building. As of March 31, 2010, we had

remaining entitlements for approximately 1.0 million square feet of office and retail use on 223 acres. Regional utility and road infrastructure is in place with capacity to serve Lantana at full build-out permitted under our existing entitlements. Lantana also includes 100,000 square feet of potential development for commercial property under an option contract with Advance Micro Devices, Inc. (NYSE: AMD), which expired unexercised in April 2010.

RESULTS OF OPERATIONS

We are continually evaluating the development potential of our properties and will continue to consider opportunities to enter into transactions involving our properties. As a result, and because of numerous other factors affecting our business activities as described herein, our past operating results are not necessarily indicative of our future results.

Summary operating results follow (in thousands):

		First C	uarter	
		2010		2009
Revenues:				
Real estate operations	\$	1,003	\$	359
Commercial leasing		1,297		1,173
Total revenues	<u>\$</u>	2,300	\$	1,532
Operating loss	\$	(2,803)	\$	(2,879)
Benefit from income taxes	\$	881	\$	897
Net loss attributable to Stratus common stock	\$	(1,709)	\$	(1,737)

We have two operating segments, "Real Estate Operations" and "Commercial Leasing" (see Note 7). The following is a discussion of our operating results by segment.

Real Estate Operations

Summary operating results for real estate operations follow (in thousands):

		First Quarter			
	2	2010		2009	
Revenues:					
Developed property sales	\$	874	\$	191	
Commissions, management fees and other		129		168	
Total revenues		1,003		359	
Cost of sales, including depreciation		(2,161)		(1,128)	
General and administrative expenses		(1,040)		(1,296)	
Operating loss	\$	(2,198)	\$	(2,065)	

Developed Property Sales. Property sales totaled \$0.9 million, which included 13 lots from the Meridian development in the Circle C community, for first-quarter 2010 and \$0.2 million, which included three lots in the Meridian development, for first-quarter 2009. The increase in developed property sales revenues resulted from a higher number of lots sold at Meridian primarily related to the completion of sales under homebuilder contracts in first-quarter 2010.

Cost of Sales. Cost of sales includes cost of property sold, project operating and marketing expenses and allocated overhead costs, partly offset by reductions for certain municipal utility district reimbursements. Cost of sales totaled \$2.2 million for first-quarter 2010 and \$1.1 million for first-quarter 2009. Cost of sales for first-quarter 2010 increased compared to first-quarter 2009 primarily because of an increase in developed property sales in 2010. We sold 13 lots in first-quarter 2010 at an average cost of \$44,000 per lot, compared to three lots in first-quarter 2009 at an average cost of \$43,000 per lot. Cost of sales for our

real estate operations also include significant, recurring costs (including property taxes, maintenance and marketing), which totaled \$1.6 million in first-quarter 2010 and \$1.0 million in first-quarter 2009 and do not vary significantly with the number of property sales.

We are anticipating continued lower levels of lot sales in the next several quarters because of the continued weakness in the U.S. and Austin real estate markets.

General and Administrative Expenses. Consolidated general and administrative expenses decreased to \$1.8 million for first-quarter 2010 from \$2.1 million for first-quarter 2009, primarily because of lower business development costs. General and administrative expenses allocated to real estate operations decreased to \$1.0 million for first-quarter 2010 from \$1.3 million for first-quarter 2009, primarily as a result of lower projected real estate operations revenues as a percentage of total projected revenues in 2010. For more information about the allocation of general and administrative expenses to our operating segments, see Note 7.

Commercial Leasing

Summary commercial leasing operating results follow (in thousands):

	First Quarter			
	2010			2009
Rental income	\$	1,297	\$	1,173
Rental property costs		(745)		(831)
Depreciation		(365)		(373)
General and administrative expenses		(792)		(783)
Operating loss	\$	(605)	\$	(814)

Rental Income. Rental income increased in first-quarter 2010 compared to first-quarter 2009, primarily because of a \$0.1 million increase in rental income at 5700 Slaughter, which was in the initial leasing stage in 2009.

Rental Property Costs. Rental property costs decreased to \$0.7 million for first-quarter 2010 from \$0.8 million for first-quarter 2009. The decrease in rental property costs in first-quarter 2010 is primarily the result of a decrease in property management expenses.

Non-Operating Results

Interest Income. Interest income totaled less than \$0.1 million in the first quarter of 2010 and \$0.2 million in the first quarter of 2009. The decrease in interest income primarily reflects a decrease in Barton Creek Municipal Utility District (MUD) reimbursements and lower cash balances in first-quarter 2010. Interest income included interest on Barton Creek MUD reimbursements totaling \$0.2 million in first-quarter 2009. There were no Barton Creek MUD reimbursements in first-quarter 2010.

Other Income, net. We recorded other income of \$0.2 million in first-quarter 2010, which primarily reflects a reimbursement of deferred financing costs for extinguished debt.

Loss on Interest Rate Cap Agreement. We recognized a loss on the interest rate cap agreement of less than \$0.1 million in the first quarters of 2010 and 2009. The interest rate cap agreement relates to the W Austin Hotel & Residences project construction loan (see Note 3).

Equity in Unconsolidated Affiliate's Loss. We account for our 50 percent interest in our unconsolidated affiliate, Crestview Station, using the equity method. Crestview Station sold substantially all of its multi-family and commercial properties in 2007 and one commercial site in the first quarter of 2008. Our equity in Crestview Station's losses totaled \$0.1 million for the first quarters of 2010 and 2009, primarily reflecting operating losses recognized by Crestview Station in both periods, because there were no sales.

Benefit from Income Taxes. We recorded an income tax benefit of \$0.9 million for the first quarters of 2010 and 2009. The difference between our consolidated effective income tax rates for the first quarters of 2010 and 2009 and the U.S. federal statutory rate of 35 percent was primarily attributable to state income tax expense and other permanent items.

Net Loss Attributable to Noncontrolling Interest in Subsidiary. Net loss attributable to nonconrolling interest in subsidiary totaled \$0.1 million in the first quarters of 2010 and 2009 related to the W Austin Hotel & Residences project (see Note 3).

CAPITAL RESOURCES AND LIQUIDITY

At March 31, 2010, we had \$5.9 million in cash and cash equivalents. We also had \$16.3 million outstanding and approximately \$25.7 million in availability under our revolving credit facility at March 31, 2010. Subsequent to March 31, 2010, we concluded several financing transactions, including an extension and modification of our revolving credit facility with Comerica (see "Credit Facility and Other Financing Arrangements" and Note 9).

Comparison of Three-Months 2010 and 2009 Cash Flows

Cash used in operating activities increased to \$15.9 million during the first three months of 2010, compared with \$9.9 million during the first three months of 2009, primarily because of a \$3.4 million decrease in MUD reimbursements, a \$2.1 million increase in deposits and a \$1.9 million increase in cash used in development of real estate properties, partly offset by a \$0.7 million increase in proceeds from developed property sales. As stated previously, the continued weakness in the U.S. real estate market has negatively affected sales of lots, and we expect this trend to continue in the near-term. Expenditures for purchases and development of real estate properties for the first three months of 2010 and 2009 included development costs for properties held for sale, including the residential portion of the W Austin Hotel & Residences project (\$9.6 million in 2010 and \$6.5 million in 2009), and the Barton Creek, Lantana and Circle C communities. Capital expenditures for the W Austin Hotel & Residences project, including both residential and commercial leasing expenditures, are expected to approximate \$100 million for the remainder of 2010 and will be funded with borrowings under two loan agreements (see "Credit Facility and Other Financing Arrangements").

Cash used in investing activities totaled \$10.3 million during the first three months of 2010 and \$5.9 million during the first three months of 2009. Commercial leasing development expenditures for the first three months of 2010 and 2009 included development costs for the W Austin Hotel & Residences project totaling \$10.0 million in 2010 and \$6.1 million in 2009. We received proceeds from matured U.S. treasury securities of \$0.5 million in the first three months of 2009. We also contributed capital of \$0.3 million to Crestview Station in the first three months of 2009.

Cash provided by financing activities totaled \$16.7 million in the first three months of 2010 and \$15.2 million in the first three months of 2009. Noncontrolling interest contributions from Canyon-Johnson for the W Austin Hotel & Residences project totaled \$10.3 million in the first three months of 2010 and \$11.0 million in the first three months of 2009. In the first three months of 2010, net borrowings from our revolving credit facility totaled \$4.2 million and borrowings from the 5700 Slaughter term loan totaled \$4.5 million, partly offset by financing costs of \$0.2 million. Debt repayments on project and term loans totaled \$2.1 million in 2010. Borrowings from the Barton Creek Village term loan totaled \$4.7 million in the first three months of 2009. See "Credit Facility and Other Financing Arrangements" for a discussion of our outstanding debt at March 31, 2010. We used \$0.4 million in the first three months of 2009 to repurchase shares of our common stock on the open market.

In 2001, our Board of Directors approved an open market share purchase program for up to 0.7 million shares of our common stock. During first-quarter 2010, there were no purchases under this program. A total of 161,145 shares remain available under this program as of March 31, 2010. Our modified unsecured term loans prohibit common stock purchases while any of the loans are outstanding.

Credit Facility and Other Financing Arrangements

At March 31, 2010, we had total debt of \$87.7 million, compared with \$81.1 million at December 31, 2009. Our debt outstanding at March 31, 2010 consisted of the following:

• \$16.3 million of borrowings outstanding and \$2.9 million of letters of credit issued under our \$45.0 million revolving credit facility with Comerica, resulting in availability of approximately \$25.7 million. We used the proceeds from these borrowings for general corporate purposes, including overhead and development costs. The revolving credit facility is secured by assets at Barton Creek, Lantana and Circle C.

- \$38.0 million of borrowings outstanding under seven unsecured term loans, which include two \$5.0 million loans, an \$8.0 million loan, a \$7.0 million loan, a \$6.0 million loan and two \$3.5 million loans.
- \$20.9 million of borrowings outstanding under the Lantana promissory note, which matures in January 2018 and is secured by our buildings at 7500 Rialto Boulevard.
- \$4.6 million of borrowings outstanding under a term loan, which matures in April 2014 and is secured by Barton Creek Village.
- \$3.4 million of borrowings outstanding under the Beal Bank loan, which matures in October 2014 and is secured by the assets in the W Austin Hotel & Residences project.
- \$4.5 million of borrowings under a \$5.4 million term loan, which matures in January 2015 and is secured by 5700 Slaughter.

Subsequent to March 31, 2010, we concluded the following transactions:

- On April 6, 2010, we and Canyon-Johnson entered into the \$30 million Ford loan agreement secured by a second lien on the W Austin Hotel & Residences project assets. The Ford loan agreement matures in March 2012 and the applicable interest rate is 17.5 percent. Additionally, the Ford loan agreement provides for a profits interest in our joint venture with Canyon-Johnson (see Note 9).
- On April 7, 2010, we extended and modified our credit facility with Comerica such that the existing \$45 million facility was replaced with a \$35 million revolving loan and a \$10 million term loan. The applicable interest rate for the revolving loan is the London Interbank Offered Rate (LIBOR) plus 4 percent, with a minimum rate of 6 percent, and the applicable interest rate for the term loan is LIBOR plus 5 percent, with a minimum rate of 7 percent. Both loans mature in May 2012.
- On April 7, we extended and modified our seven unsecured term loans. We repaid \$2.0 million on March 19, 2010, and the remaining maturities are \$2.0 million in June 2010, \$9.0 million in December 2011, \$3.5 million in December 2012, \$15.0 million in December 2013 and \$8.5 million in December 2014. The applicable interest rate for all seven unsecured term loans is 8.75 percent.

See Note 9 for further discussion of debt agreements entered into subsequent to March 31, 2010.

NEW ACCOUNTING STANDARD

Refer to Note 8 for discussion of a new accounting standard.

CAUTIONARY STATEMENT

Our discussion and analysis contains forward-looking statements regarding future reimbursements for infrastructure costs, future events related to financing and regulatory matters, projected capital expenditures, the expected results of our business strategy, and other plans and objectives of management for future operations and activities. Important factors that could cause actual results to differ materially from our expectations include economic and business conditions, business opportunities that may be presented to and pursued by us, changes in laws or regulations and other factors, many of which are beyond our control, and other factors that are described in more detail under "Risk Factors" located in Item 1A of our 2009 Form 10-K.

Item 4. Controls and Procedures.

- (a) Evaluation of disclosure controls and procedures. Our chief executive officer and chief financial officer, with the participation of management, have evaluated the effectiveness of our "disclosure controls and procedures" (as defined in Rules 13a-15(e) and 15(d)-15(e) under the Securities Exchange Act of 1934) and determined that our controls and procedures were effective as of the end of March 31, 2010.
- (b) <u>Changes in internal control</u>. There was no change in our internal control over financial reporting that occurred during the quarter ended March 31, 2010, that has materially affected, or is reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

Item 2. <u>Unregistered Sales of Equity Securities and Use of Proceeds.</u>

The following table sets forth shares of our common stock we repurchased during the three months ended March 31, 2010.

Period	(a) Total Number of Shares Purchased	(b) Average Price Paid Per Share	(c) Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs ^a		(d) Maximum Number of Shares That May Yet Be Purchased Under the Plans or Programs ^a
January 1 to 31, 2010	-	_		-	161,145
February 1 to 28, 2010	-	-		-	161,145
March 1 to 31, 2010	_ _	-			161,145
Total	<u>-</u>	-			

a. In February 2001, our Board of Directors approved an open market share purchase program for up to 0.7 million shares of our common stock. The program does not have an expiration date. Our modified unsecured term loans prohibit common stock purchases while any of the loans are outstanding.

Item 6. Exhibits.

The exhibits to this report are listed in the Exhibit Index beginning on page E-1 hereof.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

STRATUS PROPERTIES INC.

By: /s/ Erin D. Pickens

Erin D. Pickens Senior Vice President and Chief Financial Officer (authorized signatory and Principal Financial Officer)

Date: May 17, 2010

STRATUS PROPERTIES INC. EXHIBIT INDEX

			Incorporated by Reference		
Exhibit Number	Exhibit Title	Filed with this Form 10-Q	Form	File No.	Date Filed
3.1	Amended and Restated Certificate of Incorporation of Stratus.	101m 10 Q	10-Q	000-19989	05/17/2004
3.2	Certificate of Amendment to the Amended and Restated Certificate of Incorporation of Stratus, dated May 14, 1998.		10-Q	000-19989	05/17/2004
3.3	Certificate of Amendment to the Amended and Restated Certificate of Incorporation of Stratus, dated May 25, 2001.		10-K	000-19989	03/22/2002
3.4	By-laws of Stratus, as amended as of November 6, 2007.		10-Q	000-19989	08/11/2008
<u>10.1</u>	Loan Agreement by and between CJUF II Stratus Block 21 LLC and Hunter's Glen/Ford Investments I LLC effective as of March 31, 2010.	X			
10.2	Promissory Note by and between CJUF II Stratus Block 21 LLC and Hunter's Glen/Ford Investments I LLC effective as of March 31, 2010.	X			
10.3	Profits Interest Agreement by and between CJUF II Stratus Block 21 LLC and Hunter's Glen/Ford Investments I LLC effective as of March 31, 2010.	X			
10.4	Fourth Modification and Extension Agreement by and between Stratus Properties Inc., Stratus Properties Operating Co., L.P., Circle C Land, L.P., Austin 290 Properties, Inc., Calera Court, L.P. and Comerica Bank effective as of March 31, 2010.	X			
10.5	Loan Modification Agreement by and between Stratus Properties Inc. and American Strategic Income Portfolio Inc. effective as of March 31, 2010.	X			
<u>10.6</u>	Loan Modification Agreement by and between Stratus Properties Inc. and American Strategic Income Portfolio IncII effective as of March 31, 2010.	X			
<u>10.7</u>	Loan Modification Agreement by and between Stratus Properties Inc. and American Strategic Income Portfolio IncII effective as of March 31, 2010.	X			
10.8	Loan Modification Agreement by and between Stratus Properties Inc. and American Strategic Income Portfolio IncIII effective as of March 31, 2010.	X			
10.9	Loan Modification Agreement by and between Stratus Properties Inc. and American Strategic Income Portfolio IncIII effective as of March 31, 2010.	X			
10.10	Loan Modification Agreement by and between Stratus Properties Inc. and American Select Portfolio Inc. effective as of March 31, 2010.	X			

			Incorporated by Reference		
Exhibit Number	Exhibit Title	Filed with this Form 10-Q	Form	File No.	Date Filed
10.11	Loan Modification Agreement by and between Stratus Properties Inc. and American Select Portfolio Inc. effective as of March 31, 2010.	X			
10.12*	Change of Control Agreement between Stratus Properties Inc. and William H. Armstrong III, effective as of March 9, 2010.		8-K	000-19989	03/12/2010
10.13*	Change of Control Agreement between Stratus Properties Inc. and Erin D. Pickens, effective as of March 9, 2010.		8-K	000-19989	03/12/2010
31.1	Certification of Principal Executive Officer pursuant to Rule 13a-14(a)/15d-14(a).	X			
31.2	Certification of Principal Financial Officer pursuant to Rule 13a-14(a)/15d-14(a).	X			
32.1	Certification of Principal Executive Officer pursuant to 18 U.S.C. Section 1350.	X			
<u>32.2</u>	Certification of Principal Financial Officer pursuant to 18 U.S.C. Section 1350.	X			

extain instruments with respect to long-term debt of Stratus have not been filed as exhibits to this Quarterly Report on Form 10-Q since the total amount of a authorized under any such instrument does not exceed 10 percent of the total assets of Stratus and its subsidiaries on a consolidated basis. Stratus agrees to copy of each such instrument upon request of the Securities and Exchange Commission.

es management contract or compensatory plan or arrangement.

LOAN AGREEMENT MADE BY AND BETWEEN CJUF II STRATUS BLOCK 21 LLC

AND

HUNTER'S GLEN/FORD INVESTMENTS I LLC, as Lender

Dated as of March 31, 2010

TABLE OF CONTENTS

		Page
	ARTICLE 1 INCORPORATION OF RECITALS AND EXHIBITS	2
1.1	Incorporation of Recitals.	2
1.2	Incorporation of Exhibits.	2
	ADTICLE A DEED ITIONS	2
2.1	ARTICLE 2 DEFINITIONS Defined Terms.	2 2
2.1	Other Definitional Provisions.	14
2,2	Office Definitional Flovisions.	14
	CLE 3 BORROWER'S REPRESENTATIONS AND WARRANTIES	14
3.1	Representations and Warranties.	14
3.2	Survival of Representations and Warranties.	19
	ARTICLE 4 LOAN AND LOAN DOCUMENTS	20
4.1	Funding of the Loan.	20
4.2	Loan Documents.	20
4.3	Term of the Loan, Extensions.	21
4.4	Prepayments.	22
4.5	Required Principal Payments.	22
4.6	Receipt of Payments.	22
4.7	Protective Advances.	22
4.8	Lender's Inability to Fund.	22
	ARTICLE 5 INTEREST	23
5.1	Interest Rate.	23
	ARTICLE 6 COSTS OF MAINTAINING LOAN	23
6.1 Borr	ower Withholding.	23
0.1 Bon	over withholding.	23
	ARTICLE 7 LOAN EXPENSE AND ADVANCES	23
7.1	Loan and Administration Expenses.	23
7.2	Loan Fees.	24
7.3	[reserved]	24
7.4	[reserved]	24
7.5	Lender's Attorneys' Fees and Disbursements.	24
7.6	Time of Payment of Fees and Expenses.	24
7.7	Expenses and Advances Secured by Loan Documents.	24
7.8	Right of Lender to Make Advances to Cure Borrower's Defaults.	24
	ARTICLE 8 NON-CONSTRUCTION REQUIREMENTS PRECEDENT	25
8.1	Non-Construction Conditions Precedent.	25
	ARTICLE 9 CONSTRUCTION REQUIREMENTS PRECEDENT	30
9.1	Construction Documents Required as of Closing.	30
9.2	Construction Deliveries Required as of the Full Loan Opening.	32
7.∠	Constituent on Derivenes required as of the Pull Loan Opening.	32
	ARTICLE 10 BUDGET, CONTINGENCY FUND AND CHANGE ORDERS	33
10.1	Budget.	33
10.2	Budget Line Items.	33
10.3	[reserved]	34
	(i)	

10.4 10.5	[reserved] Change Orders.	34 34
	ARTICLE 11 [reserved]	34
	ARTICLE 12 [reserved]	34
	ARTICLE 13 [reserved]	34
141	ARTICLE 14 SALE OF RESIDENTIAL UNITS	34
14.1 14.2	Price List Schedule. Sales Agreements.	34 34
14.2	Purchaser Deposits.	35
14.3	Residential Unit Sales.	36
14.5	Sales Operations and Seller's Obligations; Amendment and Termination of Sales Agreements.	37
14.6	Delivery of Sales Information and Documents.	37
14.7	Borrower's Acknowledgment Regarding Buyer Financing.	37
14.8	Condominium Regime.	37
14.9	Release of Residential Units.	42
14.10	Application of Sales Proceeds.	43
	ARTICLE 15 OTHER COVENANTS	44
15.1	Borrower further covenants and agrees as follows:	44
15.2	Single Purpose Entity Covenants.	54
15.3	Authorized Representative.	56
15.4	Negative Covenants.	56
	ARTICLE 16 CASUALTIES AND CONDEMNATION	57
	nder's Election to Apply Proceeds on Indebtedness.	57
16.2	Borrower's Obligation to Rebuild and Use of Proceeds Therefor.	58
	ARTICLE 17 ASSIGNMENTS BY LENDER AND BORROWER	58
17.1	Assignments and Participations.	58
17.2	Prohibition of Assignments and Transfers by Borrower.	58
17.3	Prohibition of Transfers in Violation of ERISA.	59
17.4	Successors and Assigns.	60
	ARTICLE 18 TIME OF THE ESSENCE	60
18.1	Time is of the Essence.	60
	ARTICLE 19 EVENTS OF DEFAULT	60
19.1	Events of Default.	60
	ARTICLE 20 LENDER'S REMEDIES IN EVENT OF DEFAULT	62
20.1	Remedies Conferred Upon Lender.	62
	ADDICUE AL CENERAL PROVICIONO	
21.1	ARTICLE 21 GENERAL PROVISIONS	63
21.1 21.2	Captions. Modification; Waiver.	63 63
21.2	Governing Law.	63
	quiescence Not to Constitute Waiver of Lender's Requirements.	63
21.5	Disclaimer by Lender.	63

(ii)

	Partial Invalidity; Severability. Definitions Include Amendments. Execution in Counterparts. Entire Agreement. Reserved. èrences to Senior Loan Agreement and Concurrent Lenders Approval. Jurisdiction. Reserved. Binding Effect. Waiver of Accord and Satisfaction.	64 64 65 65 65 65 65 66 66
	ARTICLE 22 NOTICES	66
	ARTICLE 23 WAIVER OF JURY TRIAL	68
EXHIBITS TO L	OAN AGREEMENT	
Exhibit A Exhibit B Exhibit C Exhibit D Exhibit D Exhibit F Exhibit F Exhibit G Exhibit H Exhibit I Exhibit I Exhibit I Exhibit I Exhibit L Exhibit L Exhibit M Exhibit M-1 Exhibit M-2 Exhibit N Exhibit O Exhibit O Exhibit C	Legal Description of Land Construction Schedule Permitted Exceptions No Litigation Certificate Insurance Requirements During and Post Construction Architect's Certificate Budget [reserved] Approved Plans and Specifications and Approved Finish Standards Subcontracts [reserved] [reserved] [reserved] List of Sales Agreements List of Sales Agreements List of Sales Agreements Discrepancy Items Price List Schedule [reserved] Approved Form of Sales Agreement [reserved] Leasing Parameters and Allowable Tenant Improvements [reserved] Form of Fourth Estoppel and Agreement from City of Austin Due Diligence Materials	
	(iii)	

LOAN AGREEMENT

Project Commonly Known a

"W Hotel and Residences" Block 21, Austin, Texas

THIS LOAN AGREEMENT ("Agreement") is made as of March 31, 2010, by and between CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company ("Borrower"), and HUNTER'S GLEN/FORD INVESTMENTS I LLC, a Texas, its successors and assigns ("Lender").

WITNESSETH:

RECITALS

A. Borrower is the owner in fee simple of an approximately 76,176 square foot parcel of land commonly known as "Block 21," bounded by Second, Third, Guadalupe and Lavaca Streets, City of Austin, County of Travis, State of Texas, and legally described in Exhibit A attached hereto (the "Land"). Borrower proposes to construct on the Land a mixed use project to be known as the "W Hotel and Residences," consisting of a building of thirty-six (36) stories (the "Building") and other facilities containing: (i) one hundred fifty-nine (159) residential condominium units (each, a "Residential Unit") on twenty (20) floors, from floor 18 through floor 37 of the Building, containing at least 272,272 Saleable Square Feet (with each capitalized term used and not defined in these Recitals being defined hereinbelow) of interior space and with interior finished ceiling heights of at least ten (10) feet (outside of areas containing mechanical runs), (ii) a "W" flagged hotel with two hundred fifty-two (22) guest rooms, situated on ten (10) floors, from floor 6 through floor 16 of the Building, to be furnished and managed pursuant to the Hotel Operating Agreement (as hereinafter defined), and containing at least 100,408 square feet of interior room space and 88,212 square feet of hotel operating space, collectively with, on floor 2 through floor 4 of the Building, 9,583 square feet of meeting space, a 8,060 square foot fitness facility, a 9,935 square foot pool and pool deck, and a business center (collectively, the "Hotel"), (iii) on floor 1 through floor 3 of the Building, 18,341 net Rentable Square Feet of retail space (the "Retail Space") and restaurant space (provided that such restaurant space will not consist of more than 15,900 net Rentable Square Feet exclusive of any outdoor space and Hotel kitchen space) (the "Restaurant Space") and 37,382 net Rentable Square Feet of office space (the "Office Space"), (iv) a live performance venue, on the top three (3) floors of an attached 4-story structure, containing at least 86,750 sq

1

- B. Borrower has entered into that certain Amended and Restated Construction Loan Agreement (the "Senior Loan Agreement") dated October 21, 2009, between Borrower and Beal Bank Nevada ("Senior Lender"), pursuant to which Senior Lender committed to make construction advances to construct the Project (hereinafter defined) in an amount not to exceed \$120,000,000 (the "Senior Loan"), said Senior Loan being evidenced by that certain Amended and Restated Promissory Note (the "Senior Note") dated October 21, 2009, payable to the order of Senior Lender, in the original principal amount of \$120,000,000, and secured by that certain Amended and Restated Construction Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing ("Senior Deed of Trust") dated October 21, 2009, executed by Borrower for the benefit of Senior Lender, and recorded as Instrument No. 2009176430 in the Official Real Property Records of Trust County, Texas. The Senior Note, the Senior Loan Agreement, the Senior Deed of Trust and the other documents executed in connection with the Senior Loan are collectively referred to as the "Senior Loan Documents".
- C. Borrower has requested that Lender, and Lender has agreed, to extend a loan to Borrower in the aggregate amount of \$30,000,000 (the "Loan") to provide additional funds for the construction of the Project, which shall in all respects be subordinate and inferior to the Senior Loan and the Senior Loan Documents. In that regard, Lender, Senior Lender and Borrower have entered into that certain Subordination and Standstill Agreement (the "Intercreditor Agreement") dated the same date as this Agreement, pursuant to which the parties thereto have made certain agreements with respect to the respective rights and remedies of Senior Lender and Lender.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties agree as follows:

ARTICLE 1

INCORPORATION OF RECITALS AND EXHIBITS

1.1 Incorporation of Recitals.

The foregoing preambles and all other recitals set forth herein are made a part hereof by this reference.

1.2 Incorporation of Exhibits.

Exhibits A through T to this Agreement, attached hereto, are incorporated in this Agreement and expressly made a part hereof by this reference.

ARTICLE 2

DEFINITIONS

2.1 Defined Terms.

The following terms as used herein shall have the following meanings:

Affiliate: With respect to a specified person or entity, any individual, partnership, corporation, limited liability company, trust, unincorporated organization, association or other entity that, directly or indirectly, through one or more intermediaries, controls or is controlled by or is under common control with such person or entity, including, without limitation, any general or limited partnership in which such person or entity is a partner.

Agreement: This Loan Agreement.

Applicable Condominium Laws: As such term is defined in Section 14.8(c).

Appraisal: An MAI certified appraisal of the Project performed in accordance Lender's appraisal requirements by an appraiser selected and retained by Lender.

Approved Finish Standards: As such term is defined in Section 9.1(f).

Approved Lease: As such term is defined in Section 15.1(m).

Approved Plans and Specifications: As such term is defined in Section 9.1(f).

Architect: BOKA Powell, L.L.C.

Architect's Certificate: A certificate in the form of Exhibit F attached hereto executed by the Architect in favor of Lender.

Associations: As such term is defined in Section 14.8(d).

Authorized Representative: William H. Armstrong, an individual, or such other individual that is designated in accordance with Section 15.3 of this Agreement.

Bankruptcy Code: Title 11 of the United States Code entitled "Bankruptcy" as now or hereafter in effect, or any successor thereto or any other present or future bankruptcy or insolvency statute.

Borrower: As such term is defined in the opening paragraph of this Agreement.

Budget: The budget for the Project specifying all costs and expenses of every kind and nature whatsoever to be incurred by Borrower in connection with the Project (including, the Loan) prior to the Maturity Date, as approved by Lender as set forth in Section 10.1.

Budget Line Item: As such term is defined in Section 10.2.

Building: As such term is defined in Recital A.

Business Day: Any Monday through Friday, excluding days on which Lender is closed for business.

Change Order: Shall mean any of the following: (i) a request for changes in the Approved Plans and Specifications (other than minor field changes involving no extra cost) or for a change to the General Contract Price, (ii) an amendment to the General Contract, (iii) a construction change directive or (iv) a written order for a minor change in the work issued by the Architect.

City Documents: Declaration of Restrictive Covenants (as amended, the "Declaration") dated as of December 15, 2006, by the City of Austin, a Texas home rule city and municipal corporation, recorded in the Official Public Records of Travis County, Texas as Document No. 2006240877; and Special Warranty Deed (as amended, the "Deed") dated as of December 15, 2006, by the City, in favor of Stratus Block 21 Investments, L.P., a Texas limited partnership, recorded in the Official Public Records of Travis County, Texas as Document No. 2006240878, as both were amended by Estoppel Certificate and Agreement dated April 26, 2008 and recorded in the Official Public Records of Travis County, Texas under Document No. 2008078527, by Estoppel Certificate and Agreement dated May 13, 2008 and recorded in the Official Public Records of Travis County, Texas under Document No. 2008085863, and by Estoppel Certificate and Agreement dated October 19, 2009 and recorded in the Official Public Records of Travis County, Texas under Document No. 2009176432.

CJUF: CJUF Block 21 Member, LLC., a Delaware limited liability company.

Closing: The date of the Closing Funding.

Closing Funding: The first disbursement of Loan proceeds into Borrower's account with Senior Lender in accordance with Section 4.1, which shall be advanced on or about the date hereof.

Collateral Assignment of Hotel Documents: That certain Assignment of Hotel Documents collaterally assigning Borrower's interests in the Hotel Documents (and related documents) to Lender.

Commercial Space: Together, the Office Space, the Restaurant Space and the Retail Space.

Completion Date: With respect to the Residential Units, July 25, 2011; with respect to the Hotel, January 7, 2011; with respect to the Venue May 25, 2011; with respect to the Commercial Space, January 7, 2011; and with respect to the entire Project, December 31, 2011.

Condominiums: All condominium units, including, without limitation, Residential Units, included within the Project.

Condominium Documents: As such term is defined in Section 8.1(s).

Condominium Marketing License Agreement: That certain Condominium Marketing License Agreement dated as of October 26, 2006 by and between Stratus Block 21 Investments, L.P. (predecessor in interest to Borrower), and Starwood Hotels & Resorts Worldwide, Inc.

Construction or construction: The construction and equipping of the Improvements in accordance with the Approved Plans and Specifications, and related improvements required to be performed by Borrower under Sales Agreements (including all off-site improvements reasonably required for use and operation of the Improvements) and the installation of all personal property, fixtures and equipment required for the operation of the Project or required under Sales Agreements.

Construction Schedule: The schedule attached hereto as Exhibit B establishing a timetable for completion of the Construction, showing, on a monthly basis, the anticipated progress of the Construction, and showing that the Improvements can be completed on or before the applicable Completion Date and that the Residential Units will be delivered prior to any outside dates, if any, provided for in the Sales Agreements.

Control: As such term is used with respect to any person or entity, including the correlative meanings of the terms "controlled by" and "under common control with," shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management policies of such person or entity, whether through the ownership of voting securities, by contract or otherwise.

Declaration of Condominium: The Master Condominium Declaration and the Residential Condominium Declaration, individually or collectively, as the context shall infer.

Deed of Trust: That certain Second Lien Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing, executed by Borrower for the benefit of Lender and its successors and assigns, securing this Agreement, the Note, and all obligations of Borrower in connection with the Loan, granting a second priority lien on Borrower's fee interest in the Project, subject only to the Senior Deed of Trust and the Permitted Exceptions.

Default or default: Any event, circumstance or condition, which, if it were to continue uncured, would, with notice or lapse of time or both, constitute an Event of Default hereunder.

Deposits: The Earnest Money Deposits and the Upgrade Deposits.

Design Professionals: As such term is defined in Section 9.1(a).

Earnest Money Deposits: As such term is defined in Section 14.3 (a).

Environmental Indemnity: An environmental indemnity from Borrower and Guarantor, jointly and severally, indemnifying Lender with regard to all matters related to Hazardous Material and other environmental matters.

Environmental Proceedings: Any environmental proceedings, whether civil (including actions by private parties), criminal, or administrative proceedings, relating to the Project.

Environmental Report: An environmental report prepared at Borrower's expense by a qualified environmental consultant approved by Lender in its sole discretion addressed to Lender

(or subject to separate letter agreement permitting Lender to rely on such environmental report), or otherwise delivered to Lender, which complies with the USEPA "all appropriate inquiry" rule contained in 40 CRF Part 312. That certain Phase I Environmental Site Assessment dated October 9, 2009, prepared by Terracon Consultants, Inc. for the benefit of Beal Bank Nevada and CJUF II Stratus Block 21 LLC and Stratus Block 21 Investments, L.P., is acceptable to Lender.

Equity Investment: The sum of the Initial Equity Investment plus all other amounts of equity required to be provided by Borrower pursuant to the Senior Loan Agreement to keep the Senior Loan in balance and otherwise complete Construction and perform all other obligations of Borrower in regard to the Senior Loan and/or under this Agreement and the other Loan Documents.

ERISA: The Employee Retirement Income Security Act of 1974, as amended, and the regulations promulgated thereunder from time to time.

Escrow Agent: As such term is defined in Section 14.3(a).

Escrow Agreement: As such term is defined in Section 8.1(w).

Event of Default: As such term is defined in Section 19.1.

Excess Parking Spaces: As such term is defined in Section 14.1.

Full Loan Opening: As such term is defined in the Senior Loan Agreement.

General Contract: As such term is defined in Section 9.1(a).

General Contract Price: As such term is defined in Section 9.1(a).

General Contractor: Austin Building Company.

Governmental Approvals: Collectively, all consents, licenses, and permits and all other authorizations or approvals required from any Governmental Authority for the Construction in accordance with the Approved Plans and Specifications or the sale of the Residential Units.

Governmental Authority: Any federal, state, county or municipal government, or political subdivision thereof, any governmental or quasi-governmental agency, authority, board, bureau, commission, department, instrumentality, or public body, or any court, administrative tribunal, or public utility.

Guarantor: Stratus Properties, Inc., a Delaware corporation.

Guarantor Financial Covenants: The covenants of Guarantor set forth in Section 26 of the Guaranty.

Guaranty: That certain Guaranty Agreement, of even date herewith, executed by Guarantor in favor of Lender, by which Guarantor guarantees the payment and performance of all obligations of Borrower with regard to the Loan, including, without limitation, (i) payment of all amounts due with regard to the Loan and (ii) the lien-free and timely completion of the Project in accordance with all provisions of this Agreement and Borrower's obligation to keep the Senior Loan In Balance pursuant to the Senior Loan Agreement and to pay for all cost overruns.

Hard Costs: Any and all costs related to or incurred in connection with the construction of the Project, including, without limitation, the cost of all labor, materials and equipment, but excluding any fees for architectural and engineering services, marketing fees, financing costs, developers' fees and other similar soft fees and costs. The Hard Costs include the items delineated as such on the Budget.

Hazardous Material: Means and includes gasoline, petroleum, asbestos containing materials, explosives, radioactive materials or any hazardous or toxic material, substance or waste which is defined by those or similar terms or is regulated as such under any Law of any Governmental Authority having jurisdiction over the Project or any portion thereof or its use, including: (i) any "hazardous substance" defined as such in (or for purposes of) the Comprehensive Environmental Response, Compensation and Liability Act, 42 U.S.C.A. § 9601(14) as may be amended from time to time, or any so-called "superfund" or "superlien" Law, including the judicial interpretation thereof; (ii) any "pollutant or contaminant" as defined in 42 U.S.C.A. § 9601(33); (iii) any material now defined as "hazardous waste" pursuant to 40 C.F.R. Part 260; (iv) any petroleum, including crude oil or any fraction thereof; (v) natural gas, natural gas liquids, liquefied natural gas, or synthetic gas usable for fuel; (vi) any "hazardous chemical" as defined pursuant to 29 C.F.R. Part 1910; (vii) any mold or fungus that may cause an allergic, toxic or inflammatory response in humans arising from exposure to such mold or fungus in indoor air, and (viii) any other toxic substance or contaminant that is subject to any other Law or other past or present requirement of any Governmental Authority. Any reference above to a Law, includes the same as it may be amended from time to time, including the judicial interpretation thereof.

Hotel: As such term is defined in Recital A.

Hotel Documents: The Condominium Marketing License Agreement, the Hotel Operating Agreement, and/or the Technical Services Agreement, individually or collectively, as the context may infer.

Hotel Operator: W Hotel Management, Inc., a Delaware corporation, an Affiliate of Starwood Hotel & Resorts Worldwide, Inc.,

Hotel Operating Agreement: That certain W Austin Hotel Operating Agreement by and between Stratus Block 21 Investments, L.P., and Starwood Hotel & Resorts Worldwide, Inc., dated as of October 26, 2006, as amended by First Amendment to Operating Agreement dated January 30, 2008, and Second Amendment to W Austin Hotel Operating Agreement, dated

May 6, 2008, as such agreement was assigned by Stratus Block 21 Investments, L.P. to Borrower by virtue of that certain Assignment and Assumption Agreement dated as of July 30, 2007; and as such agreement was assigned by Starwood Hotel & Resorts Worldwide, Inc. to Hotel Operator by virtue of that certain Assignment and Assumption Agreement dated as of July 30, 2007.

HUD: United States Department of Housing and Urban Development.

ILSA: The Interstate Land Sales Full Disclosure Act, 42 USC 1701 et. seq., as amended.

Improvements: All of the improvements referred to in Recital A hereto and more particularly described in the Approved Plans and Specifications and any offsite improvements reasonably required to be constructed by Borrower for the use or operation of the improvements described in Recital A.

Including or including: Means "including, but not limited to" of determination in question.

Indemnified Party: As such term is defined in Section 15.1(t).

Initial Equity Investment: As such term is defined in Section 8.1(a).

Insurance Policy: As such term is defined in Section 8.1(e).

Intercreditor Agreement: As such term is defined in Recital C.

Interest Rate: The interest rate in effect from time to time under the Note.

Internal Revenue Code: The Internal Revenue Code of 1986, as amended from time to time.

Land: As such term is defined in Recital A.

Laws: Collectively, all federal, state and local laws, statutes, codes, ordinances, orders, rules and regulations, including judicial opinions or precedential authority in the applicable jurisdiction.

Lease and Leases: The collective reference to all leases, subleases and occupancy agreements affecting the Project or any part thereof now existing or hereafter executed and all amendments, modifications or supplements thereto approved in writing by Lender. Lease refers to any of such leases, subleases or occupancy agreements.

Lender: As such term is defined in the opening paragraph of this Agreement and including any successor owner or holder of the Loan or any part of the Loan (other than a participation interest in the Loan) from time to time.

List Price: As such term is defined in Section 14.1.

Loan: The loan, in the principal amount of \$30,000,000.00 to be provided by Lender to Borrower pursuant to this Agreement and which is evidenced by the Note.

Loan Documents: The collective reference to this Agreement, the documents and instruments listed in Section 4.2. and all the other documents and instruments entered into from time to time, evidencing or securing the Loan or any obligation of payment thereof or performance of Borrower's or Guarantor's obligations in connection with the transaction contemplated hereunder, each as amended and/or amended and restated.

Loan Term: The period of time commencing on the date of this Agreement through and including the date the Loan is repaid in full.

Master Association: As such term is defined in Section 14.8(d).

Master Condominium: As such term is defined in Section 14.8(c).

Master Condominium Declaration: That certain Declaration of Condominium Regime for Block 21 Master Condominiums to be recorded against the Project upon completion thereof, which will subdivide the Project into eleven (11) "Master Units."

Material Adverse Change or material adverse change: While the Senior Loan is outstanding, a Material Adverse Change means a Material Adverse Change as determined by Senior Lender under the Senior Loan Agreement. After the Senior Loan is paid in full, then Material Adverse Change means if, in Lender's sole and reasonable discretion, the operations or financial condition of a person, entity, or property has changed in a manner likely to impair materially the value of Lender's security for the Loan, prevent timely repayment of the Loan, or otherwise prevent the applicable person or entity from timely performing any of its material obligations under the Loan Documents.

Material Contracts: The City Documents, the Hotel Documents, the General Contract, each Sale Agreement,

Maturity Date: As such term is defined in Section 4.3.

Net Operating Income: For the applicable month, the gross income from the Project less, to the extent actually paid during the applicable period, a prorated management fee (not to exceed 4% of annual gross revenues), customary monthly operating expenses (not including accrued, unpaid interest on the Loan), and reasonable prorated tax and insurance reserves; provided, however, for the Hotel, Net Operating Income shall be deemed the amount distributable by Hotel Operator to Borrower pursuant to the Hotel Operating Agreement.

Net Profits Agreement: Means that certain Profits Interest Agreement dated on of the date hereof executed by Borrower for the benefit of Lender.

Net Sales Proceeds: The greater of (i) gross sales price paid by any Residential Unit Purchaser for its respective Residential Unit (exclusive of Residential Unit customization items

paid for from Upgrade Deposits, but inclusive of all Upgrade Profits and inclusive of all fees and other amounts paid by Residential Unit Purchasers in excess of the purchase price) minus brokerage commissions (limited to 3% of such gross sales price if no Residential Unit Purchaser's broker is to be paid by Borrower and 6% of such gross sales price if both the Residential Unit Purchaser's broker and Borrower's broker are to be paid by Borrower), Starwood Hotels & Resorts Worldwide, Inc. licensing fees in the amount of 4.5% of the gross sales price for each Residential Unit sold, title costs, legal fees and other customary closing costs associated with the sale of such Residential Unit that are paid or incurred by Borrower, provided that in calculating Net Sales Proceeds closing costs shall be excluded from this clause to the extent funded from the Loan (rather than being paid from gross sales proceeds) or (ii) 89% of such gross sales price if both the Residential Unit Purchaser's broker and Borrower's broker are to be paid by Borrower and 92% of such gross sales price if the Residential Unit Purchaser's broker is not to be paid by Borrower.

Non-Disturbance Agreement: That certain Subordination and Non-Disturbance Agreement by and among Hotel Operator, Lender and Borrower, dated as of March 31, 2010,

Note: That certain Promissory Note, of even date herewith, in the amount of Thirty Million Dollars (\$30,000,000), executed by Borrower and payable to the order of Lender, evidencing the Loan.

OFAC: As such term is defined in Section 3.1(v).

Office Space: As such term is defined in Recital A.

Parking Garage: As such term is defined in Recital A.

Parking Space: As such term is defined in Recital A.

Permitted Exceptions: Those matters listed on Exhibit C attached hereto, to which title to the Project may be subject at the Closing, and thereafter such other title exceptions as are acceptable to Lender in its sole discretion and approved by Lender in writing.

Person: Any natural person, partnership, limited liability company, corporation, trust, Governmental Authority or other entity.

Plans and Specifications: As such term is defined in Section 9.1(f).

Pre-sale Requirement: As such term is defined in Section 8.1(v).

Price List Schedule: As such term is defined in Section 14.1.

Proceeding: As such term is defined in Section 21.12.

Proceeds: As such term is defined in Section 16.1(a).

Project: The collective reference to (i) the Land, together with all buildings, structures and improvements located or to be located thereon, including the Improvements, (ii) all rights, privileges, easements and hereditaments relating or appertaining thereto, including, without limitation, all of Borrower's right, title and interest under the Condominium Documents, including, without limitation, all telecommunication easements, development rights and Declarant/special Declarant rights, and (iii) all personal property, fixtures and equipment required or beneficial for the operation thereof.

Project Costs: Hard Costs, Soft Costs and other costs associated with the construction of the Project as set forth in the Budget.

Qualifying Sales Agreement: (i) an agreement for the sale of a Residential Unit meeting all of the conditions required by this Agreement, and (ii) those certain Sales Agreements described on Exhibit M attached hereto that have been delivered to Lender prior to the date of this Agreement.

Release Price: As such term is defined in Section 14.9 for the Residential Unit being released.

Remaining Units: Residential Units which have not been conveyed to Residential Unit Purchasers as of the time of determination of the Remaining Units (and, therefore, remain as collateral for the Loan).

Rentable Square Feet: The number of indoor net rentable square feet in any particular portion of the Commercial Space or other space, as measured from the interior of the glass in the exterior walls, the middle of demising walls between rentable spaces and to the public side of any common area walls, but excluding balconies, terraces, hallways, common areas, lobbies, loft space or "loft walls" and the structural walls and areas of exit stairs, elevator shafts, and common mechanical shafts. (One such rentable square foot is referred to in the singular as a "Rentable Square Foot".)

Required Permits: Each building permit, environmental permit, utility permit, land use permit and any other permits, approvals or licenses issued by any Governmental Authority that are required in connection with the Construction, marketing, sale or operation of the Project.

Residential Association: As such term is defined in Section 14.8(d).

Residential Condominium: As such term is defined in Section 14.8(c).

Residential Condominium Declaration: That certain sub-condominium declaration, entitled the "Subordinate Declaration of Condominium Regime for Block 21 Hotel Residential Condominiums," to be recorded against the Residential Units and Hotel.

Residential Unit: As such term is defined in Recital A. For the sake of clarity, the Hotel is not included in the defined term "Residential Units."

Residential Unit Purchaser: The contract purchaser(s) under each Sales Agreement.

Restaurant Space: As such term is defined in Recital A.

Retail Space: As such term is defined in Recital A.

Saleable Square Feet: The number of indoor net saleable square feet in a Residential Unit or other space, as measured from the exterior of the exterior walls, the middle of demising walls between Residential Units and to the public side of any common area walls, but excluding balconies, terraces, common hallways, common mechanical shafts, lobbies, loft space or "loft walls" and the structural walls and areas of exit stairs, elevator shafts, and other common areas. (One such saleable square foot is referred to in the singular as a "Saleable Square Foot".)

Sales Agreement: As such term is defined in Section 14.2.

Sales Notice: As such term is defined in <u>Section 14.9</u>.
Sales Report: As such term is defined in <u>Section 14.6</u>.

Senior Deed of Trust: As such term as defined in Recital B.

Senior Loan Documents: As such term as defined in Recital B.

Senior Loan Agreement: As such term as defined in Recital B.

Senior Note: As such term as defined in Recital B.

Soft Costs: All costs incurred or to be incurred in connection with the Project, other than the Hard Costs, including, without limitation, interest on the Loan, fees incurred in connection with the Loan, commissions, appraisal fees, architectural and engineering fees, title and recording charges, legal fees, real estate taxes and other impositions and sales and marketing costs. Soft Costs shall only include the items delineated as such on the Budget.

Soil Report: A soil test report prepared by a licensed engineer satisfactory to Lender indicating to the satisfaction of Lender that the soil and subsurface conditions underlying the Project will support the Improvements. The geotechnical report covering the Land prepared by Henley Johnston & Associates, Inc., dated February, 2007, submitted to and approved by Senior Lender is deemed acceptable to Lender.

State: The State of Texas.

Subcontracts: Subcontracts for labor or materials to be furnished to the Project.

Substantial Completion: The satisfaction of all of the following conditions: (a) the date when the Construction shall have been completed (except for Punch List Items and minor items which can be fully completed without material interference with the use and operation of the Project) in accordance with the Approved Plans and Specifications; and (b) all material permits

and approvals required for the normal use and occupancy of the Project (including a temporary certificate of occupancy) shall have been issued by the appropriate Governmental Authority and shall be in full force and effect to such extent under items (a) and (b) so that Borrower has the absolute right and ability under applicable Laws to convey and deliver Residential Units to the respective Residential Unit Purchasers and open and operate the Hotel.

Technical Services Agreement: That certain Technical Services Agreement dated as of October 26, 2006 by and between Stratus Block 21 Investments, L.P. (predecessor in interest to Borrower), and Starwood Hotels & Resorts Worldwide, Inc.

Tenant: The tenant under a Lease.

Title Insurer: Commonwealth Title Insurance Company, or such other title insurance company licensed in the State as may be approved in writing by Lender.

Title Policy: That certain Lender's Title Insurance Policy, issued by the Title Insurer to Lender with regard to the Deed of Trust, insuring the lien of the Deed of Trust as a valid lien upon the Project and all appurtenant easements, and subject to no other exceptions other than the Senior Deed of Trust and the Permitted Exceptions.

Transfer: Any sale, transfer, lease (other than a Lease approved by Lender), conveyance, alienation, pledge, assignment, mortgage, encumbrance, hypothecation or other disposition of (a) all or any portion of the Project or any portion of any other security for the Loan, (b) all or any portion of Borrower's right, title and interest (legal or equitable) in and to the Project or any portion of any other security for the Loan, or (c) any interest in Borrower or any interest in any entity, which directly or indirectly holds an interest in, or directly or indirectly controls, Borrower.

TUCA: As such term is defined in Section 14.8(c).

Unavoidable Delay: Any delay in the construction of the Project, caused by natural disaster, fire, earthquake, hurricanes, tropical storms, floods, war, acts of terrorism, explosion, extraordinary adverse weather conditions, inability to procure or a general shortage of labor, equipment, facilities, energy, materials or supplies in the open market, failure of transportation, strikes or lockouts, or like causes, so long as such cause is not within the reasonable control of Borrower, but in no event to exceed ninety (90) days in the aggregate. In no event shall lack of funds be deemed an Unavoidable Delay.

Upgrade Deposit: As such term is defined in Section 14.3(c).

Upgrade: As such term is defined in Section 14.3(c).

Upgrade Profit: The amount by which Borrower's costs of providing any Upgrade is less than the cost charged to the Residential Unit Purchaser for such Upgrade.

Venue: As such term is defined in Recital A.

Venue Documents: The documents hereafter entered into by Borrower as herein provided for the operation of the Venue. It is currently contemplated that the Venue Documents will include a Management Agreement with Live Nation Worldwide, Inc. and a Block 21 Master Agreement with Capital of Texas Telecommunications Council.

Waste Management Plan: As such term is defined in Section 9.2(b).

2.2 Other Definitional Provisions.

All terms defined in this Agreement shall have the same meanings when used in the Note, Deed of Trust, any other Loan Document, or any certificate or other document made or delivered pursuant hereto. The words "hereof," "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement.

ARTICLE 3

BORROWER'S REPRESENTATIONS AND WARRANTIES

3.1 Representations and Warranties.

To induce Lender to execute this Agreement and perform its obligations hereunder, Borrower hereby represents and warrants to Lender as follows:

- (a) Borrower has good and indefeasible fee simple title to the Project, subject only to the Senior Deed of Trust and the Permitted Exceptions.
- (b) Except as described on Exhibit D hereto, no litigation or proceeding is pending, or to the best of Borrower's actual knowledge threatened in writing, against Borrower or Guarantor, that could, if adversely determined, be reasonably expected to cause a Material Adverse Change with respect to Borrower, Guarantor or the Project. There are no pending Environmental Proceedings and Borrower has no actual knowledge of any threatened Environmental Proceedings or any facts or circumstances that are reasonably likely give rise to any future Environmental Proceedings.
- (c) Borrower is a duly organized and validly existing limited liability company and has full power and authority to execute, deliver and perform all Loan Documents to which Borrower is a party, and such execution, delivery and performance have been duly authorized by all requisite action on the part of Borrower; Borrower has been a single purpose entity in compliance with Section 15.2 hereof since its formation.
- (d) No consent, approval or authorization of or declaration, registration or filing with any Governmental Authority or nongovernmental person or entity, including any creditor, partner, member or shareholder of Borrower, Guarantor, is required in connection with the execution, delivery and performance of this Agreement or any of the Loan Documents other than the recordation of the Deed of Trust and the Declaration of Condominium for the Project and the filing of UCC-1 Financing Statements, except for such consents, approvals or authorizations of or declarations or filings with any Governmental Authority or non-governmental person or entity

where the failure to so obtain would not have a material adverse effect on Borrower, Guarantor or which have been obtained as of any date on which this representation is made or remade.

- (e) The execution, delivery and performance of this Agreement, the execution and payment of the Note and the granting of the Deed of Trust and other security interests under the other Loan Documents have not constituted and will not constitute, upon the giving of notice or lapse of time or both, a breach or default under any other agreement to which Borrower or Guarantor is a party or may be bound or affected, or a violation of any law or court order that may affect the Project, any part thereof, any interest therein, or the use thereof.
- (f) Each of the Material Contracts, other than the Hotel Documents, is in full force and effect. There is no Default or Event of Default under the Senior Agreement, any of the Senior Loan Documents, this Agreement, the other Loan Documents, or, to the knowledge of Borrower, any of the Material Contracts (other than the Hotel Documents), nor, to Borrower's knowledge, is there any condition that, after notice or the passage of time or both, would constitute a Default or an Event of Default under any of said documents. No counterparty to any Material Contract has sent to Borrower any notice alleging a default on the part of Borrower under any Material Contract or any other material notice of an adverse matter relating to any such Material Contract, except for the notices, if any, which have been provided by Borrower to Lender. Borrower is not aware of any material default or failure to perform which exists under any Material Contract on the part of any party to such Material Contract. Certain Residential Unit Purchasers have not yet fully funded the required Earnest Money Deposit as reflected on Exhibit M-1.
- (g) (i) No condemnation of any portion of the Project, (ii) no condemnation or relocation of any roadways abutting the Project, and (iii) no proceeding to deny access to the Project from any point or planned point of access to the Project, has commenced or, to Borrower's actual knowledge, is contemplated by any Governmental Authority.
- (h) The amounts set forth in the Budget present a full and complete itemization by category of all costs, expenses and fees that Borrower reasonably expects to pay or reasonably anticipates becoming obligated to pay with respect to the Loan and to complete the Construction (including all off-site improvements to be paid for by Borrower), operate the Project and market and sell the Residential Units. Borrower is unaware of any other such costs, expenses or fees that are material and are not covered by the Budget. Borrower further warrants that neither Borrower, Guarantor, nor any of their respective Affiliates are receiving any other payments, distributions, or other consideration directly or indirectly from Borrower, the Project, its seller, contractors or any other party associated with the Project other than the Permitted Affiliate Expenses.
- (i) Neither the construction of the Improvements nor the use of the Project when completed in accordance with the Approved Plans and Specifications and the contemplated accessory uses will violate (i) any Laws (including subdivision, zoning, building, environmental protection and wetland protection Laws), or (ii) any building permits, restrictions of record, or agreements affecting the Project or any part thereof. Neither the zoning authorizations,

approvals or variances nor any other right to construct or to use the Project is to any extent dependent upon or related to any real estate other than the Land. All Governmental Approvals required for the Construction in accordance with the Approval Plans and Specifications have been obtained (except for those Governmental Approvals that cannot or need not be obtained until a later stage of the Construction or completion of Construction, in which case such Governmental Approvals will be obtained by Borrower on a timely basis and copies will be delivered to Lender on the earliest possible date) and all Laws relating to the Construction and operation of the Improvements have been complied with in all material respects and to Borrower's knowledge, after due inquiry, all permits and licenses, required for the operation of the Project that cannot be obtained until the Construction is completed can be obtained if the Improvements are completed in accordance with the Approved Plans and Specifications.

- (j) The Project will have adequate water, gas, if applicable, and electrical supply, storm and sanitary sewerage facilities, other required public utilities, fire and police protection, and means of access between the Project and public highways, and none of the foregoing will be foreseeably delayed or impeded by virtue of any requirements under any applicable Laws.
- (k) No brokerage fees or commissions are payable by or to any person in connection with this Agreement or the Loan to be disbursed hereunder, other than to Holliday Fenoglio Fowler, Greg Barnes and Quincy Adams. Using Loan proceeds, at Closing, Borrower will pay THREE HUNDRED FIFTY THOUSAND AND NO/100 DOLLARS (\$350,000.00) to Holliday Fenoglio Fowler, which firm will promptly pay \$75,000.00 to Greg Barnes and \$75,000.00 to Quincy Adams out of such \$350,000.00 received at Closing.
- (l) All financial statements and other information previously furnished by Borrower or Guarantor to Lender in connection with the Loan are true, complete and correct in all material respects and fairly present the financial conditions of the subjects thereof as of the respective dates thereof and do not fail to state any material fact necessary to make such statements or information not misleading, and no Material Adverse Change with respect to Borrower, Guarantor has occurred since the respective dates of such statements and information. Neither Borrower nor Guarantor has any material liability, contingent or otherwise, not disclosed in such financial statements and all charges payable with respect to the Project are current and not in default. Except as previously disclosed in writing to Lender, neither Borrower nor Guarantor, nor any officer or director of Borrower, nor any equity owner of Borrower, Guarantor, or any off Borrower's or Guarantor's respective Affiliates (excluding investors in beneficial owners of CJUF and any shareholders of Guarantor other than those that are considered "insiders" under SEC regulations): (i) has ever been the subject of any criminal proceedings (other than minor traffic violations); (ii) has ever been the owner, whether directly or indirectly, to a deed in lieu of foreclosure; or (iv) is currently a party to any material pending litigation or administrative proceedings, or subject to any judicial or non-judicial orders or consent agreements.
- (m) Except as disclosed in any Environmental Report delivered by Borrower to Lender prior to the date hereof, (i) to Borrower's actual knowledge, the Project is in a safe

condition, and, except for small quantities of Hazardous Materials lawfully used in the ordinary course of construction, maintenance and operation of the Project, is free of all Hazardous Materials and is in compliance with all applicable Laws; (ii) except for small quantities of Hazardous Materials lawfully used in the ordinary course of construction, maintenance and operation of the Project, nor not, to the actual knowledge of Borrower, any other person or entity, has ever caused or permitted any Hazardous Material to be placed, held, located or disposed of on, under, at or in a manner to affect the Project, or any part thereof, and the Project has never been used (whether by Borrower or, to the actual knowledge of Borrower, by any other person or entity) for any activities involving, directly or indirectly, the use, generation, treatment, storage, transportation, or disposal of any Hazardous Material; (iii) neither the Project nor Borrower is subject to any existing, pending, or, to Borrower's actual knowledge, threatened investigation or inquiry by any Governmental Authority, and the Project is not subject to any remedial obligations under any applicable Laws pertaining to health or the environment; and (iv) to the actual knowledge of Borrower, there are no underground tanks, vessels, or similar facilities for the storage, containment or accumulation of Hazardous Materials of any sort on, under or affecting the Project.

- (n) For all purposes the Project may be mortgaged, conveyed and otherwise dealt with as an independent parcel and is a separate real estate tax parcel.
- (o) Borrower and its agents have not entered into any Leases, subleases or other arrangements for occupancy of space within the Project (other than Sales Agreements that permit occupancy by the Residential Unit Purchasers following closing thereunder and the Hotel Operating Agreement).

When the Construction is completed substantially in accordance with the Approved Plans and Specifications, no building or other improvement will encroach upon any property line, building line, setback line, side yard line or any recorded or visible easement (or other easement of which Borrower is aware or has reason to believe may exist) in violation thereof. Borrower is not a party in interest to any plan defined or regulated under ERISA, and the assets of Borrower are not "plan assets" of any employee benefit plan covered by ERISA or Section 4975 of the Internal Revenue Code.

- (p) Borrower is not a "foreign person" within the meaning of Section 1445 or 7701 of the Internal Revenue Code.
- (q) Borrower uses no trade name other than (i) its actual name set forth herein, (ii) "W Hotel and Residences" and (iii) "W Austin Hotel". The principal place of business of Borrower is as stated in Article 22.
 - (r) Borrower's place of organization is Delaware.
- (s) Except as set forth in Exhibit M, there are no Sales Agreements to purchase Residential Units. The Sales Agreements listed on Exhibit M are in full force and effect and such Sales Agreements are not subject to any rights of rescission. Borrower hereby represents that Exhibit M is a true, accurate and complete schedule of all Sales Agreements and sets forth:

(i) the name of Residential Unit Purchaser, (ii) the Residential Unit being purchased, (iii) any Upgrades, (iv) any Upgrade Deposit, (v) the purchase price, and (vi) the Eamest Money Deposit. All Sales Agreements are (or when entered into, and after expiration of statutory rescission periods, will be) Qualifying Sales Agreements. Except for the Sales Agreements referenced on Exhibit M-1 where delinquent Earnest Money Deposit installments are noted, no event of default, or any event that, with the passage of time or the giving of notice, or both, would constitute an event of default, has occurred pursuant to the terms of any of the Sales Agreements on the part of Borrower or, to Borrower's actual knowledge, the other parties thereto. No Residential Unit Purchaser under the Sales Agreements listed on Exhibit M has terminated its respective Sales Agreement and there are no side agreements with any Residential Unit Purchasers modifying any of the terms of the Sales Agreements or otherwise.

- (t) All Sales Agreements are exempt from or will comply with the requirements of ILSA, and Laws of the State (and any applicable local Laws), so that (i) the sale of the Residential Units is lawful and will not be subject to interruption due to a violation of Laws, (ii) no Sales Agreement is terminable under any of such Laws (other than the termination rights contained in such Sales Agreement), and (iii) neither Borrower nor the Project will be subject to any civil or criminal penalties by reason of failure to comply with such Laws. The marketing and sale of Residential Units by Borrower (and any marketing or sales of Residential Units) is, and at all times has been, in compliance with all Laws pertaining to the sale of condominiums (and/or residential real estate generally). All consents and approvals needed for the sale of Residential Units under applicable federal, state and local Laws have been received and remain in full force and effect.
- (u) The Hotel Documents are in full force and effect. No event of default, or any event that, with the passage of time or the giving of notice, or both, would constitute an event of default, has occurred pursuant to the terms of the Hotel Documents, either on the part of Borrower or, to Borrower's actual knowledge, the other parties thereto. Hotel Operator, or any other party to any Hotel Document, has not sent to Borrower any notices of default under any Hotel Document, nor has Hotel Operator, or any other party to any Hotel Document, sent to Borrower any other written notices of a material nature. The Hotel Documents have not been amended (except as set forth in the definition of Hotel Documents). There are no other agreements, written or oral, with Hotel Operator, Starwood Hotels & Resorts Worldwide, Inc., or any Affiliates of the foregoing, that supplement or modify any of the terms of any of the Hotel Documents or otherwise.
- (v) Neither Borrower, Guarantor nor any other person owning an interest in Borrower is (or will be) a person with whom Lender is restricted from doing business under regulations of the Office of Foreign Asset Control ("OFAC") of the Department of the Treasury of the United States of America (including, those persons named on OFAC's Specially Designated and Blocked Persons list) or under any statute, executive order (including the September 24, 2001 Executive Order Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism), or other governmental action and is not and shall not knowingly engage in any dealings or transactions or otherwise be associated with such persons. In addition, Borrower hereby agrees to provide Lender with any additional information

that Lender deems reasonably necessary from time to time in order to ensure compliance with all applicable Laws concerning money laundering and similar

- (w) Borrower shall have complied, in all respects, with the provisions of the USA PATRIOT Act of 2001, as applicable to Borrower and the Project.
- (x) The Project complies and, when constructed, shall comply with all requirements of (i) the City Documents, (ii) that certain Restrictive Covenant by Zoning Case No. C14-06-0190, dated December 5, 2006 by The City of Austin recorded as Instrument No. 200634734, and (iii) that certain Restrictive Covenant Regarding Unified Development and Maintenance of Drainage Facilities, dated January 9, 2008 by CJUF II Stratus Block 21, LLC recorded as Instrument No. 2008013723.
 - (y) All statements set forth in the Recitals are true and correct in all material respects.
- (z) Construction of the Improvements to date is in compliance with the Approved Plans and Specifications, the Governmental Approvals and the requirements of each applicable Material Contract. To date, Borrower has contributed \$146,047,370 of its Initial Equity Investment to pay Budget Line Item Costs as shown on Exhibit G.
- (aa) To Borrower's actual knowledge, all factual information with respect to the Borrower, the Guarantor, the Project, the Senior Loan and/or the Loan, set forth in the Senior Loan Documents or in the reports, other papers and data set forth in the items listed on Exhibit T (the "Due Diligence Materials") and, to Borrower's actual knowledge, all factual statements and representations made to the Lender by or on behalf of the Borrower in the Due Diligence Materials, were, at the time the same were so furnished or made, when taken together with all such other factual information, reports and other papers and data previously so furnished in connection with this Agreement and the Loan and all such other factual statements and representations previously so made in connection with this Agreement and the Loan, complete and correct in all material respects, and, to Borrower's actual knowledge, did not as of the date so furnished or made and as of the date hereof, contain any untrue statement of a material fact or omit to state any material fact necessary in order to make the statements contained therein not misleading in light of the circumstances in which the same were made. Borrower has not intentionally withheld any material information from Lender concerning the Borrower, the Guarantor, the Project or the loan evidenced by the Senior Loan Documents.

3.2 Survival of Representations and Warranties.

Borrower agrees that all of the representations and warranties set forth in Section 3.1 and elsewhere in this Agreement are true in all material respects as of the date hereof, will be true in all material respects at Closing and, except for matters that have been disclosed by Borrower and approved by Lender in writing or otherwise permitted by this Agreement, will be true in all material respects at all times thereafter (including at Full Loan Opening) until the Loan has been repaid and Borrower's obligations hereunder have been satisfied in full. Each request for a disbursement under the Loan Documents shall constitute a reaffirmation of such representations and warranties, as deemed modified in accordance with the disclosures made and approved as

aforesaid, as of the date of such request. It shall be a condition precedent to the Closing Funding and each subsequent disbursement that each of said representations and warranties is true and correct in all material respects as of the date of such requested disbursement. In addition, at Lender's request, Borrower shall reaffirm such representations and warranties in writing prior to each disbursement hereunder.

ARTICLE 4

LOAN AND LOAN DOCUMENTS

4.1 Funding of the Loan.

(a) The Loan will be advanced to Borrower in three draws. The first draw in the amount of \$10,000,000 will be funded at Closing (the "Closing Funding"); the second draw in the amount of \$11,000,000 will be funded on April 16, 2010; and the third draw in the amount of \$9,000,000 will be funded on May 17, 2010. Lender's obligation to fund such second draw and third draw is referred to in this Agreement as the "Unfunded Commitments." Lender will disburse the Unfunded Commitment into Borrower's account with Senior Lender and Borrower shall be entitled to withdraw amounts (in one or more draws) from that account to pay Project Costs upon satisfaction of the conditions in the Senior Loan Agreement for a disbursement of proceeds to pay such Project Costs, including Senior Lender's review and approval of the applicable draw request.

4.2 Loan Documents.

Borrower agrees that it will, on or before the date hereof, execute and deliver or cause to be executed and delivered to Lender the following documents in form and substance acceptable to Lender:

- (a) The Note.
- (b) The Deed of Trust.
- (c) The Guaranty.
- (d) The Environmental Indemnity.
- (e) The Collateral Assignment of Hotel Documents.
- (f) A collateral assignment, to the extent assignable, of construction documents, including, without limitation, the General Contract, all architecture, Design Professional and engineering contracts, Plans and Specifications, permits, licenses, approvals and development rights, together with consents to the assignment and continuation agreements from the General Contractor, the architect, real estate broker and other parties reasonably specified by Lender.

- (g) A collateral assignment, to the extent assignable, of all Sales Agreements, Earnest Money Deposits, Upgrade Deposits and all other documents relating to the establishment of a condominium regime at the Project.
- (h) Such UCC financing statements as Lender determines are advisable or necessary to perfect or notify third parties of the security interests intended to be created by the Loan Documents.
 - (i) A collateral assignment, to the extent assignable, of any management contract entered into with respect to the Project.
 - (j) The Intercreditor Agreement.
 - (k) A Pledge and Security Agreement.
 - (1) Net Profits Agreement.
- (m) Such other documents, instruments or certificates as Lender and its counsel may reasonably require, including such documents as Lender in its reasonable discretion deems necessary or appropriate to effectuate the terms and conditions of this Agreement and the Loan Documents, and to comply with the laws of the State.

4.3 Term of the Loan, Extensions.

- (a) All principal, interest and other sums due under the Loan Documents shall be due and payable in full on the second (2nd) anniversary of the date hereof (the "*Maturity Date*"). The Loan may become due and payable prior to the Maturity Date as a result of an acceleration thereof as provided in the Loan Documents.
- (b) If, not more than fifteen (15) days prior to the Maturity Date (or the extended Maturity Date as provided below), Borrower has not provided Lender with assurances satisfactory to Lender that Borrower has or will have sufficient funds to pay the Loan in full on the Maturity Date (or the extended Maturity Date) in accordance with the terms of the Loan Documents and in compliance with the Intercreditor Agreement, Lender shall have the option, but not the obligation, of extending the Maturity Date of the Loan for two (2) additional one-year periods by giving Borrower written notice of Lender's exercise of its right to extend the Maturity Date. The terms of the Note during each extended term will remain the same as before the extension, except that Borrower will, on or before the original Maturity Date or the extended Maturity Date, as applicable, pay to Lender an extension fee in the amount of \$50,000, which shall be funded from sources other than revenues or Proceeds from the Project. Borrower and Lender will execute such documentation as Lender requires to evidence the extension of the term in each case.

4.4 Prepayments.

Subject to the terms of the Intercreditor Agreement, the Loan may not be prepaid, in whole or in part, at any time prior to the first (1st) anniversary of the date of the Closing except for (a) any insurance proceeds or condemnation awards that are applied to the payment of the Loan pursuant to the Loan Documents, or (b) in accordance with Section 14.10. Borrower may prepay the Loan, in whole or in part, without premium or fee, subsequent to the first (1st) anniversary of the date of the Closing. Except for prepayments to Lender described in subpart (a) above, Borrower must provide to Lender at least five (5) business days prior written notice of each prepayment.

4.5 Required Principal Payments.

The unpaid Principal Balance of and all accrued and unpaid interest on the Loan shall be due and payable on the Maturity Date, subject to the terms of the Intercreditor Agreement. In addition, as Residential Units are sold on or after the first anniversary date of the Note, the Net Sales Proceeds, or portion thereof, from a Residential Unit Sale that are released by Senior Lender to Borrower in accordance with the terms of the Senior Loan Agreement, must be paid to Lender as a required principal payment on the Note.

4.6 Receipt of Payments.

All payments received by Lender prior to or at 2:00 p.m. (Dallas time) on a Business Day shall be credited to Borrower on the day of receipt; all payments received after 2:00 p.m. (Dallas time) on a Business Day shall be deemed received on the next succeeding Business Day.

4.7 Protective Advances.

Notwithstanding any provision herein or in the Note to the contrary, Lender has the right under the Intercreditor Agreement to cure Borrower's defaults thereunder, and with respect thereto, in the event Lender advances any or expends any funds to cure any default by Borrower under the Senior Loan Documents, such advances or expenditures will be deemed additional advances under the Note, shall bear interest as provided in the Note, shall be secured by the Deed of Trust and the other Loan Documents, and shall be payable on demand.

4.8 Lender's Inability to Fund.

Notwithstanding any provisions of Section 4.4, if Lender is prevented from funding the Unfunded Commitment by reason of bankruptcy or insolvency proceedings or an order from regulatory authorities in connection with Lender's insolvency or failure to meet regulatory requirements, then Lender shall notify Borrower of that circumstance and, so long as no Event of Default exists, Borrower may prepay the Loan in full without the payment of any prepayment premium or fee. This Section 4.8 is in addition to the legal rights and remedies of Borrower under this Agreement or at law or equity.

ARTICLE 5

INTEREST

5.1 Interest Rate.

The Loan shall bear interest as set forth in the Note. Interest shall be paid on the Loan when and as set forth in the Note.

ARTICLE 6

COSTS OF MAINTAINING LOAN

6.1 Borrower Withholding.

If by reason of a change in any applicable Laws occurring after the date hereof, Borrower is required by Law to make any deduction or withholding in respect of any taxes (other than taxes imposed on or measured by the net income of Lender or any franchise tax imposed on Lender), duties or other charges from any payment due under the Note to the maximum extent permitted by law, the sum due from Borrower in respect of such payment shall be increased to the extent necessary to ensure that, after the making of such deduction or withholding, Lender receives and retains a net sum equal to the sum that it would have received had no such deduction or withholding been required to be made.

ARTICLE 7

LOAN EXPENSE AND ADVANCES

7.1 Loan and Administration Expenses.

Borrower unconditionally agrees to pay all reasonable expenses of the Loan, including all amounts payable pursuant to Section 7.2 and any and all other fees owing to Lender pursuant to the Loan Documents, and also including, without limiting the generality of the foregoing, all recording, filing and registration fees and charges, mortgage, intangible or documentary taxes, escrow charges, title charges, all insurance premiums, title insurance premiums and other charges of the Title Insurer, printing and photocopying expenses, survey fees and charges, cost of certified copies of instruments, cost of premiums on the Title Policy, charges of the Title Insurer or other escrowee for administering disbursements, all appraisal fees, insurance consultant's fees, investigator's fees, environmental consultant's fees, reasonable travel related expenses and all reasonable costs and expenses incurred by Lender in connection with the determination of whether or not Borrower has performed the obligations undertaken by Borrower hereunder or has satisfied any conditions precedent to the obligations of Lender hereunder. Borrower shall pay the airfare and other reasonable travel expenses for each officer or analyst of Lender who inspects the Project as part of Lender's due diligence. The amount charged for airfare shall be the lesser of (i) the actual cost thereof incurred by Lender, or (ii) \$500.00 per person per visit. Borrower agrees to pay all brokerage, finder or similar fees or commissions payable in connection with the transactions contemplated hereby and shall indemnify, defend, and hold Lender harmless against all claims, liabilities, costs and expenses (including attorneys'

fees and expenses) incurred in relation to any such claim by broker, finder or similar person alleging to have dealt with Borrower in connection with this

7.2 Loan Fees.

Borrower shall pay to Lender on or before the date of this Agreement a loan commitment fee in the amount of \$50,000, which shall be fully earned and non-refundable when paid.

7.3 [reserved]

7.4 [reserved]

7.5 Lender's Attorneys' Fees and Disbursements.

Borrower agrees to pay Lender's reasonable attorneys fees and disbursements incurred in connection with this Loan both before and after the date hereof (and whether or not the Loan closes), including (i) the preparation of this Agreement, any intercreditor agreements and the other Loan Documents and the preparation of the closing binders, (ii) the disbursement and administration of the Loan and (iii) the enforcement of the terms of this Agreement and the other Loan Documents. The reasonable legal fees and disbursements to be paid by Borrower under this Section 7.5 and the reasonable legal fees and disbursements to be paid by Borrower under all other applicable provisions of this Loan Agreement and the other Loan Documents shall include the reasonable fees and expenses of Lender's inside counsel charged at a rate not higher than the rate charged by Lender's outside counsel for an attorney with equivalent experience.

7.6 Time of Payment of Fees and Expenses.

Borrower shall pay all expenses and fees incurred by Lender in connection with the Loan as of the date of this Agreement at Closing. At Closing, Lender will pay from the Closing Funding all reasonable Loan expenses and all fees payable to Lender and not previously paid. Lender may require the payment of outstanding fees and expenses as a condition to any disbursement of proceeds of the Loan.

7.7 Expenses and Advances Secured by Loan Documents.

Any and all advances or payments made by Lender under this <u>Article 7</u> from time to time, and any amounts expended by Lender pursuant to <u>Section 20.1</u>, shall, as and when advanced or incurred, constitute additional indebtedness evidenced by the Note and secured by the Deed of Trust and the other Loan Documents and shall bear interest at the rate then applicable under the Note.

7.8 Right of Lender to Make Advances to Cure Borrower's Defaults.

In the event that Borrower fails to perform any of Borrower's covenants, agreements or obligations contained in this Agreement or any of the other Loan Documents (and applicable grace or cure periods have expired, unless in Lender's judgment an emergency or other exigent

circumstance exists, in which case Lender need not wait for such period to expire), Lender may (but shall not be required to) perform any of such covenants, agreements and obligations, and any reasonable amounts expended by Lender in so doing shall constitute additional indebtedness evidenced by the Note and secured by the Deed of Trust and the other Loan Documents and shall bear interest at the Interest Rate. Lender may expend any such amounts even if such advance would result in the balance owing to Lender exceeding the stated amount of the Note.

ARTICLE 8

NON-CONSTRUCTION REQUIREMENTS PRECEDENT

8.1 Non-Construction Conditions Precedent.

Borrower agrees that Lender's obligation to make the Loan is conditioned upon Borrower's delivery, performance and satisfaction of the following conditions precedent in form and substance satisfactory to Lender in its reasonable discretion (except as otherwise expressly set forth herein) prior to (or at the time of) Closing,:

(a) Equity: Borrower represents and warrants to Lender that prior to or at Closing, Borrower has or will have contributed equity in the amount of \$146,047,370 (the "Initial Equity Investment"), to pay Project Costs in full compliance with the Senior Loan Agreement. Borrower represents and warrants that the Initial Equity Investment plus the proceeds of the Loan, totaling \$176,047,370 satisfies all of the capital required to be funded as a condition to Full Loan Opening as provided in the Senior Loan Agreement. Borrower represents and warrants that in connection with the monthly draw submittal, review and approval process, Borrower has provided Senior Lender evidence of payment of the Initial Equity Investment and such documentation has been reviewed and approved by Senior Lender and, upon request by Lender, Borrower will provide copies of such documentation to Lender. In all events, any equity contribution shall be subordinate to the rights of Lender and general unsecured creditors of Borrower. Borrower may not be indebted to any Person for any equity contribution. Borrower has provided Lender correct copies of the Amended and Restated Operating Agreement, as amended, of Borrower, setting forth each partner's equity contribution to Borrower.

Borrower's Equity Investment requirement shall be increased as a result of any changes to the Budget and as necessary to maintain the Senior Loan in balance as described in the Senior Loan Agreement.

- (b) Fees and Expenses: Borrower shall have paid all of Lender's fees and expenses as required by Article 7 or elsewhere in this Agreement, to the extent due and payable.
- (c) <u>Title and Other Documents</u>: Borrower shall have furnished to Lender the Title Policy, in form as required by Lender, with the premium for such Title Policy paid in full, together with legible copies of all documents creating Permitted Exceptions and all other legal documents affecting the Project or the use thereof. The Title Policy shall be subject only to the Senior Deed of Trust and the Permitted Exceptions. Any exception for the rights of Residential Unit Purchasers shall only be permissible if the Title Insurer insures such rights are subordinate to the Deed of Trust.

- (d) <u>Survey</u>: Borrower shall have furnished to Lender an ALTA/ACSM "Class A" Land Title Survey of the Project prepared by a licensed surveyor satisfactory to Lender.
- (e) <u>Insurance Policies</u>: Borrower shall have furnished to Lender, prior to the date hereof, certificates evidencing that insurance coverages are in effect with respect to the Project and Borrower, in accordance with the Insurance Requirements attached hereto as <u>Exhibit E</u> and incorporated herein by reference as if fully set forth herein (or such other insurance coverages reasonably acceptable to Lender), for which the premiums have been prepaid, and with endorsements satisfactory to Lender. On or before the date hereof, Borrower shall provide a copy of the insurance policy with respect to the Project and Borrower in accordance with <u>Exhibit E</u> (the "Insurance Policy").
- (f) No Litigation: Prior to the Closing and in connection with each request from Borrower for an advance of Loan proceeds, Borrower shall have furnished evidence that no litigation or proceedings shall be pending or threatened that is reasonably likely to cause a Material Adverse Change with respect to Borrower, Guarantor or the Project. A swom certificate, in form attached hereto as Exhibit D executed by an officer or manager of Borrower will serve as the evidence of no litigation or proceeding hereunder for each advance of Loan proceeds.
- (g) <u>Utilities</u>: Borrower represents and warrants that prior to closing the Senior Loan, Borrower furnished to Senior Lender (by way of utility letters or otherwise) evidence establishing to the reasonable satisfaction of Senior Lender that the Project, when constructed, will have adequate water supply, storm and sanitary sewerage facilities, telephone, gas (if applicable), electricity, fire and police protection, means of ingress and egress to and from the Project and public highways and any other required public utilities and that the Project is benefited by insured easements as may be required for any of the foregoing. Borrower further represents and warrants to Lender that the information provided to Senior Lender confirming the adequacy of utilities for the Project remains true and correct. Upon Lender's request, Borrower will provide the information previously furnished to the Senior Lender as referenced in this subpart (g) to the Lender.
- (h) <u>Attorney Opinions</u>: Borrower shall have furnished to Lender an opinion from counsel for Borrower and Guarantor, in a form satisfactory to Lender, covering due authorization, execution and delivery and enforceability of the Loan Documents and also containing such other legal opinions as Lender shall reasonably require, with customary assumptions and qualifications.
 - (i) Appraisal: Borrower has provided Lender a copy of the appraisal obtained in connection with closing the Senior Loan.
- (j) <u>Searches</u>: Borrower shall have furnished to Lender current bankruptcy, federal tax lien and judgment searches and searches of all Uniform Commercial Code financing statements filed in each place UCC Financing Statements are to be filed hereunder for Borrower and Guarantor, demonstrating the absence of (i) any UCC Financing Statement that reflect

Borrower as debtor other than those filed in connection with the Senior Agreement and (ii) any materially adverse claims.

- (k) Financial Statements: Tax Retums: Borrower shall have furnished to Lender current annual financial statements of Borrower and Guarantor, each in form and substance and certified by such individual as is acceptable to Lender. Borrower shall have furnished to Lender any federal and state tax returns of Borrower and Guarantor for the past two (2) years. Borrower and Stratus Properties shall have signed and delivered to Lender an Internal Revenue Service Tax Return Verification Form (IRS Form 4506-T). Borrower and Guarantor shall provide such other additional financial information as Lender reasonably requires, including financial statements of income and expenses for the Project and tax returns for all entities reporting the income and expenses on the Project.
 - (1) Price List Schedule. Borrower shall have furnished to Lender the Price List Schedule, as approved by Senior Lender.
- (m) Other Agreements: Borrower represents and warrants that (i) Borrower has delivered to Senior Lender executed copies of any material marketing, brokerage and development agreements entered into by Borrower in connection with the Project, the Construction and/or the sale of Residential Units, each of which Senior Lender has approved; (ii) Borrower will continue to submit to Senior Lender all such materials required to be submitted pursuant to the Senior Loan; and (iii) Borrower will, upon Lender's request, promptly submit copies of materials referenced in (i) and (ii) to Lender.
- (n) Flood Hazard: Borrower represents and warrants that Senior Lender has received evidence that the Project is not located in an area designated by the Secretary of Housing and Urban Development as a special flood hazard area and, upon Lender's request, Borrower will promptly provide such evidence to Lender.
 - (o) Zoning: Lender has received evidence satisfactory to Lender that the Project is in compliance with all zoning and land use laws.
- (p) Organizational Documents: Borrower shall have furnished to Lender proof satisfactory to Lender of authority, formation, organization and good standing in the state of its incorporation or formation and, if applicable, qualification as a foreign entity in good standing in the state of its incorporation or formation, of all corporate, partnership, trust and limited liability company entities (including Borrower and Guarantor) executing any Loan Document, whether in their own name or on behalf of another entity. Borrower shall also provide an organizational chart as well as certified resolutions in form and content satisfactory to Lender, authorizing execution, delivery and performance of the Loan Documents, and such other documentation as Lender may require to evidence the authority of the persons executing the Loan Documents.
- (q) No Default; No Material Adverse Change; No Condemnation; etc.: There shall be no Default or continuing Event of Default by Borrower under the Senior Loan Agreement, under any of the Senior Loan Documents, hereunder or under any other Loan Document; there shall have not occurred a Material Adverse Change in the financial condition of Borrower or

Guarantor or the condition of the Project that has not been cured or satisfied; and neither the Project nor any part thereof shall have suffered any material casualty or be subject to any existing or threatened condemnation or taking by eminent domain proceeding or otherwise.

- (r) <u>Easements</u>: Borrower represents and warrants that Borrower furnished to Senior Lender all easements reasonably required for the construction, maintenance or operation of the Project, and such easements shall be insured by the Title Policy and, upon Lender's request, Borrower shall promptly provide such evidence to Lender.
- (s) Condominium Documents: Borrower represents and warrants that Borrower provided Senior Lender a copy of all proposed and/or executed condominium information statements, condominium plats and/or plans, declarations (including exhibits), budgets, owners' association certificates of formation, bylaws, resolutions, rules and regulations, agreements regarding cost sharing, filings, escrow agreements and other documents pertaining to the establishment of the condominium regimes at the Project's compliance with all applicable local, state and federal Laws relating to condominiums (collectively, the "Condominium Documents"), including, without limitation, the draft Master Condominium Declaration and Residential Condominium Declaration. The Condominium Documents previously provided to Senior Lender for review will be revised, in a manner reasonably acceptable to Senior Lender, as further described in Section 14.8. Borrower agrees that it shall not record any Declaration of Condominium or other Condominium Document without the prior written consent of Senior Lender, as further described in Section 14.8. Without limiting the foregoing, Borrower agrees that it shall deliver to Senior Lender the final form of Condominium Documents, including, without limitation, condominium plats and plans for both the Block 21 Master and Block 21 Hotel/Residential Condominiums, not later than thirty (30) days prior to the date that Borrower desires to record any Declaration of Condominium or other Condominium Document. Upon request, Borrower shall promptly provide copies of all material referenced in this subpart(s) to Lender.
- (t) General Contractor and Design Professional Agreements. Borrower represents and warrants that Borrower provided Senior Lender agreements, in form as required by Senior Lender, executed by the General Contractor and each Design Professional confirming, among such other things as Senior Lender may require, (i) the amounts paid to date to such parties for the Construction and design of the Project as of the date of the Senior Loan Agreement and the amount of Retainage then held by or on behalf of Borrower, (ii) that all sub-contractors of the respective parties have been paid all amounts owed to such sub-contractors for work performed with regard to the Project to the date of such agreements and (iii) that no mechanic's or materialmen's liens or lien claims exist as of the date of such agreements for the benefit of the General Contractor or any of the Design Professionals or any of their respective sub-contractors for any work theretofore performed with regard to the Project. In addition, prior to each monthly draw and payment of Project Costs, Borrower has and will continue to deliver to Senior Lender a sworn statement of the General Contractor, approved by Borrower and Lender, covering all work done and to be done, together with lien waivers covering all work and materials for which payments have been made by Borrower as of the date that is five (5) Business Days prior to the

Closing. Upon request, Borrower shall promptly provide copies of all material referenced in this subpart (t) to Lender.

- (u) <u>Lease</u>. Borrower represents and warrants that: (i) Borrower has delivered to Senior Lender the forms of Lease to be used for the leasing of Retail Space and (ii) prior to Full Loan Opening, Borrower shall have delivered to Lender the form of Lease to be used for the leasing of Restaurant Space and Office Space as required by the Senior Loan. Upon request, Borrower shall promptly provide copies of all material referenced in this subpart (u) to Lender.
- (v) <u>Pre-sales</u>. Borrower represents and warrants that: (i) Borrower delivered to Senior Lender, prior to closing the Senior Loan, Qualifying Sales Agreements (in full force and effect) for at least 70 Residential Units relating to the sale of at least 70,761 Saleable Square Feet together with Parking Spaces for such applicable Residential Units at prices greater than or equal to the List Price for each and every such Residential Unit, constituting, in the aggregate, gross sales of greater than or equal to \$72,995,425.00 (collectively, the "Pre-sale Requirement"); and (ii) the mix of Residential Unit types and locations within the Project subject to those sales agreements was acceptable to Senior Lender. Upon request, Borrower shall promptly provide copies of all material referenced in this subpart (v) to Lender.
- (w) Earnest Money and Upgrade Deposits. All existing Earnest Money Deposits and Upgrade Deposits, if any, (in the amounts set forth on Exhibit M) shall have been deposited with the Escrow Agent or with Senior Lender in accounts in accordance with Section 14.3. The Earnest Money Deposits are being held by Escrow Agent in accordance with the terms and provisions of each Sales Agreement (such provisions are referred to herein as the "Escrow Agreement"). Within a reasonable time after Closing, Borrower shall deliver to Lender a letter from the Escrow Agent agreeing to deliver, subject to the terms of the Intercreditor Agreement, such Earnest Money Deposits to Lender when and as Borrower has a right to receive such Earnest Money Deposits for application in accordance with Section 14.3 and Section 14.9 below and recognizing that Lender will be entitled, subject to the Intercreditor Agreement, to all right, title and interest of Borrower in and to such Earnest Money Deposits upon foreclosure of Borrower's interest in the Sales Agreements and the Property after an Event of Default. Prior to the Closing, Borrower will provide to Lender evidence satisfactory to Lender that all Earnest Money Deposits for all of the Sales Agreement included within the Pre-sale Requirement have been deposited with, and are currently held by, the Escrow Agent in the amounts shown on Exhibit M attached hereto.
- (x) Required Condominium Approvals. Except with respect to the requirements to be satisfied post Closing as set forth in Section 14.8 below, Borrower represents and warrants that Borrower furnished to Senior Lender evidence satisfactory to Senior Lender that Borrower has received all approvals required for the recordation of the Declaration of Condominium, the sale or marketing of the Residential Units under the requirements of ILSA, applicable laws of the State and any applicable local Laws. Borrower has also furnished to Senior Lender as filed copies of Borrower's HUD Property Report and any additional reports currently required by the State of Texas or City of Austin. Upon request, Borrower will promptly provide copies of the evidence referred to in this subpart (x) to Lender.

- (y) <u>Material Contracts</u>. Borrower shall have provided to Lender the executed Hotel Documents, Non-Disturbance Agreement and estoppel letters addressed to Lender with respect to the Hotel Documents, and all of the foregoing must be in form and substance reasonably satisfactory to Lender. A condition to Lender's obligation to make any advance of proceeds of the Loan, whether at the Closing or thereafter, is that no material default on the part of any party thereto may exist under any Material Contract and each Material Contract (other than, solely with regard to the Closing, the Venue Documents) must be in full force and effect.
- (z) <u>City Estoppel and Agreement</u>. Borrower represents and warrants that prior to closing of the Senior Loan, Borrower (i) provided to Senior Lender an Estoppel Certificate and Agreement from the City of Austin, a Texas home rule city and municipal corporation ("City") in form and substance satisfactory to Senior Lender addressing certain matters relative to the City Documents; and (ii) issued a written directive to the City to pay any repurchase price payable under the City Documents to Senior Lender, which direction was accepted by the City. Prior to Full Loan Opening under the Senior Loan, Borrower shall provide to Senior Lender a Fourth Estoppel Certificate and Agreement from the City in all material respects in the form of Exhibit S or otherwise in form and substance satisfactory to Senior Lender. Upon request, Borrower will promptly provide copies of all materials referenced in this subpart (z) to Lender.
- (aa) Patriot Act: Borrower represents and warrants that (i) prior to closing the Senior Loan, Borrower provided Senior Lender with proof that Borrower complied in all respects with the provisions of the USA PATRIOT Act of 2001, as applicable, including without limitation, furnishing to Senior Lender proof that Borrower has taken all action necessary to comply with Section 326 of such Act; and (ii) Borrower continues to comply with such federal requirements. Upon Lender's request, Borrower shall promptly provide Lender copies of the material referenced in this subpart (aa).
- (bb) Additional Documents. Borrower shall have furnished to Lender such other materials, documents, papers or requirements regarding the Project, Borrower, the Pledgors and/or the Guarantor as Lender shall reasonably request.

ARTICLE 9

CONSTRUCTION REQUIREMENTS PRECEDENT

9.1 Construction Documents Required as of Closing.

Borrower shall cause to be furnished to Lender the following, in form and substance satisfactory to Lender in its reasonable discretion (except as otherwise expressly set forth herein), and Lender shall have approved the following in its reasonable discretion (except as otherwise expressly set forth herein), prior to (and at the time of) Closing (or such later time as is specified in any subparagraph hereof) as additional conditions to Lender's obligations to fund the Loan:

(a) Fully executed copies of a general contract with the General Contractor pertaining to the construction of the Project together with all Change Orders entered into between General Contractor and Borrower as of the date of this Agreement (collectively, the "General Contract"), with a guaranteed maximum price not to exceed \$192,702,489.00 (the "General Contract")

- Price"). Borrower represents and warrants that the following have been delivered to Senior Lender (and Borrower will deliver copies to Lender upon request): (i) those certain fully-executed Subcontracts for the trades and divisions set forth on Exhibit I; (ii) copies of all other direct contracts previously entered into or to be entered into by Borrower for construction, purchase of materials or furniture, fixtures or equipment; and (iii) all contracts with the architects, engineers, third-party owner's representatives and other design professionals (the "Design Professionals"). None of General Contractor and any Design Professionals shall be an Affiliate of Borrower or Guarantor. Lender hereby approves the General Contract and approves Austin Building Company as General Contractor. Borrower represents and warrants that (x) the General Contract was not entered into until after the Senior Deed of Trust was recorded and (y) Borrower provided evidence satisfactory to Senior Lender that the General Contract was not entered into until after the Senior Deed of Trust was recorded.
 - (b) A schedule of values, as included in the General Contract;
- (c) Borrower represents and warrants that Borrower has not received any notice of default from the General Contractor and that no more than two percent (2%) of the total General Contract is "Allowance" (as such term is defined under the AIA A201-1997 General Conditions of the Contract for Construction) (excluding general conditions, General Contractor's fees and the Contractor's Contingency);
- (d) Borrower represents and warrants that in lieu of payment and performance bonds, Senior Lender has accepted subguard insurance with terms, limits, and endorsements (including a Financial Interest Endorsement naming Senior Lender) acceptable to Lender in its reasonable discretion covering all Subcontracts and, upon request, Borrower will provide a Financial Interest Endorsement to be issued to Lender;
- (e) Borrower represents and warrants that (i) copies of each of the Required Permits have been provided to Senior Lender, except for those Required Permits that cannot be issued until a later stage or completion of Construction, in which event such Required Permits will be obtained by Borrower on a timely basis in accordance with all recorded maps and conditions and applicable building, land use, zoning and environmental codes, statutes and regulations and will be delivered to Senior Lender promptly thereafter; and (ii) upon Lender's request, Borrower will promptly provide Lender copies of all Required Permits provided or to be provided to Senior Lender;
- (f) The Plans and Specifications and the Finish Standards described on Exhibit I are each approved by Lender as of the date hereof. As so approved, they are sometimes referred to as the "Approved Plans and Specifications" and the "Approved Finish Standards," respectively. Borrower shall provide Lender with copies of all Change Orders approved by Senior Lender. Borrower shall finish all Residential Units and the Hotel to the Approved Finish Standards as part of the Construction required hereunder;
 - (g) The Construction Schedule;
 - (h) The Soil Report;

- (i) The Environmental Report. Borrower has provided the Environmental Report to Lender, along with a reliance letter from Terracon Consultants, Inc. in form applicable to Lender. Borrower shall also have caused to be furnished to Lender any environmental disclosure statement required pursuant to the law of the State:
- (j) Borrower shall provide Lender copies of all progress Observations and Draw Review Reports prepared by Newbanks, Inc. / Dallas ("Newbanks"), signed by Brett Patton, beginning with the initial monthly report and continuing with all subsequent reports issued as of Closing. Such reports shall contain (i) an analysis demonstrating the adequacy of the Budget to complete the Project and (ii) a confirmation that the Construction Schedule is realistic. Subsequent to Closing, Borrower shall use good faith efforts to cause Newbanks to continue to monitor construction of the Project and shall visit the Project at least one (1) time each month, and shall certify as to amounts of construction costs for all requested fundings, and Borrower shall obtain and provide copies of such certifications to Lender. At any time, Lender shall, at Borrower's sole expense, engage its own consultant to provide the required reports;
 - (k) Each Architect's Certificate;
 - (l) The Budget;
- (m) Original executed consents, in form and substance satisfactory to Lender, of the General Contractor, the Architect, and any other Design Professional to the Collateral Assignment of Construction Documents; and
- (n) Such other papers, materials and documents as Lender may reasonably require with respect to the Construction, the Project, Borrower and/or the Guarantor.

9.2 Construction Deliveries Required as of the Full Loan Opening.

Prior to the Full Loan Opening, in addition to fully satisfying the conditions set forth above in <u>Section 9.1</u>, Borrower shall also provide the following documents and satisfy the following conditions:

- (a) Borrower shall have satisfied Senior Lender's requirement to provide Senior Lender executed contract(s) for all owner direct cost items (except for non-material items) signed by the applicable suppliers of such items, including, without limitation, those for the purchase of the furniture, fixtures and equipment for the Project and copies of such materials and evidence of such approval shall have been provided to Lender;
- (b) Borrower shall have satisfied Senior Lender's requirement that all of the work contemplated by that certain Waste Management Plan dated March 4, 2008 and prepared by Terracon Consultants, Inc., and any amendments or updates thereto (the "Waste Management Plan") contemplated to be performed on or before the Full Loan Opening, has been completed or otherwise addressed to Senior Lender's reasonable satisfaction, such evidence to include, but not be limited to, a written statement from an authorized representative of Terracon Consultants, Inc., affirming that the recommendations of such Waste Management Plan were complied with

and no further actions, other than the ongoing use and maintenance of the water filtration system during the operation of the Project, are required to address the conditions at the Project site that were disclosed in any of Environmental Reports delivered to Senior Lender and Borrower will provide copies of all such materials to Lender:

- (c) Borrower shall have satisfied Senior Lender's requirement that Hotel Operator, and, as may be required by the Hotel Documents, Starwood Hotels & Resorts Worldwide, Inc., shall have approved the Approved Plans and Specifications and Approved Finish Standards and Borrower shall have provided to Lender evidence of such approval, reasonably satisfactory to Lender;
 - (d) To the extent not previously delivered at Closing, the Required Permits, and updates thereof;
- (e) Fully executed Venue Documents and collateral assignment, estoppel and subordination agreements relating thereto, in forms, on terms and with counterparties acceptable to Senior Lender and Borrower shall have provided Lender evidence of such approval, reasonably satisfactory to Lender.

ARTICLE 10

BUDGET, CONTINGENCY FUND AND CHANGE ORDERS

10.1 Budget.

The proceeds of the Loan shall be used only for Costs that are in accordance with the Budget as the same may be amended pursuant to the terms of the Senior Loan Agreement. The Budget shall specify the amount of cash equity invested in the Project, and all costs and expenses of every kind and nature whatsoever to be incurred by Borrower in connection with the Project. The Budget is attached hereto as Exhibit G and made a part hereof.

10.2 Budget Line Items.

- (a) The Budget shall include as line items ("Budget Line Items") the cost of all labor, materials, equipment, fixtures and furnishings needed for the completion of the Construction, and all other costs, fees and expenses relating in any way whatsoever to the Construction of the Improvements, marketing and sales costs, commissions, operating deficits, real estate taxes, and all other sums due in connection with Construction and operation of the Project, the Loan, and this Agreement. The Budget shall not contain any line items payable to Borrower, Guarantor or any Affiliate of either Borrower or Guarantor and Borrower and Guarantor shall not pay or cause to be paid any Loan proceeds to any Affiliate of either, except for the Permitted Affiliate Expenses.
- (b) Borrower shall have the right to reallocate cost savings effected by a final Change Order or other appropriate final documentation to other Budget Line Items in accordance with the terms of the Senior Loan Agreement

10.3 [reserved]

10.4 [reserved]

10.5 Change Orders.

Borrower agrees that no changes will be made in the Approved Plans and Specifications except as in accordance with the Senior Loan Agreement, and Borrower must notify Lender in writing of any such change at least seven (7) days prior to the date such change order will be implemented.

ARTICLE 11

[reserved]

ARTICLE 12

[reserved]

ARTICLE 13

[reserved]

ARTICLE 14

SALE OF RESIDENTIAL UNITS

14.1 Price List Schedule.

Borrower has submitted to Lender the written Price List Schedule (the "Price List Schedule") stating, at a minimum, a detailed breakdown of each condominium Residential Unit's size, type, location and gross sales price ("List Price") for each Residential Unit (including, but not limited to, a weighted average gross sales price (excluding Parking Spaces and Upgrades) of not less than \$620 per Saleable Square Foot for each Residential Unit), which Lender has approved. The approved Price List Schedule is attached to this Agreement as Exhibit M-2. Subject to the terms of the Intercreditor Agreement, no changes to the approved Price List Schedule shall be permitted without Lender's prior written approval, in Lender's reasonable discretion. The List Prices for each Residential Unit include the price of one (1) Parking Space per bedroom in such Residential Unit, up to two (2) Parking Spaces per Residential Unit, except for the penthouse Residential Units on floors 35, 36 and 37, which are each allocated three (3) Parking Spaces but does not include storage space. Of the 300 Parking Spaces allocated to the Residential Units collectively, 275 Parking Spaces shall be included in the List Prices of such Residential Units as set forth above, and 25 Parking Spaces (the "Excess Parking Spaces") shall not be included in the List Price of any Residential Unit and shall be available for sale at an additional charge.

14.2 Sales Agreements.

Each Residential Unit shall be sold under a written agreement (the "Sales Agreement") in the form of the sales agreements attached hereto as Exhibit O, or on a form otherwise approved

by Senior Lender, in all material respects conforming to all Laws, including those requiring disclosures to prospective and actual buyers. Lender has previously approved the form of the Sales Agreement to be used for such Residential Unit sales, which form of sales contract is attached hereto as Exhibit O and has approved the Sales Agreement for each of the pending sales listed on Exhibit M-1. No Residential Unit may be leased, sold or conveyed under any lease, conditional sales contract or other arrangement where Borrower retains a deferred portion of the purchase price or any residual or contingent interest in the Residential Unit, including any purchase money security interest, without the express prior written consent of Senior Lender in each instance. All agreements (each, also a "Sales Agreement") with respect to the sale of any portion of the Project other than the Residential Units shall be subject to the prior written approval of Lender in its reasonable discretion; provided, however, that (i) no portion of the Project other than Residential Units that are sold as permitted by this Agreement, may be sold without the prior written consent of Lender, which consent may be granted or withheld by Lender, in its sole discretion and (ii) Borrower may not enter into any Sales Agreement for any part of the Project, other than Sales Agreements for Residential Units which satisfy the requirements set forth in this Agreement, without the Senior Lender's prior written consent, evidence of which approval shall be promptly provided to Lender. Borrower shall deliver to Lender a copy of each executed Sales Agreement within thirty-one (31) days of its execution. Borrower shall promptly enforce the obligations of the purchasers under each Sales Agreement upon default by any such purchaser, except as Senior Lender may otherwise agree in writing in its reasonable discretion (copies of which will be promptly provided to Lender).

14.3 Purchaser Deposits.

(a) On or before Closing (as to earnest money deposits received prior to Closing), and promptly after receiving any additional earnest money deposits made by the Residential Unit Purchasers from time to time (as to earnest money deposits received following Closing), Borrower shall deposit with Armbrust & Brown, L.L.P. and Heritage Title Company of Austin, Inc., jointly and severally, as escrow agent (or any successor agent approved by Senior Lender) (together, jointly and severally, the "Escrow Agent"), in accordance with the Escrow Agreement, and shall cause Escrow Agent to maintain in one or more accounts titled in the name and subject to the control of Escrow Agent all earnest money deposits paid by Residential Unit Purchasers under the Sales Agreements (the "Earnest Money Deposits").

(b) Except as reflected on Exhibit M, Borrower shall require all Residential Unit Purchasers to make cash Eamest Money Deposits of at least ten percent (10%) of the gross sales price at the time of Sales Agreement execution. If Earnest Money Deposits are held by an Escrow Agent, Borrower shall cause Escrow Agent to deliver to Lender within seven (7) days of the end of each calendar month a statement indicating the amount of funds on deposit representing Earnest Money Deposits, together with information on the date of deposit, and to which Residential Unit all such deposits apply. Borrower shall not accept any non-cash Earnest Money Deposits or non-cash Upgrade Deposits. Borrower shall not be permitted to use or apply the Earnest Money Deposits prior to the closing of the sale of a Residential Unit, whereupon the related Earnest Money Deposit shall be considered part of the Net Sales Proceeds and applied as set forth in Section 14.9.

- (c) If a Residential Unit Purchaser desires to alter the Approved Finish Standards, which results in additional costs to Borrower to complete a Residential Unit (an "Upgrade"), Borrower shall require such Residential Unit Purchaser to deposit funds with Senior Lender in accordance with the terms of the Senior Loan Agreement prior to the commencement of any work relating to such Upgrade (the "Upgrade Deposit").
- (d) The Deposits shall be segregated from other funds and shall be held, applied or returned, as applicable, in accordance with the terms of the respective Sales Agreement and applicable Laws. To the extent Borrower becomes entitled to retain any Deposits that have not been previously applied in accordance herewith (i.e. upon the forfeiture of any deposit by a Residential Unit Purchaser), such amounts shall be paid to Senior Lender and applied to the Senior Loan in accordance with the terms of the Senior Loan Agreement. To the extent that a Residential Unit Purchaser becomes entitled to return of its Deposits under its Sales Agreement or under applicable Laws, so long as the Upgrade Deposit has not already been spent by Borrower, Borrower shall be entitled to withdraw from the escrow account such Deposit for return to the Residential Unit Purchaser in accordance with the terms of the Senior Loan Agreement, with a copy of such notice to be simultaneously sent to Lender if Deposits are held by an Escrow Agent; however, the return of such Deposit shall not affect any obligations of Borrower to Lender or any rights or remedies of Lender.
- (e) Subject to the rights of Senior Lender under the Senior Loan Documents, Borrower hereby grants to Lender a security interest in all of Borrower's right, title and interest in and to the Deposits and all accounts holding any such Deposits from time to time. Upon request by Lender, Borrower shall promptly provide to Lender such documentation as Lender determines is necessary to confirm and perfect such security interest. Such security interest is subject to the rights of Residential Unit Purchasers in and to such Deposits in accordance with the terms of their respective Sales Agreements and applicable Laws

14.4 Residential Unit Sales.

(a) Borrower may enter into Sales Agreements for Residential Units and sell Residential Units only in accordance with the terms of the Senior Loan Agreement. Borrower shall not, without the prior written consent of Senior Lender (a copy of which shall be promptly provided to Lender), enter into any Sales Agreement that is not a Qualifying Sales Agreement, or any other documentation pertaining to the sale of any portion of the Project; provided, however, that Borrower may enter into a Qualifying Sales Agreement subject to a commercially reasonable financing contingency for up to sixty (60) days provided such Qualifying Sales Agreements will not be deemed Qualifying Sales Agreements until the expiration or satisfaction of the financing contingency. Borrower shall not terminate a Qualifying Sales Agreement for any reason other than the default of the Residential Unit Purchaser thereunder without Senior Lender's prior written consent (a copy of which shall be promptly provided to Lender).

14.5 Sales Operations and Seller's Obligations; Amendment and Termination of Sales Agreements.

- (a) Borrower shall at all times maintain adequate marketing capability, and shall timely perform all obligations required to be performed by it under each Sales Agreement.
- (b) Borrower may not amend or modify any Sales Agreement except in accordance with the terms of the Senior Loan Agreement. In such case, Borrower will provide to Lender a copy of the amendment promptly following its execution.
- (c) Borrower may not terminate a Sales Agreement without Senior Lender's prior written consent (a copy of which shall be promptly provided to Lender) except as a result of the default of the Residential Unit Purchaser under such Sales Agreement and provided the Deposits paid under such Sales Agreement are released to Borrower, as the seller, subject to the requirements of the Loan Documents in regard to such Deposits.

14.6 Delivery of Sales Information and Documents.

Within thirty (30) days after the end of each month, Borrower shall deliver to Lender a sales report (the "Sales Report") showing (a) each Residential Unit under a Sales Agreement, including the date of the Sales Agreement, the Residential Unit number, the base contract price plus the charge for Upgrades, the amount of the Earnest Money Deposit and, if applicable, the Upgrade Deposit, and the date expected to close; and (b) for each Residential Unit closed, the Residential Unit number, the base contract price plus the charge for Upgrades, the date closed, and the Net Sales Proceeds from the sale of such Residential Unit; and (c) for each Sales Agreement cancelled by a Residential Unit Purchaser, the Residential Unit number, the date cancelled, the reason the Sales Agreement was cancelled by the Residential Unit Purchaser and a statement regarding whether the Deposit was returned to the Residential Unit Purchaser or retained by Borrower (and remitted to Senior Lender). Borrower shall also promptly deliver to Lender such other sales information and documents that Lender from time to time may reasonably request, including operating statements, all new Sales Agreements, and notice of or information regarding any claimed breach or disayowal of buyer's or seller's obligations under any one or more Sales Agreements.

14.7 Borrower's Acknowledgment Regarding Buyer Financing.

Borrower acknowledges that Lender is not committed to, and does not intend to offer to, provide any financing to or for the buyers of any individual Residential Units or any other part of the Project.

14.8 Condominium Regime.

(a) Borrower shall promptly, and in any event prior to the closing of the first purchase and sale of any Residential Unit, cause the Condominium Documents, including, without limitation, the Master Condominium Declaration and the Residential Condominium Declaration, each in form and containing terms approved by Senior Lender in compliance with

the Senior Loan Agreement. Copies of all documents, correspondence or similar materials related to the Condominium Documents or the condominium regime created thereby which are delivered to or received by Senior Lender or filed with any governmental agency shall be provided to Lender promptly after received, issued, or otherwise available to Borrower.

- (b) Notwithstanding the foregoing or any other provision of this Agreement, Lender shall not be deemed to have represented or warranted that the Condominium Documents comply with Applicable Condominium Laws. Lender shall have no liability for the failure of the Condominium Documents to comply with any Applicable Condominium Law.
 - (c) Borrower represents and warrants that:
 - (i) The sale of Units, as defined in the Declarations, has complied, and at all times will comply, with ILSA and regulations promulgated in connection therewith and Texas Uniform Condominium Act, Tex. Prop. Code, Ch. 82 (2009) ("TUCA").
 - (ii) Borrower has delivered to Senior Lender true, correct and complete copies of all ILSA-related disclosure statements and property reports that, as of the date hereof, have been prepared or delivered in connection with the Condominiums, the Condominium Documents or the Project, together with evidence those items that are, as of the date hereof, required by law to be delivered to prospective purchasers of Residential Units, have been so delivered and, upon request, will promptly provide the same to Lender.
 - (iii) From time-to-time as Residential Units are sold, Borrower shall deliver to Lender evidence those items required by Applicable Condominium Laws to be delivered to prospective purchasers of Residential Units have been so delivered in a timely manner.
 - (iv) There shall be no material agreements relating to the Condominiums entered into after date of this Agreement other than the Condominium Documents unless the same have been approved by Senior Lender or otherwise expressly permitted by the Senior Loan Documents, and copies of such will be promptly provided to Lender.
 - (v) Borrower shall at all times (a) exercise its rights and powers as "Declarant" "Member" or "Owner" (as defined in the Condominium Documents) strictly in accordance with this Loan Agreement, the Condominium Documents and the Applicable Condominium Laws, and (b) take all steps and action necessary to assure that all officers and directors of the Associations appointed or elected by Borrower or any principal or affiliate of Borrower, exercise their rights and powers in accordance with the Applicable Condominium Laws, including the Texas Business Organizations Code, and the terms and provisions of this Agreement.
 - (vi) Not later than the recordation of the Master Condominium Declaration and the Residential Condominium Declaration, Block 21 Hotel/Residential Condominiums ("Residential Condominium") and the Block 21 Master Condominiums ("Master")

Condominium") shall each comply with all of the requirements of TUCA, ILSA, and each other state and federal law applicable to the creation, management, ownership or operation of the Residential Condominium or Master Condominium or sale of units located therein (the "Applicable Condominium Laws"). Borrower shall indemnify and hold Lender harmless from any and all costs, damages, claims, or losses incurred by Lender, including attorneys' fees and court costs, arising from the failure of the Condominium Documents or the Condominiums to comply with Applicable Condominium Laws.

- (d) (i) Not later than the date of recordation of the Master Condominium Declaration or the Residential Condominium Declaration, whichever is first recorded:
 - (x) The Block 21 H/R Condominium Community, Inc., a Texas nonprofit corporation (the "Residential Association"), described in the Residential Condominium Declaration shall have been formed as a nonprofit corporation under the laws of the State of Texas, the board of directors shall have been appointed or elected in accordance with the requirements of the Residential Condominium Declaration and the applicable law and all officers contemplated by the Residential Condominium Declaration and the bylaws of the Residential Association shall have been properly appointed or elected, as the case may be.
 - (y) The Block 21 Master Condominium Community, Inc., a Texas nonprofit corporation (the "Master Association"), described in the Master Condominium Declaration shall have been formed as a nonprofit corporation under the laws of the State of Texas, the board of directors shall have been appointed or elected in accordance with the requirements of the Master Condominium Declaration and the applicable law and all officers contemplated by the Master Condominium Declaration and the bylaws of the Master Association shall have been properly appointed or elected, as the case may be. The Residential Association and Master Association are referred to collectively as the "Associations."
 - (ii) Not later than 15 days after the recordation of the Master Condominium Declaration or the Residential Condominium Declaration, whichever is first recorded, Borrower shall cause the dedicatory instruments with respect to the Condominiums described in Texas Property Code Section 202.006, as the same may be amended or superseded, to be recorded in Travis County, Texas and certified copies of the same delivered to Lender.
 - (iii) Concurrently with the recordation, filing, or adoption, as the case may be, of the Condominium Documents true, correct, and legible copies thereof, including without limitation, a certified copy of each of the recorded Master Condominium Declaration or the Residential Condominium Declaration, the Associations' certificates of formation, bylaws, minute books, other books and records, and any rules and regulations and other Condominium Documents which may have been promulgated, shall

have been delivered to Senior Lender and, where applicable, showing the recordation or filing data. After the date thereof, Borrower shall not amend or supplement or consent to the amendment or supplement of any of the Condominium Documents unless Senior Lender first approves such amendment or supplement in writing. Borrower covenants and agrees to make or cause to be made each revision to any Condominium Document requested by Senior Lender in Senior Lender's reasonable judgment. Copies of all materials referenced in this subpart (iii) will be promptly provided to Lender.

- (e) Concurrently with the recordation of the Declarations, Borrower shall cause to be delivered to Senior Lender and Lender an opinion, reasonably acceptable to Senior Lender, from Armbrust & Brown, LLP or, at Borrower's option, other Texas legal counsel acceptable to Senior Lender, opining that the Condominium Documents are in compliance with the Applicable Condominium Laws.
- (f) Concurrently with the recordation of the Master Condominium Declaration and the Residential Condominium Declaration, Borrower shall deliver true, complete, and legible copies of the following to Senior Lender and Lender in accordance with the Senior Loan Agreement:
 - (i) A current accounting for the Associations' funds and financial statements certified by Borrower and stating unconditionally that the financial statements present fairly the financial position of each Association in conformity with generally accepted accounting principles;
 - (ii) The budgets for the Master Condominium and Residential Condominium in the form required by TUCA, including Section 82.153 (a) 6 and (b), for the calendar year(s) the Master Condominium Declaration and the Residential Condominium Declaration are recorded and the proposed budgets for the calendar year following such recordation;
 - (iii) A current inventory of tangible personal property owned: by the Associations; or by Borrower and used in connection with the ownership or operation of the Condominiums.
 - (iv) A copy of the final as built plans and specifications used for the construction of the improvements in the Condominiums, together with a CAD version of the same;
 - (v) Certified duplicate copies of all insurance policies currently in force, in which the Borrower, the Residential Unit Owners, the Associations, and/or the directors and officers of the Associations are named as insured persons;
 - (vi) Copies of any certificates of occupancy that have been issued with respect to any improvements comprising the Condominiums;

- (vii) Any other permits issued by governmental bodies applicable to the Condominiums and which are currently in force or which were issued within one year prior to the date construction of the Project was commenced, including without limitation, all licenses for the use of public property;
- (viii) Written warranties of the contractor, subcontractors, suppliers, and manufacturers that are still effective related to the Project or any portion thereof or interest therein;
- (ix) A roster of Owners, Eligible Mortgagees, and Mortgagees, each as defined in the Declarations, and their addresses and telephone numbers;
 - (x) Employment contracts in which the Associations are a contracting party;
- (xi) Service contracts in which the Associations are a contracting party or in which the Associations or the Owners have any obligation to pay a fee to the persons performing the services;
- (xii) To the extent not previously delivered to Lender, true, correct and complete copies of all other documents and records the Associations are required to keep pursuant to TUCA § 82.114; and
- (xiii) Copies of all recorded deeds and all recorded and unrecorded leases evidencing ownership or leasehold rights in general common elements within the Condominiums.
- (xiv) During the term of the Loan, to the extent of any material change in any of the foregoing or the Condominium Documents, Borrower shall promptly deliver a copy of the same to Lender. It is the intention of the Borrower and Lender that Lender shall have at all times duplicate copies of all of the foregoing that are in the possession or control of Borrower, including in its capacity as Declarant under the Declarations.
- (g) The addresses of the respective Associations for the purposes of Article 18 of the Master Declaration and Article 16 of the Residential Declaration are:

Block 21 H/R Condominium Community, Inc. c/o Ambrust & Brown, LLP Attn: Bob Burton 100 Congress Avenue, Suite 1300 Austin, Texas 78701

Block 21 Master Condominium Community, Inc. c/o Armbrust & Brown, LLP Attn: Bob Burton 100 Congress Avenue, Suite 1300 Austin, Texas 78701 Borrower may change the addresses set forth above upon thirty (30) days prior written notice to Lender specifying the new address or addresses. On behalf and for the benefit of Lender, concurrently with the recordation of the Master Condominium Declaration and the Residential Condominium Declaration, Borrower shall deliver written notice in accordance with the Declarations to the Associations that Lender is an "Eligible Mortgagee" with respect to the Condominiums. Such notice shall include the name and address of Lender as set forth in this Agreement. Evidence of the delivery of the foregoing shall be delivered to Lender not later than five (5) days after filing of the Certificates of Formation of the Associations or each Association, if filed at different times. If the address of an Association shall change, Borrower shall (i) immediately notify Lender in writing of the new address and (ii) send notice to such Association at such new address that Lender is an "Eligible Mortgagee", as defined in the Condominium Documents, together with such other information as may be required to assure Lender is recognized by each Association as an Eligible Mortgagee. Time is of the essence of the foregoing sentence.

(h) Borrower shall not permit control of any Association to be turned over to the Residential Unit owners more than thirty (30) days prior to the date that is required by Texas law.

14.9 Release of Residential Units.

- (a) At Borrower's request upon or promptly following the closing of a Residential Unit sale, Lender shall issue a partial release of the Residential Unit from the Deed of Trust, so long as all of the following conditions are satisfied at the time of, and with respect to, the partial release:
 - (i) The Master Declaration of Condominium and the Residential Condominium Declaration have each been recorded in the public records of Travis County, Texas;
 - (ii) No Event of Default exists:
 - (iii) Such sale is pursuant to a Qualifying Sales Agreement;
 - (iv) the Release Price (defined below) for the Residential Unit in an amount determined as set forth below has been funded into the Net Sales Proceeds Account under the Senior Loan Agreement (or in the event the Senior Loan has been paid in full, to Lender, to be applied to the Loan):
 - (v) Lender receives from Borrower a written notice of the sale in question (each a "Sales Notice") which will include a copy of the closing statement and applicable Residential Unit release no later than 5:00 p.m. (Dallas time) on the day of such Residential Unit sale; and
 - (vi) All escrow, closing and recording costs have been paid at no expense to Lender.

- (b) The execution of a Sales Agreement shall not by itself satisfy the conditions for release of the Residential Unit that is being sold; those conditions must be satisfied in full at the time the Residential Unit is to be released.
- (c) If Lender does not require satisfaction of all of the conditions described above before releasing one or more Residential Units, that alone shall not be a waiver of such conditions, and Lender reserves the right to require their satisfaction in full before releasing any further Residential Units from the Deed of Trust, subject to the Intercreditor Agreement.
- (d) The "Release Price" for a Residential Unit (including one (1) Parking Space per bedroom in such Residential Unit, up to two (2) Parking Spaces per Residential Unit, except for the penthouse Residential Units, which are each allocated three (3) Parking Spaces) shall be 100% of Net Sales Proceeds from the sale of the applicable Residential Unit and associated Parking Spaces (including any additional charge or fee in excess of the purchase price payable to Borrower or its Affiliates).
- (e) Borrower shall pay all reasonable costs and expenses associated with the sale of the Residential Unit, including title expenses, reasonable legal fees, brokerage and sales commissions and other closing costs, from the portion of the sales price in excess of such Release Price and, if such excess is insufficient for such purpose, shall pay such excess costs from its own funds.
- (f) If the Senior Loan has been paid in full, and if requested by Lender, Borrower shall also deposit with Lender (to the extent not required to be deposited in condominium association bank accounts) all amounts deducted or set aside for real estate taxes or retained for assessments or working capital (and pledges its interest in such account(s) to Lender). Lender shall hold such amount subject to the rights of Residential Unit Purchasers and the condominium association therein, provided, however, Lender shall permit Borrower to use the funds deposited in such accounts for their legally required purposes.

14.10 Application of Sales Proceeds.

Except as otherwise expressly provided herein, any Release Price, or portion thereof, released by Senior Lender to Borrower in accordance with the terms of the Senior Loan Agreement prior to the first anniversary date of the Note will be offered to Lender as a prepayment of the Loan. Lender, in its sole discretion, will determine whether to accept and apply such released proceeds to the Loan or to reject such offered proceeds. Lender will make such determination within five (5) days of offer. Section 4.5 of this Loan Agreement addresses Net Sales proceeds released by Senior Lender on and after the first anniversary date of the Note.

With respect to any Release Price received by Lender in accordance with the preceding paragraph, (i) Lender may elect in its sole discretion to apply Release Price proceeds (to the extent payable to Lender) first to payment of accrued interest (but shall never be obligated to do so, even if doing so would cure a default in the payment of interest) and (ii) if an Event of Default exists, Lender may apply all amounts received to the indebtedness under the Loan Documents in such order as Lender may elect in its sole discretion.

ARTICLE 15

OTHER COVENANTS

15.1 Borrower further covenants and agrees as follows:

(a) Full Opening of Senior Loan on or Prior to Full Loan Opening Date. Borrower shall cause all conditions precedent to the Full Opening of the Senior Loan (as provided in the Senior Loan Agreement) to be complied with, and Borrower shall qualify for Full Loan Opening (as provided in Senior Loan Agreement), no later than July 31, 2010. If Borrower fails to satisfy timely such requirement, and Senior Lender terminates its commitment to fund the Senior Loan, an Event of Default shall exist under this Agreement.

(b) Compliance with Condominium Documents. Once the Declaration of Master Condominium and/or Residential Condominium Declaration are recorded, and subject to the provisions of Section 82.112(b) of the TUCA, Borrower shall pay all general and special assessments for common charges and expenses and insurance premiums made against or relating to the Residential Units owned by Borrower or otherwise payable by Borrower under the Condominium Documents as the same shall become due and payable and prior to delinquency, and not later than the fifteenth (15th) day of each month provide to Lender evidence of such payments, and in the event Borrower shall fail to make such payments as the same become due and payable and prior to delinquency, Lender may from time to time at its option, but without any obligation to do so and without notice or demand upon Borrower, make such payments, and all expenses paid by Lender for such purpose, including, without limitation, attorneys' fees, shall be added to the outstanding principal amount of the Loan and shall be payable on demand and bear interest at the Interest Rate until repaid. Borrower shall not (and shall not permit any Borrower appointed directors to), without the prior written consent of Lender which may be granted or withheld in Lender's sole discretion), give any consent or perform any action in furtherance of any material modification or amendment of the Condominium Documents, including any modifications or amendments to the Condominium Documents which would permit a Residential Unit Purchaser to rescind its Sales Agreement under applicable Laws. Borrower shall comply with all of the material terms, covenants and conditions on its part to be performed under the Condominium Documents, as the same shall be in force and effect from time to time; provided, however, that if Borrower fails to cure such non-compliance within any applicable cure periods provided in the Condominium Documents, Lender may from time to time at its option, but without any obligation to do so, cure or remedy any such default by Borrower (Borrower hereby authorizing Lender, subject to the terms of the Intercreditor Agreement, to enter upon the Project as may be necessary for such purposes), and all reasonable expenses paid by Lender for such purpose, including, without limitation, reasonable attorneys' fees, shall be added to the outstanding principal balance of the Loan and shall be payable on demand and bear interest at the Interest Rate until repaid. Borrower shall deliver to Lender a true and complete copy of each and every notice of default, if any, received by Borrower with respect to Borrower under any of the Condominium Documents or applicable law regarding the condominium regimes created thereby. Borrower shall not (and shall not permit any Borrower appointed director to), without the prior written consent of Lender (but subject to the terms of the Intercreditor Agreement), exercise any right it may have to vote for (x) the expenditure of

insurance proceeds (which are governed by <u>Article 16</u> below) or condemnation awards for the repair or restoration of the Project or (y) any additions or improvements to the common elements of the Project.

- (c) Construction of Improvements. The Improvements shall be constructed and fully equipped in a good and workmanlike manner with materials of high quality, substantially in accordance with the Approved Plans and Specifications (or in accordance with any changes therein that may be approved in writing by Senior Lender). Such construction and equipping shall be commenced and completed, as applicable, and shall be prosecuted with due diligence and continuity in substantial accordance with the Construction Schedule, the requirements of all Sales Agreements and the Operating Agreement. Without limiting the foregoing, Borrower has commenced construction of the Project, and shall substantially complete (i) the entire Project no later than the Completion Date applicable to the entire Project, and (ii) each particular portion of the Project no later than the Completion Date may be extended in accordance with the Senior Loan Agreement.
- (d) Payment for Work. Borrower agrees to fully pay and discharge when due and payable all claims for labor done and material and services furnished in connection with the construction of the Project and to take all other steps to forestall the assertion of claims against the Project or the Loan.
- (e) <u>Inspection by Lender</u>. Borrower shall reasonably cooperate with Lender in arranging for inspections by representatives of Lender of the progress of the Construction from time to time including an examination of (i) the Improvements, (ii) all materials to be used in the Construction, (iii) all plans and shop drawings that are or may be kept at the construction site, (iv) any contracts, bills of sale, statements, receipts or vouchers in connection with the Improvements, (v) all work done, labor performed, and materials furnished in and about the Improvements, (vi) all books, contracts and records with respect to the Improvements, and (vii) any other documents relating to the Improvements or the Construction.
- (f) <u>Materialmen's Liens and Contest Thereof.</u> Borrower shall not suffer or permit any materialmen's lien claims to be filed or otherwise asserted against the Project or any funds due to the General Contractor, and shall promptly discharge the same in case of the filing of any claims for lien or proceedings for the enforcement thereof, provided, however, that Borrower shall have the right to contest in good faith by appropriate legal proceeding and with reasonable diligence the validity of any such lien or claim, provided that Borrower posts a statutory lien bond over such lien.
- (g) Settlement of Materialmen's Lien Claims. If Borrower shall fail promptly either (i) to discharge any such lien, or (ii) to post a statutory lien bond over such lien, in each case, within thirty (30) days after the filing of the lien, Lender may, at its election (but shall not be required to), procure the release and discharge of any such claim and any judgment or decree thereon and, further, may in its sole discretion effect any settlement or compromise of the same, or may furnish such security or indemnity to the Title Insurer, and any amounts so expended by Lender, including premiums paid or security furnished in connection with the issuance of any

surety company bonds, shall be deemed to constitute disbursement of the proceeds of the Loan hereunder. In settling, compromising or discharging any claims for lien, Lender shall not be required to inquire into the validity or amount of any such claim.

- (h) Proceedings. In addition to (and not in lieu of Borrower's covenants set forth in Section 15.1(f)), if any action, claim or proceeding affecting title to the Land, the construction of the Project, Borrower, Guarantor, the rights of Lender under any Loan Document, or otherwise affecting the Project, be filed or commenced, then at the request of Lender, Borrower shall appear in and defend, at Borrower's sole cost and expense, any such action or proceeding and, if applicable, Borrower shall insure or bond over such action, claim or proceeding in accordance with Section 15.1(f). If Borrower fails to appear and defend any such action or proceeding, then Lender may commence, intervene in, and defend actions or proceedings affecting the Project or the transactions contemplated herein, and compromise or settle any claim or controversy pertaining thereto, employing legal counsel acceptable to Lender to defend such claims at Borrower's sole cost, unless due to the willful misconduct of Lender. Lender shall not be liable to Borrower for any action, error, mistake, omission or delay pertaining to the actions described in this Section or any damages resulting therefrom, unless due to the willful misconduct of Lender. Any cost incurred by Lender under this Section shall be deemed to be expenses of the Loan payable by Borrower pursuant to Article 7 of this Agreement.
- (i) Insurance. Borrower shall cause insurance policies to be maintained in compliance with Exhibit E, or such other insurance requirements as may be reasonably required by Lender, at all times (provided Lender will not require any insurance coverages that are not required by Senior Lender as long as the Senior Loan remains outstanding). Borrower shall provide Lender, prior to the date hereof, a certificate(s) demonstrating appropriate insurance coverage(s), which shall demonstrate insurance coverage that meets or exceeds the requirements on Exhibit E. Borrower shall timely pay all premiums on all insurance policies required hereunder, and as and when additional insurance is required, from time to time, during the progress of Construction, and as and when any policies of insurance may expire, furnish to Lender, premiums prepaid, additional and renewal insurance policies with companies, coverage and in amounts reasonably satisfactory to Lender in accordance with Exhibit E.
- (j) Payment of Taxes. Borrower shall pay all real estate taxes and assessments and charges of every kind upon the Project before the same become delinquent. Borrower may use Loan proceeds for such purpose to the extent available in the Budget for such purpose upon meeting all conditions precedent set forth in this Agreement to any such disbursement. Borrower shall have the right to pay such tax under protest or to otherwise contest any such tax or assessment, but only if (i) such contest has the effect of preventing the collection of such taxes so contested and also of preventing the sale or forfeiture of the Project or any part thereof or any interest therein, (ii) Borrower has notified Lender of Borrower's intent to contest such taxes, and (iii) Borrower has deposited security in form and amount reasonably satisfactory to Senior Lender or Lender, in its reasonable discretion, and has increased the amount of such security so deposited promptly after Senior Lender's request therefor. If Borrower fails to commence such contest or, having commenced to contest the same, and having deposited such security required by Lender for its full amount, shall thereafter fail to prosecute such contest in

good faith or with due diligence, or, upon adverse conclusion of any such contest, shall fail to pay such tax, assessment or charge, Lender may, at its election (but shall not be required to), pay and discharge any such tax, assessment or charge, and any interest or penalty thereon, and any amounts so expended by Lender shall be deemed to constitute disbursements of the Loan proceeds hereunder (even if the total amount of disbursements would exceed the face amount of the Note). Borrower shall furnish to Lender evidence that taxes are paid at least five (5) days prior to the last date for payment of such taxes and before imposition of any penalty or accrual of interest.

- (k) Tax Escrow Accounts. After an Event of Default and, subject to the terms of the Intercreditor Agreement, if Lender elects to require that Borrower escrow money for payment of taxes (a "Triggering Event"), Borrower shall make monthly tax escrow deposits in the amount of one-twelfth (1/12) of the annual real estate taxes as reasonably estimated by Lender, such deposit to be held in an interest bearing escrow account held by Senior Lender in Senior Lender's name and under its sole dominion and control, unless at such time the Senior Loan has been paid in full, in which case the deposits will be held by Lender. (If an Event of Default occurs and Lender requires real estate tax escrow deposits to be made, but such Event of Default is thereafter cured, then, so long as no other Triggering Event exists, Lender shall not require further real estate tax deposits to be made.) If at any time Lender determines in its sole discretion that the amount of the monthly escrow payments made pursuant to this Section 15.1(k) are not sufficient to pay in full the next installment of real estate taxes then due, then upon written notice from Lender of the amount of any expected deficiency (and regardless of whether a Triggering Event then exists), Borrower shall then deposit funds equal to such amount with Lender. All payments deposited in the escrow account, and all interest accruing thereon, are pledged as additional collateral for the Loan. Notwithstanding Lender's holding of the escrow account, nothing herein shall obligate Lender to pay any real property taxes with respect to any portion of the Project at any time an Event of Default exists.
- (1) Personal Property. All of Borrower's personal property, fixtures, attachments and equipment delivered upon, attached to or used in connection with the Construction or the operation of the Project shall always be located at the Project and shall be kept free and clear of all liens, encumbrances and security interests, other than as otherwise permitted under the Loan Documents. Borrower shall not acquire by lease any of the personal property, fixtures, attachments or equipment which is to be used in connection with the Project without Lender's prior written consent, which consent will not be unreasonably withheld with regard to leases of non-essential, movable equipment utilized in the operation of the Project, but may otherwise be granted or withheld in Lender's sole discretion. Borrower has notified Lender that Borrower does not currently intend to lease any such personal property and that the Budget includes funds necessary to purchase all such personal property.

(m) Leasing Restrictions.

(i) Borrower shall not enter into any Leases pertaining to the Project without Lender's prior written consent in its sole discretion; provided, however, Lender shall be deemed to have consented to such Lease if Senior Lender consents to the Lease.

Borrower shall provide Lender with a copy of any proposed Lease no less than ten (10) Business Days prior to the proposed execution date of such Lease. Lender's approval of any Lease for the Commercial Space shall be predicated upon, among other things: (a) current tenant financial information in sufficient detail to assess the experience and credit worthiness of the proposed tenant, (b) credit worthiness of the proposed tenant, (c) form and content of the Lease, including, among other things, the proposed tenant's obligation to provide, at least annually, its financial information to Borrower, (d) the proposed tenant's agreement to enter into an acceptable subordination, non-disturbance and attormment agreement, and (e) a minimum (7) seven year term, without any right to cancel prior to five (5) years; provided, however, Lender shall approve a five (5) year Lease term, but the tenant improvement funding permitted for such Lease shall be reduced proportionately. Except as otherwise consented to by Lender, each Lease must provide for a minimum annual triple net rent per Rentable Square Foot as specified in Exhibit Q at all times throughout the term of the related Lease, and, for purposes of calculating such rent, the total value of free rent, non-standard lease concessions and tenant improvements in excess of the budgeted amount will be amortized at a per annum rate of ten percent (10%) over the full term of the Lease, including any free rent period. The resulting amount will be deducted from the lowest stated rent at any point in the term of the Lease, and that figure must be equal to or greater than the applicable minimum rent as specified in Exhibit Q. Furthermore, Borrower will in no event offer more than six (6) months of free rent, tenant improvements in excess of twenty percent (20%) more than the budgeted amount, or other non-standard concessions unacceptable to the Lender. Each Lease must otherwise satisfy the applicable requirements set forth on Exhibit Q (including, without limitation, the amount of l

(ii) Borrower shall not execute any Lease without written acknowledgement of Lease approval by the Lender, not to be unreasonably withheld, except that Leases of Office Space proposed to be entered into with third party tenants who are not Affiliates of Borrower or Guarantor covering not more than 1,500 Rentable Square Feet individually, and 8,000 Rentable Square Feet in the aggregate, and otherwise satisfying the leasing requirements set forth above may be entered into without Lender's consent. Lender shall endeavor to provide (i) preliminary approval or disapproval of any proposed Lease within five (5) Business Days of the receipt of all pertinent information needed for approval, and (ii) final approval or disapproval of any proposed lease transaction within ten (10) Business Days of the receipt of final proposed lease documentation and any other information needed for its decision, and Lender's failure to approve a Lease within such ten (10) Business Day period shall be deemed its disapproval thereof. Notwithstanding the above, if Senior Lender has consented to any Lease, Lender's consent and approval shall not be required.

(iii) A Lease, which has been consented to by the Lender (or as to which Lender's consent is not required under the terms of clause (ii)), is hereinafter referred to

as an "Approved Lease." An Approved Lease shall not be amended or modified (in any material respect) or terminated without the Lender's prior written consent.

(iv) Borrower shall not accept any rental payment under any Approved Lease in advance of its due date, other than acceptance of a prepayment of the first month's rent upon the execution of an Approved Lease.

(n) Reserved.

(o) Lender's Attomeys' Fees for Enforcement of Agreement. In case of any Event of Default or material Default hereunder, Borrower (in addition to Lender's attorneys' fees, if any, to be paid pursuant to Section 7.5) will pay Lender's reasonable attorneys' and paralegal fees (including, without limitation, any attorney and paralegal fees and costs incurred in connection with such Event of Default or material Default, including, without limitation, any litigation or bankruptcy or administrative hearing and any appeals therefrom and any post-judgment enforcement action including, without limitation, supplementary proceedings) in connection with the enforcement of this Agreement; without limiting the generality of the foregoing, if at any time or times hereafter Lender employs counsel (whether or not any suit has been or shall be filed and whether or not other legal proceedings have been or shall be instituted) for advice or other representation with respect to the Project, this Agreement, or any of the other Loan Documents, or to protect, collect, lease, sell, take possession of, or liquidate any of the Project, or to attempt to enforce any security interest or lien in any portion of the Project, or to enforce any rights of Lender or Borrower's obligations hereunder, then in any of such events all of the reasonable attorneys' fees arising from such services, and any expenses, costs and charges relating thereto (including fees and costs of paralegals), shall constitute an additional liability owing by Borrower to Lender, payable on demand. Such attorneys' fees and expenses shall include fees and expenses of Lender's in-house counsel as specified in Section 7.5.

(p) <u>Appraisals</u>. Lender shall have the right to obtain a new or updated Appraisal of the Project at any time and from time to time, at Lender's cost; provided however, Borrower shall cooperate with Lender in this regard. If a material Default or an Event of Default exists, Borrower shall pay the reasonable costs for any such Appraisal upon Lender's request.

(q) Financial Statements and Reports.

(i) Borrower and Guarantor shall deliver or cause to be delivered to Lender annual financial statements with respect to Borrower and Guarantor within ninety (90) days after the end of its fiscal year. Guarantor's financial statements shall be audited and Borrower's financial statements shall be certified by Borrower's managing member. Guarantor and, or Borrower shall also each deliver to Lender quarterly financial statements within sixty (60) days after the end of each fiscal quarter. Each such quarterly financial statement shall be accompanied by a certification by an authorized financial officer of Guarantor as to such Guarantor's compliance with the Guarantor Financial Covenants and attaching information sufficient to demonstrate such compliance. All such financial statements shall be in a format approved in writing by Lender in Lender's

reasonable discretion and in substance acceptable to Lender. Each financial statement shall be certified as true, complete and correct by its preparer and by Borrower or, in the case of Guarantor's financial statements by the Guarantor to whom it relates. Financial statements of Guarantor shall include verifications, supporting schedules and additional statements as needed to substantiate the information contained in such statements. In addition, such financial statements shall include disclosure of any pending or threatened litigation and judgments entered against Borrower or Guarantor. Borrower and Guarantor shall deliver to Lender Borrower's, Guarantor's, and each entity in which a Guarantor has a material interest, federal and state tax returns by April 15 of each year (except that if the date on which such returns may be filed is extended beyond April 15, then such entity shall provide Lender with a copy of the extension request and shall furnish such tax returns to Lender within thirty (30) days after the date such returns are filed). Borrower shall inform Lender as to any filed or threatened (in writing) litigation which would have a material adverse effect on Borrower or Guarantor's ability to perform their respective obligations under the Loan Documents promptly after learning thereof. Within thirty (30) days following the end of each month during the term of the Loan, Borrower will provide to Lender, unaudited, certified (by Borrower) operating statements of the Project, and leasing reports relating to the Project, in such form as Lender may reasonably requires. Borrower shall during regular business hours permit Lender or any of its agents or representatives to have access to and examine all of its books and records regarding the development and operation of the Project and, in addition, agrees to provide Lender with copies of any purchase contracts pertaining to the Project. Borrower agrees that Lender may retain an investigator to research available public records and information relating to Borrow

- (ii) Borrower shall furnish to Lender within three (3) Business Days of receipt from time to time each proposed budget and "Operating Plan" received from Hotel Operator.
- (iii) Borrower shall, within five (5) Business Days of receipt from time to time furnish to Lender (1) all "Operating Reports" received under Section 10.2 of the Hotel Operating Agreement, (2) all financial statements received under Section 10.3 of the Hotel Operating Agreement, (3) all other material information or notices received from Hotel Operator, and, (4) all material notices, reports and other information received by or issued by or on behalf of Borrower under any Venue Document or any other Material Contract; and
- (iv) Borrower shall provide to Lender on or before the 15th day of each month, a certified (by Borrower) statement, in form acceptable to Lender, specifying amounts paid during the preceding month from the Equity Investment Loan proceeds or proceeds of the Senior Loan to pay costs set forth on the Budget Line Items and a reconciliation of the unfunded Equity Investment unused Loan proceeds and/or unfunded Senior Loan proceeds and unpaid Budget Line Items.

(r) Reserved.

- (s) Lost Note. Upon Lender's furnishing to Borrower an affidavit to such effect, Borrower shall, if the Note is mutilated, destroyed, lost or stolen, deliver to Lender, in substitution therefor, a new note containing the same terms and conditions as the Note
- (t) Indemnification. Borrower shall indemnify Lender, including each party owning an interest in the Loan and their respective officers, directors, employees and consultants (each, an "Indemnified Party") and defend and hold each Indemnified Party harmless from and against all claims, injury, damage, loss, liability, cost and/or expense (including reasonable attorneys' fees, costs and expenses) of any and every kind to any persons or property by reason of (i) the Construction; (ii) the sale, operation or maintenance of the Project; (iii) any claim with respect to application, disposition or return of any Eamest Money Deposit or Upgrade Deposit, (iv) any breach of representation or warranty, Default or Event of Default under this Agreement or any other Loan Document or related Document; or (v) any other matter arising in connection with the Loan, Borrower, Guarantor, any Residential Unit Purchaser or Sales Agreement, or the Project. No Indemnified Party shall be entitled to be indemnified against its own gross negligence or willful misconduct. The foregoing indemnification shall survive repayment of the Loan.
- (u) No Additional Debt. Except for the Senior Loan and the Loan, Borrower shall not incur or guarantee any indebtedness (whether personal or nonrecourse, secured or unsecured) other than customary trade payables paid within ninety (90) days after they are incurred. Except for the liens securing the Senior Loan and the Loan and except for the Permitted Exceptions, Borrower shall keep the Project free and clear of liens, provided, however, mechanics' liens may be contested in compliance with Section 15.1(f).
- (v) <u>Compliance With Laws</u>. Borrower shall comply with all applicable requirements (including applicable Laws) of any Governmental Authority having jurisdiction over Borrower or the Project.
- (w) Organizational Documents. Borrower shall not, without the prior written consent of Lender, permit or suffer (i) a material amendment or modification of its organizational documents (however, Lender's consent will not be unreasonably withheld with regard to any such amendment or modification to provide for the admission of a new member, partner or shareholder as permitted by Section 17.2 of this Agreement), (ii) the admission of any new member, partner or shareholder in violation of Section 17.2 of this Agreement, or (iii) any dissolution or termination of its existence.
- (x) <u>Furnishing Reports</u>. Upon Lender's request, Borrower shall provide Lender with copies of all inspections, reports, test results and other information received by any Borrower, that in any way relate to the Project or any part thereof.
- (y) Management Contracts. Borrower shall not enter into, modify, amend, terminate or cancel any Material Contract or any other management, sales or marketing contracts for the Project, without the prior written approval of Lender, not to be unreasonably withheld or delayed.

- (z) <u>Furnishing Notices</u>. Borrower shall provide Lender with copies of all material notices pertaining to the Project received by Borrower from any Governmental Authority, any counterparty under any Material Contract or insurance company within seven (7) days after such notice is received. Borrower shall promptly notify Lender of any judgment entered against, or any material litigation filed against, Borrower or Guarantor.
- (aa) Construction Contracts. Borrower shall not enter into, materially modify or materially amend, or terminate or cancel the General Contract, any Subcontract or any other material contracts for the Construction, without the prior written approval of Senior Lender. Borrower shall not enter into any contract which would cause the Senior Loan to cease to be In Balance. Borrower will furnish Lender promptly after execution thereof, executed copies of all contracts between Borrower, architects, engineers and contractors and all Subcontracts between the General Contractor or contractors and all of their subcontractors and suppliers, which contracts and Subcontracts may not have been furnished pursuant to Section 9.1(a) at Closing. The development agreement between the Borrower and Stratus Block 21 Investments, L.P. may be modified without Lender's consent so long as such modification does not materially adversely affect Lender and may be terminated upon satisfaction of the same conditions as are applicable to CJUF's replacement of Stratus Block 21 Investments, L.P. as the managing member of Borrower under Section 17.2 below.
- (bb) <u>Correction of Defects</u>. Within five (5) days after Borrower acquires knowledge of or receives notice of a material defect in the Improvements or any material departure from the Approved Plans and Specifications, or any other requirement of this Agreement, Borrower shall notify Lender in writing and proceed with diligence to correct all such defects and departures.

(cc) Reserved.

- (dd) Food and Liquor Licenses. Borrower does not currently believe it will, or will need to, obtain any food or liquor licenses in connection with the Project. Instead, all such licenses will be obtained by the Hotel Operator and the lessee(s) of the Restaurant Space.
- (ee) <u>Alterations</u>. Without the prior written consent of Senior Lender, Borrower shall not make any material alterations to the Project (other than completion of the Construction in accordance with the Approved Plans and Specifications).
- (ff) <u>Cash Distributions</u>. Borrower shall not make any distributions to its partners or other equity holders, Guarantor or any Affiliate of Borrower or Guarantor until the Loan has been repaid, except as permitted under an Senior Loan Agreement.
- (gg) Injunctive Proceedings. If any proceedings are filed seeking to enjoin or otherwise prevent or declare invalid or unlawful the construction, sale, occupancy, maintenance or operation of the Project, including, without limitation, the Residential Units, Borrower shall cause such proceedings to be diligently contested in good faith, and in the event of an adverse ruling or decision, shall prosecute all allowable appeals therefrom, and shall, without limiting the generality of the foregoing, resist the entry or seek the stay of any temporary or permanent injunction that may be entered and use commercially reasonable efforts to bring about a

favorable and speedy disposition of all such proceedings. Lender's reasonable costs and disbursements (including attorney's fees) in connection with any such proceedings, whether or not Lender is a party thereto, shall be deemed to be expenses of the Loan payable by Borrower in accordance with Article 7 of this Agreement.

(hh) Reserved.

- (ii) Operating Revenues. In the event the Project produces gross operating revenues, Borrower shall establish with Senior Lender or Beal Bank, an interest bearing operating account for the Project (the "Project Operating Account") into which Borrower shall deposit, on the Business Day, received or, if received on a day which is not a Business Day, the first Business Day following the day of receipt, all revenue associated with the Project (including any Net Operating Income distributed by Hotel Operator to the Borrower), and from which Borrower shall make withdrawals to pay all operating expenses (other than interest on the Loan) of the Project and, to the extent of any remaining Net Operating Income, payments on the Senior Loan as set out in the Senior Loan Agreement. Effective upon the payment in full of the Senior Loan and the release of Senior Lender's lien on the Project Operating Account, Borrower hereby pledges to Lender, (A) the Project Operating Account and all funds on deposit therein, and, (B) to the extent permitted by, and subject to the limits of, the Hotel Operating Agreement and the Non-Disturbance Agreement, all other accounts to be established in accordance with the terms and provisions of the Hotel Operating Agreement and eposit or payment in the Project Operating account has been established substantially similar in form and substance to the reasonable form of same prepared by Senior Lender's legal counsel to the extent a pledge is permitted under the Hotel Operating Agreement and the Non-Disturbance Agreement. Upon payment in full of the Senior Loan and the release of all of Senior Lender's liens and security interests against the Property, Borrower shall deliver, or shall cause to be delivered, to Lender, a control agreement from each depository bank with which the Project Operating Account and each such Hotel operating account has been established substantially similar in form and substance to the reasonable form of same prepared by Senior Lender's legal counsel to the extent a pledge is
- (jj) Environmental Remediation. Borrower has performed all recommended remediation set forth in those certain Environmental Reports dated October 24, 2007, May 3, 2007, October 18, 2006, and October 9, 2009, prepared by Terracon Consultants, Inc. in each case, by the dates such remediation is recommended to be completed in such Environmental Report, except in the case of the Waste Management Plan, the water filtration system discussed in the Waste Management System shall continue to be operated and maintained in accordance with the Waste Management Plan during the construction and operation of the Project.
- (kk) <u>Hotel Documents</u>. Borrower shall timely perform and comply with all of its obligations under the Hotel Documents and all other agreements with Hotel Operator, Starwood Hotels & Resorts Worldwide, Inc., and their affiliates with respect to the Hotel, and shall keep all

such agreements in full force and effect. Borrower shall not modify, amend or terminate any such agreements without Lender's prior written consent, to be given in its reasonable discretion.

(ll) <u>Venue Documents</u>. Once fully executed, Borrower shall timely perform and comply with all of its obligations under the Venue Documents, shall keep all such Venue Documents in full force and effect. Once fully executed, Borrower shall not modify, amend or terminate any such Venue Document without Lender's prior written consent, to be given in its reasonable discretion.

(mm) Pre-sale Requirement. The Pre-sale Requirement must remain satisfied throughout the Loan Term.

15.2 Single Purpose Entity Covenants.

Borrower hereby represents, warrants and covenants that without Lender's prior written consent, which may be withheld in Lender's sole discretion, and except as otherwise expressly permitted hereunder, Borrower has not, will not and shall not:

- (a) engage in any business or activity other than the ownership, management, construction and operation of the Project;
- (b) acquire or own any material assets other than the Project, and such incidental personal property as may be necessary for the operation of the Project;
- (c) merge into or consolidate with any Person or entity or dissolve, terminate or liquidate in whole or in part, transfer or otherwise dispose of all or substantially all of its assets or change its legal structure or acquire by purchase or otherwise all or substantially all the business or assets of, or stock or other evidence of beneficial ownership of, any Person, without in each case the prior written consent of Lender, which consent may be withheld or delayed in Lender's sole and absolute discretion;
- (d) fail to preserve its (i) existence as an entity duly organized, validly existing and in good standing under the laws of the State of Delaware or (ii) qualification to do business in the State, or without the prior written consent of Lender amend, modify, terminate or fail to comply with the provisions of Borrower's formation documents, as same may be further amended or supplemented, if such amendment, modification, termination or failure to comply would adversely affect the ability of Borrower to perform its obligations under the applicable Loan Documents or jeopardize Borrower's existence as a single purpose entity;
 - (e) own any subsidiary or make any investment in, any Person without the consent of Lender;
 - (f) commingle its assets with the assets of any of its Affiliates, or of any other Person;

- (g) incur any debt, secured or unsecured, direct or contingent (including guaranteeing any obligation), other than other indebtedness, if any, expressly permitted under this Agreement, provided that any such debt is paid before such payments are past due;
- (h) become insolvent and fail to pay its debts, liabilities and obligations of any kind, including all administrative expenses, from its own separate assets as the same shall become due:
- (i) fail to maintain its records, books or accounts and bank accounts separate and apart from those of any Affiliate of Borrower, any Affiliate of a partner of Borrower and any other Person or entity;
- (j) enter into any contract or agreement with any of its Affiliates except on terms and conditions that are intrinsically fair and substantially similar to those that would be available on an arms-length basis with third parties other than an Affiliate or as provided in this Agreement;
 - (k) seek the dissolution or winding up in whole, or in part, of Borrower;
- (l) maintain its assets in such manner that it will be costly or difficult to segregate, ascertain or identify its individual assets from those of any partner, member, constituent, principal or Affiliate of Borrower, or any member, general partner, principal or Affiliate thereof, or any other person;
 - (m) hold itself out to be responsible for the debts of another Person except as specifically permitted in this Agreement;
- (n) make any loans or advances to any third party, including any partner, member, constituent, principal or Affiliate of Borrower or any member, general partner, principal or Affiliate thereof;
- (o) fail to file its own tax returns; provided, however, that Borrower may file a consolidated tax return with any of its Affiliates, but only because such consolidated tax return is required by applicable Laws, and provided such consolidated tax return includes footnotes identifying the separate assets or liabilities of Borrower and/or its Affiliates, as applicable;
- (p) fail either to hold itself out to the public as a legal entity separate and distinct from any other Person or to conduct its business solely in its own name in order not (a) to mislead others as to the identity with which such other party is transacting business, or (b) to suggest that it is responsible for the debts of any third party (including any partner, principal, member of Affiliate or Borrower, or any partner, principal, member or Affiliate thereof) except as specifically permitted in this Agreement;
- (q) fail to maintain adequate capital for the normal obligations reasonably foreseeable in a business of its size and character and in light of its contemplated business operations;

- (r) file or consent to the filing of any petition, either voluntary or involuntary, to take advantage of any applicable insolvency, bankruptcy, liquidation or reorganization statute, or make an assignment for the benefit of creditors;
- (s) fail to maintain a reasonably sufficient number of employees in light of its contemplated business operations if any employees are required for the contemplated business operations:
- (t) fail to allocate fairly and reasonably any overhead expenses that are shared with any of their respective partners, members, principals or Affiliates of Borrower, any guarantor or indemnitor, or any partner, member, principal or Affiliate thereof, including paying for office space and services performed by any employee of any of their respective partners, of any members, principals or Affiliates, any guarantor or indemnitor, or any partner, member, principal or Affiliate of any thereof;
- (u) except to the extent required by generally accepted accounting principles or applicable law, fail to maintain separate financial statements, which shows its assets and liabilities, separate and apart from those of any other Person or entity and not have its assets listed in the financial statement of any other entity; or
 - (v) fail to correct any known misunderstanding regarding its separate identity.

15.3 Authorized Representative.

Borrower hereby appoints the Authorized Representative as its authorized representative for purposes of dealing with Lender on behalf of Borrower in respect of any and all matters in connection with this Agreement, the other Loan Documents, and the Loan. The Authorized Representative shall have the power, in his discretion, to give and receive all notices, monies, approvals, and other documents and instruments, and to take any other action on behalf of Borrower. All actions by the Authorized Representative shall be final and binding on Borrower. Lender may rely on the authority given to the Authorized Representative until actual receipt by Lender of a duly authorized resolution substituting a different person as the Authorized Representative. If Borrower appoints more than one Authorized Representative, the action of any one Authorized Representative shall be binding and sufficient.

15.4 Negative Covenants.

Borrower hereby covenants and agrees that it shall not, without Lender's prior written consent, which consent may be withheld in Lender's sole and absolute discretion:

- (a) transfer, convey or assign the Property or any part thereof (other than sales of Residential Units in accordance with Article 14), to any Person, including, but not limited to a transfer in lieu of foreclosure to Senior Lender;
- (b) permit any of its members, or constituent members, partners or shareholders thereof, to voluntarily transfer, convey or assign any of their respective direct or indirect equity

interests in Borrower to Senior Lender, other than pursuant to a foreclosure of Senior Lender's liens on such member interests under the Senior Loan Documents: or

(c) request, consent or agree to an increase in Senior Lender's loan commitment pursuant to the Senior Loan Agreement in an amount that exceeds \$120,000,000, less all prepayments of the Senior Loan received by Senior Lender, or to a conversion of the Senior Loan to a revolving credit facility.

ARTICLE 16

CASUALTIES AND CONDEMNATION

16.1 Lender's Election to Apply Proceeds on Indebtedness.

- (a) Subject to the provisions of the Intercreditor Agreement and to the provisions of Section 16.1(b) below, Lender may elect to negotiate, settle, collect, retain and apply upon the indebtedness of Borrower under this Agreement or any of the other Loan Documents all proceeds of insurance or condemnation (individually and collectively referred to as "Proceeds") after deduction of all expenses of collection and settlement, including reasonable attorneys' and adjusters' fees and charges. After the Senior Loan has been paid in full, Lender shall have the right to participate with Borrower in negotiation of any settlement, adjustment or compromise of any claim arising in connection with a casualty to the Improvements or any condemnation of all or part of the Project; provided, however, if an Event of Default exists, Lender shall have the right to settle any claim without Borrower's participation or consent. Subject to the provisions of the Intercreditor Agreement, any Proceeds remaining after repayment of the indebtedness under the Loan Documents shall be paid by Lender to Borrower.
- (b) Notwithstanding anything in Section 16.1(a) to the contrary, and subject to the provision of the Intercreditor Agreement in the event of any casualty to the Improvements or any condemnation of part of the Project, Lender agrees to make available the Proceeds for restoration of the Improvements if and as required by the Non-Disturbance Agreement or if (i) no Event of Default or material Default exists, (ii) all Proceeds are deposited with Lender, (iii) in Lender's reasonable judgment, the amount of Proceeds available for restoration of the Improvements (together with undisbursed proceeds of the Loan, if any, allocated for the cost of the Construction and any sums deposited with Lender by Borrower for such purpose) is sufficient to pay the full and complete costs of such restoration, (iv) the cost of restoration does not exceed twenty percent (20%) of the stated amount of the Note, (v) if Lender determines that the cost of restoration exceeds \$5,000,000, Lender determines in its reasonable discretion that the values set forth in the Appraisal reviewed and approved by Lender at Closing remain valid; (vi) in Lender's reasonable determination, the Project can be restored to an architecturally and economically viable project in compliance with applicable Laws, (vii) Guarantor reaffirms the Guaranty in writing, (viii) Borrower shall have provided evidence reasonably acceptable to Lender that following restoration (and completion of the Project) the Pre-Sale Requirement will remain satisfied, (ix) the Hotel Documents, and the Venue Documents, will each remain in full force and effect; and (x) in Lender's reasonable determination, such restoration is likely to be completed so that the Residential Units sufficient to satisfy the Pre-Sale Requirement may be delivered to all

Residential Unit Purchasers prior to the outside delivery dates contained in their respective Sales Agreements and in any event not later than the Maturity Date. On and after such time as the condominium is formed, Borrower agrees to vote its votes with respect to all unsold Residential Units (and to cause all Borrower appointed directors to vote their votes) in a manner consistent with the provisions of this Article 16.

16.2 Borrower's Obligation to Rebuild and Use of Proceeds Therefor.

In case Lender does not elect to apply or does not have the right to apply the Proceeds to the indebtedness, as provided in Section 16.1 above, Borrower shall:

- (a) Proceed with diligence to make settlement with insurers or the appropriate Governmental Authorities and cause the Proceeds to be deposited with Senior Lender;
- (b) In the event the Proceeds and the available proceeds of the Loan are insufficient to assure Lender that the Senior Loan will be In Balance, promptly deposit with Senior Lender any amount necessary to place the Senior Loan In Balance; and
- (c) Promptly proceed with the assumption of construction of the Improvements, including the repair of all damage resulting from such fire, condemnation or other cause and restoration to its former condition.

ARTICLE 17

ASSIGNMENTS BY LENDER AND BORROWER

17.1 Assignments and Participations.

Lender may from time to time, with written notice to Borrower, sell all or any part of the Loan and the Loan Documents (or any interest therein) and may grant participations in the Loan (i) if and to the extent required by applicable regulatory authority and (ii) to any Affiliate of Lender. Except as provided in the preceding sentence, Lender will not sell all or any part of the Loan or grant participation in the Loan without Borrower's prior written consent. Borrower agrees to reasonably cooperate with Lender's efforts to do any of the foregoing and to execute all documents reasonably required by Lender in connection therewith that do not materially adversely affect Borrower's or Guarantor's rights under the Loan Documents or materially increase Borrower's or Guarantor's obligations under the Loan Documents. No such assignment or participation will release Lender from any liability under the Loan Documents.

17.2 Prohibition of Assignments and Transfers by Borrower.

Borrower shall not assign or attempt to assign its rights under this Agreement and any purported assignment shall be void. Without the prior written consent of Lender, in Lender's sole discretion, Borrower shall not suffer or permit any change in the ownership, or management or economic interests (whether direct or indirect) of the Project, or any Transfer. Notwithstanding the foregoing, so long as all owners of interests in Borrower satisfy the requirements of Section 3.1(y) above (a) so long as CJUF continues to hold the ownership

interest in Borrower which CJUF held at Closing and no monetary Event of Default then exists, or if a monetary Event of Default does exist and is cured by Borrower or CJUF, CJUF may with prior written notice to Lender (but without Lender's consent) replace Stratus Block 21 Investments, L.P. as the managing member of Borrower (with Stratus Block 21 Investments, L.P. retaining its current ownership interest but as non-managing member), (b) even if a monetary Event of Default then exists, so long as CJUF continues to hold the ownership interest in Borrower which CJUF held at Closing, CJUF may request that Lender approve CJUF replacing Stratus Block 21 Investments, L.P. as managing member of Borrower (with Stratus Block 21 Investments, L.P. retaining its current ownership interest but as non-managing member), and Lender shall grant or deny such consent in its reasonable discretion, (c) any buyout of a member's interest or change in control permitted by the Operating Agreement of Borrower, as such Operating Agreement has been consented to by Lender, may be effected provided all membership interests in Borrower remain encumbered by the Pledge Agreement, it being agreed that when and as required by Lender, any new member in Borrower must execute and deliver to Lender a Joinder Agreement relating to the Pledge Agreement, and (d) the following Transfers of interests in Borrower's direct or indirect constituent entities shall be permitted without Lender's consent: (i) Transfers of direct or indirect interests in Canyon-Johnson Urban Fund II, L.P., (ii) Transfers of non-controlling interests in Guarantor, (iii) the transfer of an interest of no more than twenty percent (20%) in Borrower to Wheelock Street Capital (or an entity affiliated with Wheelock Street Capital) or another preferred equity provider who may provide up to \$35,000,000.00 of equity through Borrower for the Project and any transfers of direct and indirect interests in that new equity partner; (iv) Leases permitted hereunder; and (v) the sale and release o

17.3 Prohibition of Transfers in Violation of ERISA.

In addition to the prohibitions set forth in Section 17.2 above, Borrower shall not assign, sell, pledge, encumber, transfer, hypothecate or otherwise dispose of its interest or rights in this Agreement or in the Project, or attempt to do any of the foregoing or suffer any of the foregoing, nor shall any party owning a direct or indirect interest in Borrower assign, sell, pledge, mortgage, encumber, transfer, hypothecate or otherwise dispose of any of its rights or interest (direct or indirect) in Borrower, attempt to do any of the foregoing or suffer any of the foregoing, if such action would cause the Loan, or the exercise of any of Lender's rights in connection therewith, to constitute a prohibited transaction under ERISA or the Internal Revenue Code or otherwise result in Lender being deemed in violation of any applicable provision of ERISA. Borrower agrees to indemnify and hold Lender free and harmless from and against all losses, costs (including reasonable attorneys' fees and expenses), taxes, damages (including consequential damages) and expenses Lender may suffer by reason of the investigation, defense and settlement of claims and in obtaining any prohibited transaction exemption under ERISA necessary or desirable in Lender's sole judgment or by reason of a breach of the foregoing prohibitions. The foregoing indemnification shall be a recourse obligation of Borrower and shall survive repayment of the Note, notwithstanding any limitations on recourse contained herein or in any of the Loan Documents.

17.4 Successors and Assigns.

Subject to the foregoing restrictions on transfer and assignment contained in this Article 17, this Agreement shall inure to the benefit of and shall be binding on the parties hereto and their respective successors and permitted assigns.

ARTICLE 18

TIME OF THE ESSENCE

18.1 Time is of the Essence.

Time is of the essence under this Agreement.

ARTICLE 19

EVENTS OF DEFAULT

19.1 Events of Default.

The occurrence of any one or more of the following shall constitute an "Event of Default" as said term is used herein:

(a) Failure of Borrower (i) to make any principal, interest or other payment on or with regard to the Loan, when due, and such payment remains unpaid upon the expiration of five (5) days following the date written notice of such failure to pay is provided to Borrower; provided, however, the Lender shall not be obligated to provide any such written notice or cure period in regard to the payment due on the Loan on the maturity date of the Loan, whether such maturity date is the stated Maturity Date thereof or any accelerated maturity date declared by the Lender pursuant to the Loan Documents or (ii) to observe or perform any of the other covenants or conditions by Borrower to be performed under the terms of this Agreement or any other Loan Document for a period of thirty (30) days after written notice from Lender, provided that if any such failure concerning a non-monetary covenant or condition is susceptible to cure and cannot reasonably be cured within said thirty (30) day period, then Borrower shall have an additional ninety (90) day period to cure such failure and no Event of Default shall be deemed to exist under this paragraph (a) so long as Borrower commences such cure within the initial thirty (30) day period and diligently and in good faith pursues such cure to completion within the one hundred and twenty (120) day period from the date of Lender's notice; and provided further that if a different notice or grace period is specified under any other subsection of this Article 19 with respect to a particular breach, or if another subsection of this Article 19 applies to a particular breach and does not expressly provide for a notice or grace period, the specific provision shall control.

(b) An Event of Default under any of the Senior Loan Documents.

(c) [reserved]

- (d) The bankruptcy or insolvency of the General Contractor and failure of Borrower to procure a contract with a new contractor satisfactory to Lender within forty-five (45) days from the occurrence of such bankruptcy or insolvency.
 - (e) Any Transfer or other disposition in violation of Section 17.2 or 17.3.
- (f) Any warranty, representation, statement, report or certificate made now or hereafter by Borrower or any Guarantor is untrue or incorrect in any material respect at the time made or delivered, provided that if such breach is reasonably susceptible to cure, then no Event of Default shall exist so long as Borrower cures said breach (i) within the notice and cure period provided in (a)(ii) above.
- (g) Borrower or Guarantor shall commence a voluntary case concerning Borrower or Guarantor under the Bankruptcy Code; or an involuntary proceeding is commenced against Borrower or Guarantor under the Bankruptcy Code and relief is ordered against Borrower or Guarantor, or the petition is controverted but not dismissed or stayed within sixty (60) days after the commencement of the case, or a custodian (as defined in the Bankruptcy Code) is appointed for or takes charge of all or substantially all of the property of Borrower or Guarantor; or Borrower or Guarantor commences any other proceedings under any reorganization, arrangement, readjustment of debt, relief of debtors, dissolution, insolvency or liquidation or similar Law of any jurisdiction whether now or hereafter in effect relating to Borrower or Guarantor; or there is commenced against Borrower or Guarantor any such proceeding that remains undismissed or unstayed for a period of sixty (60) days; or Borrower or Guarantor fails to controver in a timely manner any such case under the Bankruptcy Code or any such proceeding, or any order of relief or other order approving any such case or proceeding is entered; or Borrower or Guarantor by any act or failure to act indicates its consent to, approval of, or acquiescence in any such case or proceeding or the appointment of any custodian or the like of or for it for any substantial part of its property or suffers any such appointment to continue undischarged or unstayed for a period of sixty (60) days.
- (h) Borrower or Guarantor shall make an assignment for the benefit of creditors, or shall admit in writing its inability to pay its debts generally as they become due, or shall consent to the appointment of a receiver or trustee or liquidator of all of its property or the major part thereof or if all or a substantial part of the assets of Borrower or Guarantor are attached, seized, subjected to a writ or distress warrant, or are levied upon, or come into the possession of any receiver, trustee, custodian or assignee for the benefit of creditors.
 - (i) Borrower or Guarantor shall dissolve or its existence shall otherwise terminate.
 - (j) Guarantor shall breach any of the Guarantor Financial Covenants.
- (k) Borrower is enjoined, restrained or in any way prevented by any court order from constructing or operating the Project or marketing or selling Residential Units for a period in excess of thirty (30) days.

- (l) One or more final, unappealable judgments are entered against (i) Borrower in amounts aggregating in excess of \$250,000, or (ii) against Guarantor in amounts aggregating in excess of \$250,000 for Guarantor, and said judgments are not paid, stayed or bonded over within thirty (30) days after entry.
- (m) Borrower or Guarantor shall fail to pay any debt owed by it or him, as applicable, when due (either by maturity (without forbearance) or acceleration) to Lender or any other party (other than a failure or default for which Borrower's maximum liability does not exceed \$200,000 and Guarantor's maximum liability does not exceed \$500,000).
 - (n) If a Material Adverse Change occurs with respect to Borrower, the Project or Guarantor.
- (o) Borrower shall fail to comply with its obligations under this Agreement and/or any other Loan Document relating to the obtaining and maintenance of insurance coverages.
 - (p) The Pre-Sale Requirement shall cease to be satisfied.
- (q) Borrower shall default under any of the Hotel Documents and shall fail to cure such default within the shorter of (i) the applicable cure period set forth in Section 19.1(a), and (ii) the applicable grace or cure period set forth in the applicable Hotel Document, or any Hotel Document shall otherwise cease to be in full force and effect (or any other event occurring thereunder which would give the Hotel Operator the right to terminate a Hotel Document).
- (r) Borrower shall default under any Material Contract, other than the Hotel Documents, and shall fail to cure such default within the shorter of the applicable cure period set forth in Section 19.1(a) or the applicable grace or cure period set forth in such Material Contract, or such Material Contract shall otherwise cease to be in full force and effect.
- (s) The occurrence of any other event or circumstance identified as an Event of Default in this Agreement or under any of the other Loan Documents and the expiration of any applicable grace or cure periods, if any, specified for such Event of Default herein or therein, as the case may be.

ARTICLE 20

LENDER'S REMEDIES IN EVENT OF DEFAULT

20.1 Remedies Conferred Upon Lender.

Upon the occurrence and during the continuance of any Event of Default, but subject in all events to the provisions of the Intercreditor Agreement, Lender may pursue any one or more of the following remedies concurrently or successively, in the sole discretion of Lender, it being the intent hereof that none of such remedies shall be to the exclusion of any other:

(a) Accelerate the Maturity Date and declare the Note to be immediately due and payable.

(b) Reserved.

(c) Exercise or pursue any other remedy or cause of action permitted under this Agreement and/or any of the other Loan Documents (including, without limitation, the exercise of the power of sale under the Deed of Trust), or conferred upon Lender by operation of Law.

Notwithstanding the foregoing, upon the occurrence of any Event of Default under Section 19.1(g) with respect to Borrower, all amounts evidenced by the Note shall automatically become due and payable, without any presentment, demand, protest or notice of any kind to Borrower.

ARTICLE 2

GENERAL PROVISIONS

21.1 Captions.

The captions and headings of various Articles, Sections and subsections of this Agreement and Exhibits pertaining hereto are for convenience only and are not to be considered as defining or limiting in any way the scope or intent of the provisions hereof.

21.2 Modification: Waiver.

No modification, waiver, amendment or discharge of this Agreement or any other Loan Document shall be valid unless the same is in writing and signed by the party against which the enforcement of such modification, waiver, amendment or discharge is sought.

21.3 Governing Law.

Irrespective of the place of execution and/or delivery, this Agreement shall be governed by, and shall be construed in accordance with, the laws of the State of Texas.

21.4 Acquiescence Not to Constitute Waiver of Lender's Requirements.

Each and every covenant and condition for the benefit of Lender contained in this Agreement may be waived by Lender, provided, however, that to the extent that Lender may have acquiesced in any noncompliance with any construction or nonconstruction conditions precedent to the Closing Funding, the Full Opening of the Loan or to any subsequent disbursement of Loan proceeds, such acquiescence shall not be deemed to constitute a waiver by Lender of such requirements with respect to any future disbursements of Loan proceeds.

21.5 Disclaimer by Lender.

This Agreement is made for the sole benefit of Borrower and Lender, and no other person or persons shall have any benefits, rights or remedies under or by reason of this Agreement, or by reason of any actions taken by Lender pursuant to this Agreement. Lender shall not be liable to any contractors, subcontractors, supplier, architect, engineer, tenant or other party for labor or

services performed or materials supplied in connection with the Construction. Lender shall not be liable for any debts or claims accruing in favor of any such parties against Borrower or others or against the Project. Lender, by making the Loan or taking any action pursuant to any of the Loan Documents, shall not be deemed a partner or a joint venturer with Borrower or fiduciary of Borrower. No payment of funds directly to a contractor or subcontractor or provider of services shall be deemed to create any third-party beneficiary status or recognition of same by Lender. Without limiting the generality of the foregoing:

- (a) Lender shall have no liability, obligation or responsibility whatsoever with respect to the Construction. Any inspections of the Construction made by or through Lender are for purposes of administration of the Loan only and neither Borrower nor any third party is entitled to rely upon the same with respect to the quality, adequacy or suitability of materials or workmanship, conformity to the Plans and Specifications, state of completion or otherwise;
- (b) Lender neither undertakes nor assumes any responsibility or duty to Borrower to select, review, inspect, supervise, pass judgment upon or inform Borrower of any matter in connection with the Project, including matters relating to the quality, adequacy or suitability of: (i) the Plans and Specifications, (ii) architects, contractors, subcontractors and material suppliers employed or utilized in connection with the Construction, or the workmanship of or the materials used by any of them, or (iii) the progress or course of Construction and its conformity or nonconformity with the Plans and Specifications; Borrower shall rely entirely upon its own judgment with respect to such matters, and any review, inspection, supervision, exercise of judgment or supply of information to Borrower by Lender in connection with such matters is for the protection of Lender only, and neither Borrower nor any third party is entitled to rely thereon; and
- (c) Lender owes no duty of care to protect Borrower or Guarantor, any Tenant or Residential Unit Purchaser or any other person or entity against negligent, faulty, inadequate or defective building or construction.

21.6 Partial Invalidity; Severability.

If any of the provisions of this Agreement, or the application thereof to any person, party or circumstances, shall, to any extent, be invalid or unenforceable, the remainder of this Agreement, or the application of such provision or provisions to persons, parties or circumstances other than those as to whom or which it is held invalid or unenforceable, shall not be affected thereby, and every provision of this Agreement shall be valid and enforceable to the fullest extent permitted by law.

21.7 Definitions Include Amendments.

Definitions contained in this Agreement that identify documents, including, but not limited to, the Loan Documents, shall be deemed to include all amendments and supplements to such documents from the date hereof, and all future amendments, modifications, and supplements thereto entered into from time to time to satisfy the requirements of this Agreement or otherwise with the consent of Lender. Reference to this Agreement contained in any of the

foregoing documents shall be deemed to include all amendments and supplements to this Agreement.

21.8 Execution in Counterparts.

This Agreement may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement.

21.9 Entire Agreement.

This Agreement, taken together with all of the other Loan Documents and all certificates and other documents delivered by Borrower to Lender, embodies the entire agreement and supersede all prior agreements, written or oral, relating to the subject matter hereof.

21.10 Reserved.

21.11 References to Senior Loan Agreement and Concurrent Lenders Approval.

Notwithstanding any provision herein to the contrary, wherever in this Agreement there are references to "in accordance with the terms of the Senior Loan Agreement", or similar words, if the Senior Loan has been paid in full, Borrower shall still comply with such provisions even though the Senior Loan Agreement is no longer in effect (except to the extent such provisions clearly no longer apply), and any approvals or consents required by Senior Lender pursuant to such provisions shall be made by Lender. Notwithstanding any provision herein to the contrary, (i) whenever both the Senior Lender's approval and/or the Lender's approval are required under the Loan Documents and the Senior Loan Documents for the same matter, an approval by the Senior Lender shall be deemed an approval by the Lender under the Loan Documents; (ii) whenever Lender approval is required under the Loan Documents and the same matter requires Senior Lender approval under the Senior Loan Documents, an approval by the Senior Lender shall be deemed an approval by the Lender under the Loan Documents; and (iii) the Loan Documents and all rights and remedies of Lender under the Loan Documents are subject and subordinate to the Senior Loan Documents as provided in the Intercreditor Agreement.

21.12 Jurisdiction.

TO THE GREATEST EXTENT PERMITTED BY LAW, BORROWER HEREBY WAIVES ANY AND ALL RIGHTS TO REQUIRE MARSHALLING OF ASSETS BY LENDER. WITH RESPECT TO ANY SUIT, ACTION OR PROCEEDINGS RELATING TO THIS AGREEMENT (EACH, A "PROCEEDING"), BORROWER IRREVOCABLY (a) SUBMITS TO THE NON-EXCLUSIVE JURISDICTION OF THE STATE AND FEDERAL COURTS HAVING JURISDICTION IN THE COUNTY OF DALLAS AND STATE OF TEXAS, AND (b) WAIVES ANY OBJECTION THAT IT MAY HAVE AT ANY TIME TO THE LAYING OF VENUE OF ANY PROCEEDING BROUGHT IN ANY SUCH COURT, WAIVES ANY CLAIM THAT ANY PROCEEDING HAS BEEN BROUGHT IN AN INCONVENIENT FORUM AND FURTHER WAIVES THE RIGHT TO OBJECT, WITH

RESPECT TO SUCH PROCEEDING, THAT SUCH COURT DOES NOT HAVE JURISDICTION OVER SUCH PARTY. NOTHING IN THIS AGREEMENT SHALL PRECLUDE LENDER FROM BRINGING A PROCEEDING IN ANY OTHER JURISDICTION NOR WILL THE BRINGING OF A PROCEEDING IN ANY ONE OR MORE JURISDICTIONS PRECLUDE THE BRINGING OF A PROCEEDING IN ANY OTHER JURISDICTION. BORROWER FURTHER AGREES AND CONSENTS THAT, IN ADDITION TO ANY METHODS OF SERVICE OF PROCESS PROVIDED FOR UNDER APPLICABLE LAW, ALL SERVICE OF PROCESS IN ANY PROCEEDING IN ANY TEXAS STATE OR UNITED STATES COURT HAVING JURISDICTION OVER THE COUNTY OF DALLAS, TEXAS MAY BE MADE, TO THE EXTENT PERMITTED BY LAW, BY CERTIFIED OR REGISTERED MAIL, RETURN RECEIPT REQUESTED, DIRECTED TO BORROWER AT THE ADDRESS INDICATED BELOW, AND SERVICE SO MADE SHALL BE COMPLETE UPON RECEIPT; EXCEPT THAT IF BORROWER SHALL REFUSE TO ACCEPT DELIVERY (AS OPPOSED TO UNABLE TO RECEIVE DELIVERY), SERVICE SHALL BE DEEMED COMPLETE FIVE (5) DAYS AFTER THE SAME SHALL HAVE BEEN SO MAILED.

21.13 Reserved.

21.14 Binding Effect.

The covenants, conditions, waivers, releases and agreements contained in this Agreement shall bind, and the benefits thereof shall inure to the parties hereto and their respective heirs, executors, administrators, successors and assigns.

21.15 Waiver of Accord and Satisfaction.

Borrower hereby expressly waives any and all rights to effect an accord and satisfaction of any secured obligation or any other debt of Borrower to Lender in accordance with section 3-311 of the UCC.

ARTICLE 22

NOTICES

Any notice, demand, request or other communication that any party hereto may be required or may desire to give hereunder shall be in writing and shall be deemed to have been properly given (a) if hand delivered, when delivered; (b) if by Federal Express or other reliable overnight courier service, on the next Business Day after delivered to such courier service or (c) if by telecopier on the day of transmission so long as copy is sent on the same day by overnight courier as set forth below:

If to Borrower:

c/o Stratus Properties

98 San Jacinto, Suite 220

Austin, Texas 78701

Attention: W.H. Armstrong III

Telephone: (512) 478-6396

Facsimile: (512) 478-5788

And to:

Armbrust & Brown, L.L.P.

100 Congress Avenue, Suite 1300

Austin, Texas 78701

Attention: Kenneth N. Jones, Esq.

Telephone: (512) 435-2312 Facsimile: (512) 435-2360

And to:

Canyon-Johnson Urban Funds

2000 Avenue of the Stars, $11^{th}\ Floor$

Los Angeles, California 90067

Attention: Head of Asset Management

Telephone: (310) 272-1500 Facsimile: (310) 272-1523

And to:

Canyon-Johnson Urban Funds

2000 Avenue of the Stars, 11th Floor

Los Angeles, California 90067

Attention: General Counsel
Telephone: (310) 272-1500
Facsimile: (310) 272-1523

And to:

DLA Piper US LLP

550 South Hope Street, Suite 2300

Los Angeles, California 90071

Attention: Steven A. Fein, Esq.

Telephone: (213) 330-7772 Facsimile: (213) 330-7572

If to Lender:

Hunter's Glen/Ford Investments I LLC 200 Crescent Court, Suite 1350 Dallas, Texas 75201

President 214.871.5131 214.871.5199 Attention: Telephone: Facsimile:

With a copy to:

Haynes and Boone LLP

2323 Victory Avenue, Suite 700

Dallas, Texas 75219

Attention: William C. Wilshusen, Esq.

Telephone 214.651.5595 Facsimile 214.200.0635

or at such other address as the party to be served with notice may have furnished in writing to the party seeking or desiring to serve notice as a place for the service of notice.

ARTICLE 23

WAIVER OF JURY TRIAL

BORROWER AND LENDER EACH WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHTS UNDER THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP THAT IS THE SUBJECT OF THIS AGREEMENT AND AGREE THAT ANY SUCH ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

BORROWER:

CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company

By: Stratus Block 21 Investments, L.P., a Texas limited partnership, Manager

By: Stratus Block 21 Investments GP, L.L.C., a Texas limited liability company, General Partner

By: /s/ Erin D. Pickens
Name: Erin D. Pickens
Title: Senior Vice President

By: CJUF II Block 21 Member, LLC, a Delaware limited liability company, Member

By: Canyon-Johnson Urban Fund II, L.P., a Delaware limited partnership, Member

By: Canyon-Johnson Realty Advisors II LLC, a Delaware limited liability company, General Partner

By: _/s/ K. Robert Tumer
Name: K. Robert Tumer
Title: Managing Partner

LENDER:

HUNTER'S GLEN/FORD INVESTMENTS I LLC

EXHIBIT A

Legal Description of Land

Lots 1 through 12, Block 21, of the Original City of Austin, Travis County, Texas, according to the map or plat of record in the General Land Office of the State of Texas, together with the area within the alley traversing said Block, which was vacated by Ordinance recorded under Document No. 1999086902 and described in Memorandum Designating the Vacation of a 20 foot wide alley on Block 21 and Block 22, in the City of Austin as recorded under Document No. 2004040650 of the Official Public Records of Travis County, Texas.

EXHIBIT B

Construction Schedule

[See Attached]

EXHIBIT C

Permitted Exceptions

- Restrictive Covenants recorded under Document No(s). 2006234734, 2006240877, 2007225317, 2008013712, 2008078527, 2008085863 and 2009176432 of the Official Public Records of Travis County, Texas.
- All interest in all oil, gas and other minerals reserved by Missouri Pacific Railroad Company in instrument recorded in Volume 8687, Page 248 of
 the Real Property Records of Travis County, Texas, as modified and amended by the Modification of Mineral Reservation dated January 7, 2000,
 recorded under Document No. 2000010490 in the Official Public Records of Travis County, Texas. (Surface Rights waived therein) (Lots 7-12).
- 3. The terms, conditions and stipulations, including but not limited to repurchase rights, set forth in that certain Special Warranty Deed dated to be effective December 15, 2006 and recorded under Document No. 2006240878 of the Official Public Records of Travis County, Texas, as further affected by Estoppel Certificate and Agreements recorded under Document No(s). 2008078527, 2008085863, and 2009176432 of the Official Public Records of Travis County, Texas and Estoppel Certificate and Agreement recorded concurrently with this Deed of Trust.
- 4. The terms, conditions and stipulations set out in that certain Restrictive Covenant Regarding Unified Development and Maintenance of Drainage Facilities dated January 9, 2008, recorded under Document No. 2008013712 of the Official Public Records of Travis County, Texas.
- The terms, conditions and stipulations set out in that certain Temporary Right of Way Encroachment License Agreement dated June 4, 2008, recorded under Document No. 2008094734 of the Official Public Records of Travis County, Texas.
- 6. The terms, conditions and stipulations set out in that certain Temporary Right of Way Encroachment License Agreement dated August 29, 2008, recorded under Document No. 2008151032 of the Official Public Records of Travis County, Texas.
- 7. Terms, conditions and stipulations set out in that certain Operating Agreement dated October 26, 2006, as amended by First Amendment to Operating Agreement dated January 30, 2008 (and by Second Amendment to Operating Agreement dated as of May 6, 2008), as evidenced by Subordination and Non-Disturbance Agreement, recorded under Document No. 2008078528 of the Official Public Records of Travis County, Texas, as further evidenced by Subordination and Non-Disturbance Agreement recorded under Document No. 2009176433 of the Official Public Records of Travis County, Texas, and as further evidenced by Subordination and Non-Disturbance Agreement in recorded concurrently with this Deed of Trust.

- Apparent easement evidenced by the location of electric meter outside of a dedicated easement as shown on the Survey dated March 24, 2010, prepared by William H. Ramsey, Registered Professional Land Surveyor No. 4532.
- 9. The terms, conditions and stipulations set out in that certain Electric Utility Easement and Vault Agreement dated August 4, 2009, recorded under Document No. 2009133082 of the Official Public Records of Travis County, Texas.
- 10. The terms, conditions and stipulations set out in that certain Estoppel Certificate and Agreement recorded under Document No. 2009176432 of the Official Public Records of Travis County, Texas.

EXHIBIT D

Hunter's Glen/Ford Investments I LLC Attn: President 200 Crescent Court, Suite 1350 Dallas, Texas 75201

Re:		an Agreement by and between CJUF II Stratus Block 21 LLC and Hunter's Glen/Ford Investments I LLC, dated as ent"); Certificate Concerning Litigation
	Dear:	
		rsuant to Section 8.1(f) of the Loan Agreement that as of the date of this certificate to the best of my knowledge ing or threatened that is reasonably and likely to cause a Material Adverse Change (as defined in the Loa
		BORROWER:
		CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company
		By: Stratus Block 21 Investments, L.P., a Texas limited partnership, Manager
		By: Stratus Block 21 Investments GP, L.L.C., a Texas limited liability company, General Partner
		By: William H. Armstrong, III, President
		Exhibit D - Page 1

EXHIBIT E

INSURANCE REQUIREMENTS DURING AND POST CONSTRUCTION

1. <u>Insurance</u>. Borrower shall obtain, and maintain at all times during term of the Loan, such insurance as Hunter's Glen/Ford Investments I LLC ("Lender") may reasonably require, including, but not limited to the insurance coverage set forth below. Unless otherwise expressly defined herein, capitalized terms set forth in this Exhibit are terms of art, as used in and understood in the insurance industry or are defined terms in the Loan Agreement to which this is attached.

A. <u>During Construction</u>.

- (a) <u>Builder's Risk</u>. From the closing of the loan until replaced by permanent property insurance, "All Risk" form of Builder's Risk Insurance, in such amount as Lender shall reasonably require, but in no event less than 100% of the replacement cost value of the Project (including Upgrades and any leasehold improvements) (the "Builder's Risk Insurance"). Such policy shall be written on a Builder's Risk Completed Value Form (100% non-reporting) or its equivalent and shall not contain a permission to occupy limitation or a coinsurance clause. Such policy shall not have exclusion for sidewalks, retaining walls or underground property. The policy shall include coverage for Flood and Earthquake with sub-limits no less than \$250,865,764 per occurrence and in the annual aggregate. Such insurance policy shall also include coverage for:
 - (i) Loss suffered with respect to Borrower's materials, equipment, machinery, and supplies whether on-site, in transit, or stored off site, with a limit in amounts sufficient to cover the replacement cost of such exposure subject to a minimum limit of \$10,000,000 for both transit and off site storage (per location) provided that Borrower shall obtain or cause to be obtained additional insurance whenever the value of materials in transit or storage exceed those limits;
 - (ii) \$21,820,000 in coverage will be provided for additional Soft Cost expenses including coverage for all types (including but not limited to interest expense; fees; and plans, specifications, blueprints and models, in connection with any restoration following an insured loss);
 - (iii) If applicable, loss or delay of rental income will be provided up to a minimum of \$12,302,173 on an actual loss sustained
- (b) <u>Comprehensive Broad Form Boiler and Machinery Insurance</u>, covering all mechanical and electrical apparatus and pressure vessels. Such insurance shall provide coverage against loss or damage from an accident to and/or caused by boilers and machinery, including but not limited to: heating

apparatus, pressure vessels, pressure pipes, electrical or air conditioning equipment on a blanket comprehensive coverage form, in such amount as Lender shall reasonably approve but no less than \$10,000,000. All exclusions for testing shall be removed.

- (c) <u>Professional Liability</u>. Borrower will require the architect, engineers (including Structural and MEP contractors) and all other design professionals retained by Borrower to purchase and maintain continuous professional liability coverage in the amount of \$1,000,000 per claim. This policy may be on a "claims made" basis, and shall include coverage for bodily injury and property damage and retroactive coverage back to the first date that professional services were provided to the Project.
- (d) <u>Commercial General Liability and Umbrella Liability</u> coverage, for contractors excluded from the Owner Controlled Insurance Program, including but not limited to, coverage for Personal Injury, Bodily Injury, Death, Property Damage, with limits of not less than \$2,000,000 per occurrence and in the annual aggregate. he policies described in this paragraph shall cover, without limitation: elevators, escalators, independent contractors, contractual liability (covering, to the maximum extent permitted by the commercial general liability policy, Borrower's obligation to indemnify Lender as required under this Exhibit) and Products and Completed Operations Liability coverage. Coverage should also include host liquor liability. Borrower shall add Lender, its directors, officers, employees and agents as additional insured.
- (e) <u>Worker's Compensation</u>. Worker's compensation insurance, for contractors excluded from the Owner Controlled Insurance Program, covering Borrower and its employees at the site to the extent required, and in the amounts required by applicable Laws. An endorsement providing U.S. Longshore and Harbor Workers Compensation Act (USL&HW) coverage should be added on an "if any" basis if there is an exposure.
- (f) <u>Employers Liability.</u> If applicable in the amount of \$1,000,000 per accident, \$1,000,000 per illness, per employee and \$1,000,000 per illness, in the aggregate, for contractors excluded from the Owner Controlled Insurance Program.
- (g) <u>Owner Controlled Insurance Program.</u> The following coverage shall be provided by the Borrower for all enrolled parties. Coverage shall include Borrower as an additional insured on the general liability and excess liability policies. If applicable, the Workers Compensation/Employers Liability policy shall also cover the Borrower.
 - (i) <u>Workers' Compensation Insurance</u> Statutory Limits of the Workers' Compensation Laws of the State of Texas, with Coverage B Employer's Liability (with limits of \$1,000,000 each accident for Bodily Injury by accident, \$1,000,000 each employee for Bodily Injury by disease

and \$1,000,000 policy limit Bodily Injury by disease), covering operations of the enrolled party performing Work at the project site.

Coverage under the CIP will remain in force through completion or other similar period described in the CIP policy. This insurance is extended for an additional period of time with respects to liability for "bodily injury" or "property damage" arising out of "repair work". This extension shall commence as of the date that work is completed and shall end as of the expiration of any express warranty for the work, or up to 24 months from the date of completion of work, whichever comes first.

(ii) Commercial General Liability Insurance - (Excluding Automobile and Professional Liability) in form providing coverage not less than a Commercial General Liability insurance policy, including hazards of explosion, collapse, underground, independent Contractor(s), employees as additional insureds, completed operations for 10 years after the Project has reached substantial completion as defined in Article 9.8 of the General Conditions of the Contract for construction, contractual liability coverage for claims arising out of the Work for personal injury, bodily injury and property damage in policy or policies of insurance such that the total available limits, reinstated annually to all insureds combined will not be less than:

\$2,000,000 per occurrence

\$2,000,000 personal and advertising injury aggregate

\$2,000,000 general aggregate

\$2,000,000 completed operations aggregate - term limit.

Coverage will apply only to work performed at the Project site. Such insurance will not include coverage for products liability for any product(s) manufactured, assembled, or otherwise worked upon away from the Project site for any enrolled party or excluded party performing such off-site work.

(iii) <u>Umbrella/Excess Liability Insurance</u> in limits of:

\$150,000,000 per occurrence

\$150,000,000 general aggregate

\$150,000,000 products/completed operations aggregate

excess of Employers Liability and Commercial General Liability stated above, to all insureds combined.

(h) <u>Automobile Liability.</u> Contractors shall be required to carry automobile liability insurance for all owned, hired and no owned vehicles with limits of at least \$1 million.

- (i) <u>Contractors Pollution Liability.</u> The Project will have a limit of \$15,000,000 per claim and a \$15,000,000 aggregate limit. The policy period shall coincide with the construction term. Coverage will apply to all claims as a result of Bodily Injury, Property Damage or Clean-up Costs caused by Pollution Conditions resulting from Covered Operations that are performed by or on the behalf of the Contractor at the project site. Coverage shall also include transportation, disposal site coverage and a minimum of ten (10) years completed operations.
- **B.** Post Construction. After the earlier of: (i) substantial completion of the Project, or (ii) cancellation or expiration of the Builder's Risk Policy, Borrower shall provide the following coverages:
 - (a) "All Risk" insurance including Flood and Earthquake, and such other insurable hazards as, under good insurance practices are insured against for other property and buildings similar to the premises in nature, use, location, height, and type of construction. The amount of such insurance shall be not less than one hundred percent (100%) of the replacement cost without depreciation of the Project. Such insurance policy shall contain an agreed amount endorsement. Flood and Earthquake sublimits shall be what is commercially available at reasonable costs in the market but in no event be less than \$100,000,000 each per occurrence and in the annual aggregate. Such insurance shall cover increased cost of law or ordinance insurance, costs of demolition and increased cost of construction with a sublimit of not less than \$10,000,000, debris removal with a sublimit of not less than \$15,000,000. Deductibles shall not exceed \$250,000 for physical damage. Cover shall also include extra expense cover with sublimits not less than \$5,000,000. Lender shall be named as sole Loss Payee and Mortgagee for the Project, for all claim proceeds, including business interruption insurance loss proceeds, in excess of \$250,000.
 - (b) Comprehensive Broad Form Boiler and Machinery. Insurance, in the minimum amount of \$10,000,000 covering all mechanical and electrical equipment against physical damage and covering, without limitation, all tenant improvements and betterments that Borrower is required to insure pursuant to any lease on a replacement cost basis. Such insurance shall provide coverage against loss or damage from an accident to and/or caused by boilers and machinery, including but not limited to: heating apparatus, pressure vessels, pressure pipes, and electrical or air conditioning equipment on a blanket comprehensive coverage form, in such amount Lender shall reasonably approve. All exclusions for testing shall be removed. Coverage shall be extended to include loss of rental income for 6 months as a result of damage from an insured peril.
 - (c) <u>Business Interruption</u>. Loss of rental income on an actual loss sustained, in an amount equal to 12 months of projected rental income, with a 120 day extended period of indemnification. Cover shall also include extra expense with a sublimit of not less than \$5,000,000. Waiting period of not greater than 30 days can apply.

- (d) <u>Commercial General Liability.</u> Commercial General Liability Insurance, for Personal Injury, Bodily Injury, Death, Accident and Property Damage providing in combination no less than \$100,000,000 per occurrence (with sublimits approved by Lender) and in the annual aggregate, <u>per location.</u> The policies described in this paragraph shall cover, without limitation: elevators, escalators, independent contractors, contractual liability and Products and Completed Operations Liability coverage. It is understood that umbrella or excess policies can be used to meet the required limits.
- (e) <u>Dram Shop</u>. Prior to any tenant selling alcoholic beverages on any part of the Project, Borrower either itself or through the Tenant shall provide evidence of so-called "Dram Shop" against claims or liabilities arising directly or indirectly to Persons or property on account of the sale or dispensing of alcoholic beverages. Coverage shall include loss of means of support. Limits shall equal those limits as may be required by applicable Laws or \$5,000,000, whichever is greater. If state law allows, Lender shall be named as an additional insured on such policy.
- (f) <u>Worker's Compensation</u>. If applicable, worker's compensation insurance covering Borrower and its employees at the site to the extent required, and in the amounts required by applicable Laws.
- (g) Employers Liability. If applicable in the amount of \$10,000,000 per accident; \$10,000,000 per illness, per employee; and \$10,000,000 per illness, in the aggregate. Such limits can be met through excess liability policies of insurance.
- (h) <u>Auto Liability for Owned (if any) and Hired and Non Owned with limits not less than \$10,000,000, unless otherwise approved by Lender.</u> If owner/Operator does not have any Automobiles , the Commercial General Liability policy must be endorsed to cover Non-Owned Automobile Liability.
- (i) <u>Directors & Officers Liability</u>. To the extent the project has the exposure at the time it becomes operational, Directors and Officers liability insurance providing cover for Side A and Side B in an amount not less than \$5,000,000 subject to industry standard deductibles. If deemed to be required, this requirement can be satisfied by the separate coverage of the individual Borrower membership entities
 - Other. Such other insurances as may be reasonably requested by Lender.

II. Requirements of Insurance Policies.

(a) All insurance policies shall be issued by an insurer or insurers with an A.M. Best rating of A:IX or better or a Standard and Poor's rating of "AA", or equivalent rating from another agency acceptable to Lender and be authorized in the state where the Project is located. All insurance acquired pursuant to this

Exhibit E - Page 5

Exhibit shall be in form, amounts and with coverage and deductibles satisfactory to Lender, in Lender's sole discretion.

- (b) The Builder's Risk insurance policies required to be carried pursuant to Section I.A., Subsections (a) and (b) of this Exhibit, and the All Risk required pursuant to Section I.B, Subsections (a) and (b), shall name Borrower as the insured and shall also name Lender as Loss Payee and Mortgagee, under a non-contributing standard mortgagee clause.
- (c) The Commercial General Liability, Automobile Liability, and Pollution Liability policies required in this <u>Exhibit E</u> shall name Lender, its directors, officers, and employees as Additional Insured.
- (d) Lender shall be named as a dual-obligee on the Performance Bond required to be obtained by General Contractor pursuant to Section I.A., Subsection (i) of this Exhibit.
 - (e) The amount of any deductible under any insurance policy must be reasonably acceptable to Lender.
- (f) Borrower may provide required insurance under blanket policies. Borrower shall not maintain any insurance on the Project that does not name Lender as Loss Payee.
- (g) Borrower shall pay the premiums for the insurance policies as the same become due and payable. Borrower shall deliver to Lender certified copies of the insurance policies required to be maintained pursuant to this Exhibit within sixty (60) days after the date of this Agreement or ten (10) days after the issuance of the policies by the insurer, whichever is later, but in all events, no later than ninety (90) days after the date of this Agreement, and failure to do so will be an immediate Event of Default. Notwithstanding the foregoing, Lender shall not be deemed by reason of the custody of such insurance policies to have knowledge of the contents thereof. Borrower also shall deliver to Lender, within ten (10) days of Lender's request, a certificate of Borrower or Borrower's insurance agent setting forth the particulars as to all such insurance policies, that all premiums due thereon have been paid currently and that the same are in full force and effect. BORROWER SHALL DELIVER A CERTIFICATE OR OTHER EVIDENCE OF INSURANCE ACCEPTABLE TO LENDER EVIDENCING THE INSURANCE REQUIRED HEREUNDER ON THE CLOSING DATE, TOGETHER WITH RECEIPTS FOR THE PAYMENT OF PREMIUMS THEREON. ALL CERTIFICATES FOR PROPERTY INSURANCE MUST BE ON ACORD FORM 27 or the equivalent; ACORD 25 certificates are acceptable for liability insurance. Not later than fifteen (15) days prior to the expiration date of each of the insurance policies or as soon as coverage has been bound, but no later than prior to the renewal date. Borrower shall deliver to Lender a certificate of insurance evidencing renewal of coverage as required herein. Within ten (10) days after such renewal, Borrower shall deliver to Lender evidence of payment of

Exhibit E - Page 6

premium satisfactory to Lender. Not later than ninety (90) days after the renewal of each of the insurance policies, Borrower shall deliver to Lender an original or certified copy (as required pursuant to this Section) of a renewal policy or policies.

- (h) Each insurance policy shall contain a provision whereby the insurer agrees that so long as the Loan is outstanding, such policy shall not be canceled without at least thirty (30) days prior written notice to Lender, except ten (10) days for non-payment of premium
- (i) In the event any insurance policy (except for general and other liability and Workers Compensation insurance) shall contain breach of warranty provisions, such policy shall provide that with respect to the interest of Lender, such insurance policy shall not be invalidated by and shall insure Lender regardless of; (a) any act, failure to act or negligence of or violation of warranties, declarations or conditions contained in such policy by any named insured; (b) the occupancy or use of the property for purposes more hazardous than permitted by the terms thereof; or (c) any foreclosure or other action or proceeding taken by Lender pursuant to any provision of this Agreement.
- (j) Any insurance maintained pursuant to this Agreement may be evidenced by blanket insurance policies covering the premises and other properties or assets of Borrower or its affiliates; provided that any such policy shall in all other respects comply with the requirements of this section. Lender, in its reasonable discretion, shall determine whether such blanket policies contain sufficient limits of insurance.
- (k) Any insurance carried by Lender shall be for its sole benefit and shall not inure to the benefit of Borrower and Insurance required from Borrower shall be primary to any available, if any, to Lender.
- (I) All required policies, other than professional liability, shall provide that insurers have waived rights of subrogation against Lender. The required insurance shall be primary without right of contribution from any insurance, which may be carried by Lender.
- (m) The required limits are minimum limits established by Lender and nothing contained herein shall be construed to mean the required limits are adequate or appropriate to protect Borrower from greater loss.

Exhibit E - Page 7

EXHIBIT F

Architect's Certificate

The firm of BOKA Powell, LLC ("Architect"), hereby certifies for the benefit of HUNTER'S GLEN/FORD INVESTMENTS I LLC and CJUF II Stratus Block 21 LLC, to the best of its current knowledge, information and belief, that:

Architect has been employed by Stratus Block 21 Investments GP, L.L.C., a Texas limited liability company ("Stratus Block 21"), pursuant to a contract dated May 3, 2006 (as amended and assigned, the "Architect Agreement"), to provide architectural and engineering services for what is commonly known as W Hotel & Residences, which is located at Block 21 in Austin, Texas (the "Project"). The Architect Agreement was assigned by Stratus Block 21 to CJUF II Stratus Block 21 LLC, a Delaware limited liability company, the current owner of the Project, by assignment dated July 30, 2007 (referred to herein as the "Owner"). The Project is located on the property described on Exhibit "A" attached hereto (the "Land") and the improvements being constructed on the Land in conjunction with the Project is referred to herein as the "Improvements."

A true and correct copy of the Architect Agreement, together with all amendments thereto, which sets out Architect's services is attached hereto as Exhibit "B."

The firm is duly licensed and in good standing under laws of the State of Texas, License No. 10147.

The foundations were designed by a structural engineer retained by Architect in accordance with the recommendations contained in a soil report dated August 10, 2007, which was prepared by Henley-Johnston & Associates, Inc.

Plans listed on the attached Schedule I comprise the plans which are necessary for the construction of the Project, excepting tenant space designs, and when the Project is built in accordance therewith the Project will (excepting completion of tenant improvements) be ready for occupancy. Calculations of the gross building and the net rentable building area are attached as Schedule II. The plans as prepared by Architect and Architect's consultants (and the Project will, when constructed in accordance therewith) comply with all applicable building, zoning, land use, subdivision, environmental, fire, safety and other applicable governmental laws, statutes, codes, ordinances, rules and regulations. In addition, (i) the Urban Design Section of the Neighborhood Planning and Zoning Department at the City of Austin approved the design of the streetscape improvements for the Project in accordance with Restrictive Covenant recorded under Document No. 2006234734 of the Official Public Records of Travis County, Texas, (ii) the Project is designed as a unified Development under its Site Development Permit No. SP-20070371C issued by the City of Austin (the "Site Development Permit") in accordance with Restrictive Covenant Regarding Unified Development and Maintenance of Drainage Facilities recorded under Document No. 2008013712 of the Official Public Records of Travis County, Texas (the "Unified Development RC"), (iii) joint access for the Project is designed in accordance with Section 4 of the Unified Development RC, and (iv) parking for the Project is designed in accordance with Section 5 of the Unified Development RC.

Exhibit F - Page 1

The attached <u>Schedule III</u>, establishing a timetable for completion of the Project and showing on a monthly basis the anticipated progress of the work, is realistic and can be adhered to.

The following design drawings or plans have been prepared with assistance from other designers or contractors.

<u>Type of Plans</u> <u>Name of Preparing Firm</u>

Construction Document Package No. 1	BOKA Powell & all Consultants
Construction Document Package No. 2	BOKA Powell & all Consultants
Construction Document Package No. 3	BOKA Powell & all Consultants
Construction Document Package No. 4	BOKA Powell & all Consultants
Construction Document Package No. 5	BOKA Powell & all Consultants

The Construction Document Pages have been supplemented by the Addenda issued as described on <u>Schedule I</u> attached hereto. The Specifications are included as part of the aforementioned Construction Documents.

Architect has received payment of all sums due under the Architect Agreement (the "Progress Payments") other than the current amount due in the amount of \$\(\) (the "Current Amount") for services provided through ______ In consideration of the payment by Owner of the Progress Payments and other good and valuable consideration, the receipt of all of which is hereby acknowledged, Architect has waived and released and, acting herein by and through me, does hereby waive and release, any and all liens, rights and interests (whether choate or inchoate and including, without limitation, all mechanic's and materialman's liens under the Constitution, statutes and laws of the State of Texas) owned, claimed or held, or to be owned, claimed or held by Architect in and to the Land and in and to the Improvements, whether now or hereafter constructed thereon, by reason of services performed by Architect to _____ (but not after such date); and the Architect for itself, its representatives and assigns does release, acquit and forever discharge Owner, its successors and assigns from any and all claims, debts, demands and causes of action that Architect has or may have as a result of the same; provided, however, that the release contained in this paragraph shall in no way diminish, modify or otherwise affect the validity of any liens, rights and interests to be owned, claimed or held by Architect by reason of services performed subsequent to _______ hereby by Architect or for the Current Amount due.

Architect has actual knowledge, and hereby certifies, that all bills owed by Architect to others for materials furnished and services performed in connection with any services or work product provided under the Architect Agreement with regard to the Land or the Improvements have been fully paid and satisfied except for amounts reflected in the Current Amount and for services and/or work product performed subsequent to ________.

Exhibit F - Page 2

BOKA POWELL, LLC

	Ву:		
	Name:		
	Title:		
	Date:		
HE STATE OF TEXAS	§ §		
OUNTY OF	§		
	acknowledged before me on tl of BOKA Powell, LLC, a		,
	Notary Public S	Signature	
	(SEAL)		
	Ext	hibit F - Page 3	

EXHIBIT G

Budget

[See Attached]

Exhibit G - Page 1

EXHIBIT H

[reserved]

Exhibit H - Page 1

EXHIBIT I

Approved Plans and Specifications and Approved Finish Standards

[See Attached]

Exhibit I - Page 1

	Subcontracts
	[See Attached]
	_[рее лииспеи]
E	xhibit J - Page 1

EXHIBIT J

EXHIBIT K

[reserved]

Exhibit K - Page 1

EXHIBIT L

[reserved]

Exhibit L - Page 1

EXHIBIT M

List of Sales Agreements

[See Attached]

Exhibit M - Page 1

EXHIBIT M-1

List of Sales Agreements Discrepancies

[See Attached]

Exhibit M-1 - Page 1

EXHIBIT M-2

Approved Price List

[See Attached]

Exhibit M-2 - Page 1

EXHIBIT N

[reserved]

Exhibit N - Page 1

EXHIBIT O

Approved Form of Sales Agreement

[See Attached]

Exhibit O - Page 1

EXHIBIT P

[reserved]

Exhibit P - Page 1

EXHIBIT Q

LEASING PARAMETERS AND ALLOWABLE TENANT IMPROVEMENTS

W Austin Commercial Space Leases

						Retail						
Term (Years)	5	7	10	AVERAG	E 5	7	10	AVERAG	E 5	7	10 /	AVERAGE
Minimum 1st Year	\$27.00	\$27.00	\$27.00	\$27.00	\$29.00	\$29.00	\$29.00	\$29.00	\$26.00	\$26.00	\$26.00	\$26.00
Base Rent	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50	\$ 0.50
Annual Bumps	\$28.00	\$28.50	\$29.25	\$28.58	\$30.00	\$30.50	\$31.25	\$30.58	\$27.00	\$27.50	\$28.25	\$27.58
Minimum Average	6%	6%	6%	6%	6%	6%	6%	6%				
Base Rent												
Additional Rent	\$28.00	\$30.00	\$35.00	\$31.00	\$80.00	\$95.00	\$120.00	\$98.33	\$28.00	\$30.00	\$35.00	\$31.00
Maximum												
Allowable Tenant Improvements	\$12.00	\$16.80	\$24.00	\$17.60	\$13.50	\$18.90	\$27.00	\$19.80	\$16.20	\$22.68	\$32.40	\$23.76
Allowance												

Maximum Commission

Notes:

- Borrower has required duty to employ Urban Partners to market retail and restaurant space. U.P. represents all retail in the Second Street District.
- 2. Retail and Restaurant commissions are calculated as 6% of base rent over the initial term of the lease. Split is 4% to tenant rep and 2% to landlord
- 3. 4.
- rep for Retail and Restaurant.

 Office commissions are calculated as 6% of gross rent over the initial term of the lease. Split is 4% to tellain tep and 2% to lar rep for Retail and Restaurant.

 Office commissions are calculated as 6% of gross rent over the initial term of the lease. Splits are the same as Retail.

 Additional Rent applies to the retail and restaurant leases. It is calculated as 6% of annual Gross Sales over the Natural Breakpoint.

 The Natural Breakpoint = Base Rent / 6%. The landlord will then receive 6% of all annual Gross Sales in excess of the Natural Breakpoint. For example: Base Rent = \$30 psf. Natural Breakpoint = \$30 / 6% = \$500 psf. Gross Sales = \$800 psf. Landlord will receive additional rent on top of Base Rent equal to \$18 psf = (\$800 - \$500) * 6%. Total Rent = Base Rent + Percentage Rent = \$30 + \$18 = \$48 psf.
- Commissions are not payable on Percentage Rent.

 Lease terms not in compliance with this Exhibit or the Amended and Restated Construction Loan Agreement require Lender's prior written consent.

 Any leasing commissions payable to parties affiliated with Borrower shall require Lender's prior written consent.

 Each Lease to a tenant affiliated with the Borrower requires Lender's prior written consent.
- 6. 7.

Exhibit Q - Page 1

Restaurant

EXHIBIT R

[Reserved]

Exhibit R - Page 1

EXHIBIT S

Form of Fourth Estoppel and Agreement from City of Austin

[See Attached]

Exhibit S- Page 1

EXHIBIT T

Due Diligence Materials Provided to Lender

[See Attached]

Exhibit T- Page 1

Exhibit 10.2

PROMISSORY NOTE

\$30,000,000.00 31,2010

FOR VALUE RECEIVED, CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company ("Maker"), does hereby promise to pay to the order of HUNTER'S GLEN/FORD INVESTMENTS I LLC ("Payee"), at 200 Crescent Court, Suite 1350, Dallas, Texas 75201, or at such other place as the holder hereof may from time to time designate in writing, in lawful money of the United States, the principal sum of THIRTY MILLION AND NO/100 DOLLARS (\$30,000,000.00), or as much thereof as is advanced on the loan evidenced hereby, with interest thereon as provided in this Note.

- 1. Certain Definitions. For the purposes hereof, the terms set forth below shall have the following meanings:
- (a) "Applicable Law" shall mean (i) the laws of the United States of America applicable to contracts made or performed in the State of Texas, now or at any time hereafter prescribing maximum rates of interest or eliminating maximum rates of interest on loans and extensions of credit, (ii) the laws of the State of Texas including, without limitation, Chapter 303 of the Texas Finance Code, as the same may be amended from time to time ("Chapter 303"), now or at any time hereafter prescribing or eliminating maximum rates of interest on loans and extensions of credit, and (iii) any other laws at any time applicable to contracts made or performed in the State of Texas which permit a higher interest rate ceiling hereunder. Notwithstanding the foregoing, in no event shall Chapter 346 of the Texas Finance Code (which regulates certain revolving loan accounts and revolving tri-party accounts) apply to this Note.
- (b) "Deed of Trust" shall mean that certain Second Lien Deed of Trust, Assignment of Rents, Security Agreement and Fixture Filing, dated of even date herewith, executed by Maker for the benefit of Payee, covering the Property as security for this Note and certain other indebtedness of Maker to Payee.
 - (c) "Final Maturity Date" shall mean March 31, 2012.
 - (d) "Fixed Rate" shall mean seventeen and one-half percent (17.5%) per annum.
- (e) "Highest Lawful Rate" shall mean at the particular time in question the maximum rate of interest which, under Applicable Law, Payee is then permitted to charge Maker on this Note. If the maximum rate of interest which, under Applicable Law, Payee is permitted to charge Maker on this Note shall change after the date hereof, the Highest Lawful Rate shall be automatically increased or decreased, as the case may be, from time to time as of the effective date of each change in the Highest Lawful Rate without notice to Maker. For purposes of determining the Highest Lawful Rate under the Applicable Law of the State of Texas, the applicable rate ceiling shall be the weekly ceiling described in and computed in accordance with the provisions of Chapter 303; provided, however, that in determining the Highest Lawful Rate, all fees and other charges contracted for, charged or received by Payee in connection with the loan evidenced by this Note which are either deemed interest under Applicable Law or required under Applicable Law to be deducted from the principal balance hereof to determine the rate of interest charged on this Note shall be taken into account. To the extent permitted by Applicable Law, Payee may from time to time substitute for the "weekly ceiling" referred to above any

ceiling under Chapter 303 or any other statute and revise the rate, index, formula or provision of law used to compute the rate hereunder as provided therein

- (f) "Intercreditor Agreement" shall mean that certain Subordination and Standstill Agreement of even date herewith among Maker, Payee and Senior Lender.
 - (g) "Loan Agreement" shall mean that certain Loan Agreement of even date herewith by and between Maker and Payee.
 - (h) "Loan Documents" has the meaning set forth in the Loan Agreement.
 - (i) "Project" has the meaning set forth in the Loan Agreement.
- (j) "Property" shall mean the real property, improvements and fixtures located in Travis County, Texas, and personal property which are more particularly described in the Deed of Trust.
 - (k) "Senior Lender" shall mean Beal Bank Nevada.
- (1) "Senior Loan" shall mean that certain construction loan from Senior Lender to Borrower in the stated principal amount of \$120,000,000.00.
- (m) "Senior Loan Agreement" shall mean that certain Amended and Restated Construction Loan Agreement dated October 21, 2009 between Senior Lender and Maker.

2. Calculation and Payment of Principal and Interest.

- (a) Subject to the provisions of Section 7 hereafter, interest on the unpaid principal balance hereof from time to time outstanding shall be computed at a rate equal to the lesser of (i) the Fixed Rate, and (ii) the Highest Lawful Rate.
- (b) Accrued and unpaid interest, computed as set forth in (a) above, or, if applicable, as set forth in Section 7 below, shall be due and payable annually on March 31, 2011 and on the Maturity Date; provided, however, that as long as there is no Event of Default by Maker on March 31, 2011 under the Loan Agreement, Maker may, at Maker's option, elect that accrued interest will not be paid on March 31, 2011, but shall be added to principal on an annual basis on each anniversary date of the date of the Note.
- (c) In accordance with the terms of Section 14.10 of the Loan Agreement, but subject to the terms of the Intercreditor Agreement, (i) as and when Residential Units (as defined in the Loan Agreement) are sold, as permitted by the Loan Documents, all Net Sales Proceeds (as defined in the Loan Agreement) from the sale of the Residential Unit sold, must be offered to Payce as a principal prepayment of this Note. In accordance with the terms of Section 4.5 of the Loan Agreement, but subject to the terms of the Intercreditor Agreement, as and when Residential Units are sold after the first anniversary of this Note, all Net Sales Proceeds from the sale of the Residential Unit sold must be paid to Payce as a required principal payment of this Note if Senior Lender declines to accept payment of such Net Proceeds as a payment on the Senior Loan.
- (d) The unpaid principal balance of and all accrued and unpaid interest upon this Note, including interest computed at the Fixed Rate and/or the Highest Lawful Rate, as hereinafter defined, whichever is applicable from time to time, are due and payable on the

maturity date hereof (the "Maturity Date") whether such Maturity Date is the Final Maturity Date or an accelerated Maturity Date, and Maker hereby agrees to pay all such amounts to Payee on such Maturity Date.

- (e) Interest on this Note shall be computed on the basis of three hundred sixty-five (365) day or three hundred sixty-six (366) day years, as the case may be.
- (f) If the date for any payment or prepayment hereunder falls on a day which is a Saturday, Sunday or holiday in the State of Texas on which banks are required or permitted by law to be closed, then for all purposes of this Note, the same shall be deemed to have fallen on the next following day, and such extension of time shall in such case be included in the calculation of interest.
- (g) All payments on this Note shall be applied first to the payment of any costs and expenses of Payee which are to be reimbursed by Maker as provided herein or in any of the other Loan Documents, then to the payment of accrued and unpaid interest and then to the payment of the principal balance hereof; provided, however, if an Event of Default, as hereinafter defined, is then in existence, payments on this Note shall be applied as Payee shall elect, in Payee's sole discretion.
- 3. Prepayment. This Note may not be prepaid, in whole or in part, except as permitted by the express terms of the Loan Agreement.
- 4. Waiver. Except to the extent otherwise expressly set forth in Paragraph 5 below, Maker and all sureties, endorsers, accommodation parties, guarantors and other parties now or hereafter liable for the payment of this Note, in whole or in part, hereby severally (i) waive demand, notice of demand, presentment for payment, notice of nonpayment, notice of default, protest, notice of protest, notice of intent to accelerate the Maturity Date, notice of acceleration of the Maturity Date, notice of dishonor and all other notices, and further waive diligence in collecting this Note, in taking action to collect this Note, in bringing suit to collect this Note, or in enforcing this Note or any of the security for this Note; (ii) agree to any substitution, subordination, exchange or release of any security for this Note or the release of any party primarily or secondarily liable for the payment of this Note; (iii) agree that Payee shall not be required to first institute suit or exhaust its remedies hereon against Maker or others liable or to become liable for the payment of this Note, or any installment hereof, made by agreement by Payee with any person now or hereafter liable for the payment of this Note, even if Maker is not a party to such agreement.

5. Events of Default

(a) Upon the happening of any of the following events (each an "Event of Default"), Payee, at its option, and without limitation of other rights Payee has under the Loan Documents, at law or in equity, may declare immediately due and payable the entire unpaid principal balance of this Note together with all interest accrued and owing hereon, plus any other sums payable at the time of such declaration pursuant to this Note or any other Loan Document, including, without limitation, the Loan Agreement and the Deed of Trust. Events of Default include the following:

(i) If Maker shall fail to pay any installment of principal and/or interest and/or other sums due under this Note as and when same becomes due and payable in accordance with the terms hereof and/or any other Loan Document, or if Maker shall

default in any other obligation under this Note, the Loan Agreement, the Deed of Trust and/or any other Loan Document which can be cured by the payment of money and if any such failure or default remains uncured upon the expiration of five (5) days following the date written notice thereof is given by or on behalf of Payee to Maker; provided, however, (x) Payee will not be obligated to give such written notice more than twice during any twelve (12) month period, and following the second such notice during a twelve (12) month period, any subsequent default or failure during the then current twelve (12) month period shall constitute an Event of Default without any notice given by or on behalf of Payee and (y) such written notice of default and five (5) day cure period provisions shall not apply to Maker's obligation to pay the balance due on this Note on the Maturity Date hereof (whether such Maturity Date is the Final Maturity Date or any accelerated Maturity Date effected pursuant to any of the Loan Documents);

(ii) The occurrence of any Event of Default, as defined in the Loan Agreement, or the occurrence of a default under any other Loan Document which remains uncured upon the expiration of any cure period applicable thereto as set forth in the Loan Document under which such default occurred.

(b) The failure to exercise the foregoing option to accelerate the Maturity Date of this Note upon the happening of one or more Events of Default shall not constitute a waiver of the right to exercise the same or any other option at any subsequent time during the continuance of an Event of Default, and no such failure shall nullify any prior exercise of any such option without the express written consent of Payee.

- 6. <u>Collateral</u>. This Note is secured, among other things, by the Deed of Trust, which contains provisions for the acceleration of the maturity hereof upon the happening of certain events.
- 7. Compliance with Law. All agreements between Maker and Payee, whether now existing or hereafter arising and whether written or oral, are hereby limited so that in no contingency, whether by reason of demand or acceleration of the Final Maturity Date or otherwise, shall the interest contracted for, charged, received, paid or agreed to be paid to Payee exceed the maximum amount permissible under Applicable Law. If, from any circumstance whatsoever, interest in regard to the loan evidenced by this Note would otherwise be payable to Payee in excess of the maximum amount permissible under Applicable Law; and if from any circumstance Payee shall be reduced to the maximum amount permissible under Applicable Law; and if from any circumstance Payee shall ever receive anything of value deemed interest in regard to the loan evidenced by this Note by Applicable Law; in excess of the maximum amount permissible under Applicable Law, an amount equal to the excessive interest shall be applied to the reduction of the principal balance hereof and not to the payment of interest, or if such excessive amount of interest exceeds the unpaid balance of principal hereof, such excess shall be refunded to Maker. All interest paid or agreed to be paid to Payee, to the extent permitted by Applicable Law, shall be amortized, prorated, allocated and spread throughout the full period (including any renewal or extension) until payment in full of the principal so that the interest hereon for such full period shall not exceed the maximum amount permissible under Applicable Law. Payee expressly disavows any intent to contract for, charge or receive interest in regard to the loan evidenced by this Note in an amount which exceeds the maximum amount permissible under Applicable Law. This paragraph shall control all agreements between Maker and Payee.
- 8. Attorneys' Fees and Costs. If an Event of Default shall occur, and in the event that thereafter this Note is placed in the hands of an attorney for collection, or in the event this Note is collected in whole or in part through legal proceedings of any nature, then and in any such case Maker promises to pay on demand by Payee all reasonable costs of collection, including, but not limited to,

reasonable attorneys' fees incurred by the holder hereof, on account of such collection, whether or not suit is filed.

- 9. <u>Cumulative Rights</u>. No delay on the part of the holder of this Note in the exercise of any power or right under this Note or under any other instrument executed pursuant hereto shall operate as a waiver thereof, nor shall a single or partial exercise of any power or right preclude other or further exercise thereof or the exercise of any other power or right. Enforcement by the holder of this Note of any security for the payment hereof shall not constitute any election by it of remedies so as to preclude the exercise of any other remedy available to it.
- 10. <u>Headings</u>. The paragraph headings used in this Note are for convenience of reference only, and shall not affect the meaning or interpretation of this Note.
- 11. Notices and Demands. Any notice or demand to be given or to be served upon Maker in connection with this Note must be in writing and shall be given by certified or registered mail, return receipt requested, properly addressed, with postage prepaid, addressed to Maker as follows:

Maker: CJUF II Stratus Block 21 LLC

c/o Stratus Properties Inc. 98 San Jacinto, Suite 220 Austin, Texas 78701

Attention: W.H. Armstrong III
Telephone: 512-478-6396
Facsimile: 512-478-5788

With a copy to: Armbrust & Brown, L.L.P.

100 Congress Avenue, Suite 1300

Austin, Texas 78701

Attention: Kenneth N. Jones, Esq. Telephone: 512-435-2312

Facsimile: 512-435-2312

And a copy to: Canyon-Johnson Urban Funds

2000 Avenue of the Stars, 11th Floor Los Angeles, California 90067

Attention: Head of Asset Management
Telephone: 310-272-1500

Telephone: 310-272-1500 Facsimile: 310-272-1537

And a copy to: DLA Piper US LLP

DLA Piper US LLP 550 South Hope Street, Suite 2300 Los Angeles, California 90071

Attention: Steven A. Fein, Esq. Telephone: 213-330-7772 Facsimile: 213-330-7572

or at such other address within the continental United States as Maker may designate from time to time by written notice given to the holder hereof. Any notice or demand will be given and deemed received as provided in the Loan Agreement.

- 12. Governing Law; Jurisdiction and Venue. This Note shall be governed by and construed in accordance with the laws of the State of Texas and the laws of the United States applicable to transactions in the State of Texas. Courts within the State of Texas shall have jurisdiction over any and all disputes between Maker and Payee, whether at law or in equity, and the proper venue for any such dispute shall be either the state courts located in Dallas County, Texas or the federal courts located in the Northern District of Texas, Dallas Division.
 - 13. Successors and Assigns. The term "Payee" shall include all of Payee's successors and assigns to whom the benefits of this Note shall inure.

14. WAIVER OF JURY TRIAL. MAKER AND PAYEE, BY ITS ACCEPTANCE OF THIS NOTE, HEREBY EXPRESSLY, VOLUNTARILY, KNOWINGLY AND IRREVOCABLY WAIVE ANY CONSTITUTIONAL OR OTHER RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN THE EVENT OF LITIGATION CONCERNING ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION (A) ARISING UNDER THIS NOTE, THE PARTIES' PERFORMANCE HEREUNDER OR ANY OTHER LOAN DOCUMENT, OR (B) IN ANY WAY CONNECTED WITH, OR RELATED OR INCIDENTIAL TO, THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO ANY LOAN DOCUMENT RELATED IN ANYWAY WHATSOEVER TO THE SUBJECT MATTER OF THIS NOTE; AND IN ANY CASE, WHETHER NOW OR HEREAFTER ARISING AND WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE. ANY PARTY MAY FILE AN ORIGINAL COUNTERPART OR COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

[Remainder of page intentionally left blank. Signature page(s) to follow.]

IN WITNESS WHEREOF, this Note has been executed and delivered under seal on the date first set froth above.

MAKER:

CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company

By:	Stratus Block 21 Investments, L.P., a Texas limited partnership, Manager						
	Ву:	Stratus Block 21 Investmen	its GP, L.L.C., a	Texas limited liabili	ity compa	ny, its Gen	eral Partner
		By: _/s/ Erin D. Pickens Erin D. Pickens Senior Vice President					
By:	CJUF	I Block 21 Member, LLC, a D	elaware limited	liability company,	Member		
By:	Cany	on-Johnson Urban Fund II, L.I	P., a Delaware lin	nited partnership, M	Iember		
	Ву:	Canyon-Johnson Realty A Partner	advisors II LLC	, a Delaware limit	ed liabili	ty compar	ny, General
		B y Tumer K. Robert Tumer Managing Partner	: <u></u>		/s/	K.	Robert
		(Signature page to Promissor	ry Note)				

PROFITS INTEREST AGREEMENT

THIS PROFITS INTEREST AGREEMENT (this "Agreement") is made and entered into as of March ___, 2010 (the "Effective Date"), between CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company ("Borrower"), and HUNTER'S GLEN/FORD INVESTMENTS I LLC, a Texas limited liability company ("Lender").

RECITALS

- 1. Lender has, contemporaneously herewith, made a loan (the "Loan") to Borrower in the original principal amount of \$30,000,000, pursuant to that certain Loan Agreement of even date herewith (as modified, amended, renewed, extended, and restated from time to time, the "Loan Agreement") and as evidenced by that certain Promissory Note of even date herewith executed by Borrower and payable to the order of Lender (as modified, amended, renewed, extended, and restated from time to time, the "Note").
- 2. The Loan is made for the purpose of financing the construction by Borrower of certain buildings and improvements on the real property that is located in Travis County, Texas and is more particularly described on <u>Exhibit "A"</u> attached hereto (the "Land;" together with all buildings and improvements now or hereafter located thereon are hereinafter referred to as the "Property").
- 3. The Note and the obligations of Borrower under the Loan Agreement and the other Loan Documents are secured by, among other things, a second priority lien and security interest on the Property.
- 4. As part of the consideration for Lender's agreement to make the Loan, Borrower has agreed to grant to Lender the Lender's Profits Interest (as herein defined) upon the terms and conditions set forth herein.

$\underline{AGREEMENT}$

NOW, THEREFORE, KNOW ALL PERSONS BY THESE PRESENTS, that for and in consideration of the premises hereof, and other good and valuable consideration, the receipt and sufficiency of which are acknowledged and confessed, the parties hereto agree as follows:

- 1. <u>Recitals</u>. Each of the foregoing recitals is acknowledged to be true and correct. The foregoing recitals are incorporated herein as contractual provisions of this Agreement.
- 2 . <u>Definitions</u>. Capitalized terms used herein and not defined herein shall have the meaning assigned to them in the Loan Agreement. In addition, the following terms shall have the meanings set forth below.

"Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, or controlled by, or under common control with, such Person.

"Approved Operating Budget" means an operating budget for the Property that is prepared by Borrower for each calendar year during the term of this Agreement and is hereafter approved in writing by the holder of the Senior Loan (until the Senior Loan has been paid in full) and after the Senior Loan has been paid in full, then approved in writing by Lender.

"Approved Sale" means any sale or other conveyance of the Property (or a portion thereof) approved in writing by Lender or deemed approved by Lender if the holder of the Senior Loan has approved in writing.

"Capital Event" means any Approved Sale, any other sale or refinancing (from a lender other than Lender) approved by Lender, and any condemnation, collection of insurance proceeds, or other capital event occurring with respect to the Property or any portion thereof.

"Cash Flow" means, for any period, (a) all receipts, revenues, income, rents, issues, profits, accounts, proceeds, loan proceeds for interest carry or operating expenses, and expense and other reimbursements received by Borrower from or in connection with the ownership, development, operation, or leasing of the Property, less (b) all reasonable and necessary costs and expenses incurred by Borrower as a result of, or in connection with, the ownership, development, operation, management, or leasing of the Property and in accordance with the Approved Operating Budget, less (c) reserves set aside in accordance with the Approved Operating Budget, less (d) payment of interest and principal on any debt secured by the Property (excluding, however, any interest or principal that is not required to be paid out of Cash Flow), less (e) deposits required for taxes, impositions, insurance and operating expenses under the terms of any loan documents for any debt secured by the Property.

"Lender's Cash Flow Profits Interest" means an amount equal to the product of (a) ninety-five percent (95%) times (b) Cash Flow.

"Lender's Profits Interest" means, without duplication, the sum of (a) Lender's Cash Flow Profits Interest, and (b) Lender's Second Tier Profits Interest; provided that (i) Lender's Profits Interest shall not exceed the Maximum Profits Interest and (ii) Lender's Profits Interest shall not be payable until after the first anniversary of the Effective Date and after the Senior Loan and the Loan have been paid in full, except as otherwise expressly permitted in this Agreement.

"Lender's Second Tier Profits Interest" means an amount equal to the product of (a) ninety-five percent (95%) times (b) Second Tier Profits.

"Maturity Date" means the Maturity Date as defined in the Note.

"Maximum Profits Interest" means the amount of \$750,000.00 if the Lender's Profits Interest is paid on the first anniversary of the Effective Date. If paid after the first anniversary of

the Effective Date, then the Maximum Profits Interest will be increased each full or partial month thereafter by the amount of \$62,500.00 until the Lender's Profits Interest is paid in full.

"Net Proceeds" means:

- (a) with respect to a Capital Event that is an Approved Sale, (1) the sum of (A) the total cash consideration paid in connection with the sale of the Property and (B) the then outstanding principal balance of, and all accrued interest on, any debt secured by the Property that is assumed or taken subject to by the transferee less (2) the sum of (A) customary and reasonable selling and closing costs (including, without limitation, title insurance premiums, survey costs, attorneys' fees, transfer taxes, brokerage commissions, tax, insurance, rent, expense, other prorations and other costs Borrower is required to pay under the purchase and sale agreement in connection with such sale of the Property), (B) the outstanding principal balance of, and all accrued interest on and prepayment penalties, if any, on, any debt secured by the Property (excluding, however, any principal and/or interest that is not required to be paid to satisfy such debt as a result of such Capital Event), and (C) funds placed in escrow by Borrower for the purpose of making repairs or for other purposes that are a condition to the closing of the sale of the Property;
- (b) with respect to a Capital Event that is a refinancing secured by a lien on the Property, the gross cash proceeds of such refinancing payable to Borrower less customary and reasonable expenses incurred in obtaining and closing such refinancing (including, without limitation, title insurance premiums, survey costs, attorneys' fees, transfer taxes, brokerage commissions, tax, insurance, rent, expense, other prorations and other reasonable costs Borrower is required to pay under such refinancing);
- (c) with respect to a Capital Event that is the receipt of proceeds from condemnation or casualty, the gross cash proceeds thereof received by Borrower less (i) amounts required to restore the Property, (ii) amounts required to be paid to the holder of the Senior Loan under the Senior Loan Documents and to Lender under the Deed of Trust, and (iii) amounts reasonably incurred by Borrower in the collection of the condemnation award or casualty insurance proceeds; and
- (d) with respect to all other Capital Events, the gross cash proceeds thereof received by Borrower less amounts reasonably incurred by Borrower in the collection of such proceeds.

"Property" has the meaning assigned to it in the Recitals hereof.

"Second Tier Profits" means:

(a) all Net Proceeds derived from all Capital Events (some of which require approval of Lender as provided herein); minus

- (b) to the extent not deducted in the calculation of Cash Flow, the aggregate of the amount of any outstanding principal of, and any unpaid interest on, the Senior Loan (including any prepayment fee or premium) and the Loan that is required to be paid out of Cash Flow.
- 3. Profits Interest. Borrower hereby grants, transfers, assigns, and conveys to Lender the Lender's Profits Interest, which grant and interest shall continue until the final payment of the Lender's Profits Interest, if any, is made to Lender as required hereby. Borrower and Lender acknowledge and agree that, except as otherwise expressly permitted in this Agreement, the Lender's Profits Interest shall not be paid to Lender unless and until (a) the Senior Loan has been paid in full, Senior Lender consents to and approves of the payment of the Lender's Profits Interest to Lender or Senior Lender declines to accept payment under the Senior Loan and allows the payment of the Lender's Profits Interest to Lender and (b) the Loan has been paid in full.
- 4. Option to Purchase. Lender hereby grants to Borrower an option for Borrower to purchase (with funds from any source) the Lender's Profits Interest and Lender's rights under this Agreement for an amount equal to the Maximum Profits Interest calculated as of the closing date on which Borrower purchases the Lender's Profits Interest pursuant to this Section 4 (the "Purchase Price"). Borrower may not exercise the option set forth in this Section 4 until the later to occur of the first anniversary of the Effective Date and the payment in full of the Loan. Borrower shall deliver to Lender written notice that Borrower has elected to exercise the option set forth in this Section 4 and setting the closing date for Borrower's purchase (which closing date shall be at least ten (10) days but not more than thirty (30) days after Borrower's written notice to Lender that Borrower has elected to exercise such option). At the closing, Borrower shall pay to Lender in cash an amount equal to the Purchase Price and Lender shall assign to Borrower, without representation, warranty or recourse, all of Lender's right, title and interest in and to this Agreement and the Lender's Profits Interest.
- 5. Term of the Agreement/Payoff/Release. This Agreement shall be effective on the Effective Date hereof and shall continue until the earlier of the date that Borrower has paid to Lender all of the Lender's Profits Interest or the sale of all of the Property by Borrower. Notwithstanding any other provision in this Agreement to the contrary, upon payment in full of the Loan and the Lender's Profits Interest from any source or sources, this Agreement will automatically terminate and be of no further force or effect. Upon such termination and written request of Borrower, Lender will execute and deliver to Borrower a full and complete release of all of Lender's rights under this Agreement, including a release of all liens and security interests, in form reasonably acceptable to Borrower and Lender.
 - 6. Reporting and Accounting.
- (a) <u>Accounting</u>. All accounts, books and records of Borrower relating to the Property shall be kept in accordance with the tax basis method, consistently applied.
 - (b) <u>Fiscal Year</u>. The fiscal year of Borrower shall be the calendar year.

- (c) <u>Statements.</u> Borrower shall timely deliver or cause to be delivered to Lender each of the financial statements required under the Loan Agreement.
- (d) Access to Books of Account. Lender shall have the right at all reasonable times during business hours, upon reasonable notice and subject to the rights of tenants, if any, of the Property, to examine and make copies of or extracts from the books of account of Borrower and the Property and such other information regarding the Property as it may desire.

Distributions and Losses.

- (a) <u>Distributions</u>. The amounts, if any, which may be payable for the Lender's Profits Interest granted to Lender hereby shall be payable to Lender, subject to compliance with the Senior Loan and the Intercreditor Agreement, (i) on a monthly basis upon receipt by Borrower of Cash Flow and (ii) upon the occurrence of a Capital Event.
- (b) <u>Losses</u>. Should the Property at any time after the date hereof incur an operating deficit or other loss, Borrower shall advance to the Property an amount sufficient to cover such operating deficit or other loss. Advances by Borrower to the Property for such purposes shall be unsecured and shall not reduce the Lender's Profits Interest. Any loans by partners or members of Borrower to Borrower shall be subordinate to Lender's rights and interests hereunder and under all other Loan Documents, and shall be repaid only out of Borrower's share of Second Tier Profits, if any. Lender shall not share in and shall have no obligation with respect to any loss suffered by the Property.

8. General Provisions.

- (a) No Partnership or Member. Lender, by its acceptance hereof, does not become a partner with or a member of Borrower, and in no event shall Lender be liable for any of the debts, obligations, or liabilities of Borrower or any Affiliate of Borrower, or claim any of the tax benefits resulting from the ownership of the Property, including, without limitation, any depreciation or investment tax credit on all or any portion of the Property, nor is Lender liable for any contributions to Borrower. Lender's only interest created hereunder is the right to a portion of the profits of Borrower, if any, arising by reason of the Property as provided in this Agreement.
- (b) <u>Power and Authority</u>. Borrower represents and warrants that: (i) Borrower is a Delaware limited liability company validly formed and existing under the laws of the State of Delaware and validly qualified to transact business in the State of Texas; (ii) Borrower is duly authorized and has all necessary partnership power and authority to grant the Lender's Profits Interest; and (iii) the execution of this Agreement does not violate or conflict with the organizational documents of Borrower or any law, rule or regulation, or any other material agreement or instrument to which any such Person is a party or by which any such Person is bound.

- (c) <u>Security.</u> Borrower agrees from time-to-time to execute and deliver to Lender UCC-1 Financing Statements and other instruments reasonably requested by Lender to evidence or perfect the Lender's Profits Interest granted to Lender hereby.
- (d) <u>Lien</u>. The obligations of Borrower hereunder are secured by the liens and security interests securing the Loan, as they may from time-to-time be released as provided in the Loan Agreement and as they may from time-to-time be modified, renewed, extended, or increased.
- (e) <u>Notices</u>. Written notice required to be given to either party shall be deemed given if sent in accordance with the provisions of the Loan Agreement.
- (f) <u>Subject to Senior Loan</u>. Lender acknowledges and agrees that the payment of and performance under this Agreement is subject to and subordinate to the Senior Loan in accordance with the Intercreditor Agreement. If any performance under this Agreement would cause a default under the Senior Loan, then such performance will not be required under this Agreement until such time as it would not cause a default under the Senior Loan.
- Speculative and Contingent Rights. Borrower and Lender acknowledge that the receipt by Lender of any amounts which may be payable for the Lender's Profits Interest is subject to numerous risks and uncertainties, including, without limitation, the possibility of unanticipated costs in the ownership, operation, and leasing of the Property (including, without limitation, those resulting from unforeseen conditions or force majeure); the possibility of utility moratoria or materials or labor costs decreasing the Cash Flow or Second Tier Profits; and the possibility of a change in interest rates or the economy in general, or the addition of or improvements to competing properties impacting the leasing, value, or sale of properties of the same type as the Property. Borrower and Lender further acknowledge that Lender's receipt of any of the Lender's Profits Interest is contingent on the availability of Cash Flow and Second Tier Profits. To the extent the grant of the Lender's Profits Interest is deemed to be interest on the Loan, the value of such grant shall be determined as of the date of this Agreement, taking into account the speculative and contingent nature of the profits interest granted hereunder. It is expressly stipulated and agreed to be the intent of Borrower and Lender at all times to comply with applicable state law or applicable United States federal law (to the extent that it permits Lender to contract for, charge, take, reserve, or receive a greater amount of interest than under state law) and that this Section 9 shall control every other covenant and agreement in this Agreement, the Loan Agreement, and the documents referenced therein. If the applicable law (state or federal) is ever judicially interpreted so as to render usurious any amount called for under this Agreement, then it is Borrower's and Lender's express intent that all excess amounts theretofore collected by Lender be repaid to Borrower with interest thereon at the rate set forth in the Note, and the provisions of this Agreement immediately be deemed reformed and the amounts thereafter collectible hereunder reduced, without the necessity of the execution of any new document, so as to comply with the applicable law, but so as to permit the recovery of the fullest amount otherwise called for hereunder and thereunder. All sums paid or agreed to be paid to Lender for the use, forbearance, and detention of the indebtedness represented by the Loan Agreement and the Note shall, to the extent permitted by applicable law, be amortized, prorated, allocated, and spread throughout the full term of such indebtedness until payment in full so that

the rate or amount of interest on account of such indebtedness does not exceed the Maximum Rate from time-to-time in effect and applicable to the Loan for so long as the Loan remains outstanding.

- Choice of Forum; Consent to Service of Process and Jurisdiction. ANY SUIT, ACTION OR PROCEEDING AGAINST ANY BORROWER WITH RESPECT TO THIS AGREEMENT OR ANY JUDGMENT ENTERED BY ANY COURT IN RESPECT HEREOF, MAY BE BROUGHT IN THE COURTS OF THE STATE OF TEXAS, COUNTY OF TRAVIS, OR IN THE UNITED STATES COURTS LOCATED IN THE STATE OF TEXAS AS LENDER IN ITS SOLE DISCRETION MAY ELECT AND BORROWER HEREBY IRREVOCABLY SUBMITS TO THE NONEXCLUSIVE JURISDICTION OF SUCH COURTS FOR THE PURPOSE OF ANY SUCH SUIT, ACTION, OR PROCEEDING. BORROWER HEREBY IRREVOCABLY CONSENTS TO THE SERVICE OF PROCESS IN ANY SUIT, ACTION, OR PROCEEDING IN SAID COURT BY THE MAILING THEREOF BY LENDER BY REGISTERED OR CERTIFIED MAIL, POSTAGE PREPAID, TO BORROWER'S ADDRESS SHOWN IN THE LOAN AGREEMENT. NOTHING HEREIN SHALL AFFECT THE RIGHT OF LENDER TO BRING ANY ACTION OR PROCEEDING AGAINST BORROWER OR WITH RESPECT TO ANY OF ITS PROPERTY IN COURTS IN OTHER JURISDICTIONS. BORROWER HEREBY IRREVOCABLY WAIVES ANY OBJECTIONS WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY SUIT, ACTION, OR PROCEEDING ARISING OUT OF OR RELATING TO THIS AGREEMENT BROUGHT IN THE COURTS LOCATED IN THE STATE OF TEXAS, COUNTY OF TRAVIS, AND HEREBY FURTHER IRREVOCABLY WAIVES ANY CLAIM THAT ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN ANY SUCH COURT HAS BEEN BROUGHT IN ANY INCONVENIENT FORUM. ANY ACTION OR PROCEEDING BY ANY BORROWER AGAINST LENDER SHALL BE BROUGHT ONLY IN A COURT LOCATED IN TRAVIS COUNTY, TEXAS.
- 11. <u>Multiple Counterparts</u>. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same agreement, and any of the parties hereto may execute this Agreement by signing any such counterpart.
- 12. Entirety. This Agreement, the Note, and the other Loan Documents referred to herein embody the final, entire agreement among the parties hereto and supersede any and all prior commitments, agreements, representations, and understandings, whether written or oral, relating to the subject matter hereof and may not be contradicted or varied by evidence of prior, contemporaneous, or subsequent oral agreements or discussions of the parties hereto. There are no oral agreements among the parties hereto. The provisions of this Agreement and the other Loan Documents to which Borrower is a party may be amended or waived only by an instrument in writing signed by the party or parties against whom the same is to be enforced.
- 13. <u>Governing Law.</u> THIS AGREEMENT AND ALL ISSUES, CLAIMS, COUNTERCLAIMS AND OTHER MATTERS RELATING HERETO OR ARISING IN CONNECTION THEREWITH (WHETHER BASED UPON TORT, CONTRACT OR OTHERWISE), SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH

THE LAWS OF THE STATE OF TEXAS (WITHOUT CONSIDERATION OF ITS CONFLICTS OF LAWS RULES) AND THE APPLICABLE LAWS OF THE UNITED STATES OF AMERICA.

1 4 . <u>Waiver of Jury Trial</u>. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, BORROWER AND LENDER HEREBY IRREVOCABLY AND EXPRESSLY WAIVE ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM (WHETHER BASED UPON CONTRACT, TORT, OR OTHERWISE) ARISING OUT OF OR RELATING TO LENDER OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THE ACTIONS OF LENDER IN THE NEGOTIATION, ADMINISTRATION, OR ENFORCEMENT THEREOF.

THE WRITTEN LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES.

THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

[Signature pages follow]

IN WITNESS WHEREOF, we hereunto set our hands, and this Agreement shall be effective as of the date first above written.

		<u>LENDER</u> :		
		HUNTER'S GLEN/FO a Texas limited liabil	FORD INVESTMENTS I LLC, ility company	
STATE OF TE	EXAS	§ §		
COUNTY OF		§ §		
This Glen/Ford Inv	instrument was ack restments I LLC, a T	cnowledged before me Fexas limited liability co	on the day of, 2010, by, of Hecompany, on behalf of said limited liability company.	Hunter's
			Notary Public in and for the State of Texas	
My	Commission	Expires:		
Notary's Print	ed Name			
Profits Interes	t Agreement - Page	9		

BORROWER:

CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company Stratus Block 21 Investments, L.P., a Texas limited partnership, Manager By: By: Status Block 21 Investments GP, L.L.C., a Texas limited liability company, General Partner By: /s/ Erin D. Pickens Erin D. Pickens, Senior Vice President By: CJUF II Block 21 Member, LLC, a Delaware limited liability company, Member Canyon-Johnson Urban Fund II, L.P., a Delaware By: limited partnership, Member Canyon-Johnson Realty Advisors II LLC, a Delaware limited liability company, General Partner By: /s/ K. Robert Turner K. Robert Turner, Managing Partner STATE OF TEXAS § § § COUNTY OF This instrument was acknowledged before me on this ___ day of __ _, 2010, by Erin D. Pickens, Senior Vice President of Stratus Block 21 Investments GP, L.L.C., a Texas limited liability company, the General Partner of Stratus Block 21 Investments, L.P., a Texas limited partnership, Manager of CJUF II STRATUS BLOCK 21 LLC, a Delaware limited liability company, on behalf of said limited partnership and limited liability companies. Notary Public in and for the State of Texas Commission Expires: My Notary's Printed Name Profits Interest Agreement - Page 10

STATE OF TEXA	AS	§	
COUNTY OF		8 §	
Johnson Realty A	Advisors II LLC, a nip, as Member of 0	Delaware limited liability CJUF II Block 21 Member	this day of, 2010, by K. Robert Turner, the Managing Partner of Canyon ty company, in its capacity as General Partner of Canyon-Johnson Urban Fund II, L.P., a Delaware Imited liability company, as Member of CJUF II STRATUS BLOCK 21, LLC tted partnership and limited liability companies.
			Notary Public in and for the State of Texas
My	Commission	Expires:	
			Notary's Printed Name
Profits Interest A	greement - Page 1	1	

EXHIBIT "A"

LEGAL DESCRIPTION OF LAND

Lots 1 through 12, Block 21, of the Original City of Austin, Travis County, Texas, according to the map or plat of record in the General Land Office of the State of Texas, together with the area within the alley traversing said Block, which was vacated by Ordinance recorded under Document No. 1999086902 and described in Memorandum Designating the Vacation of a 20 foot wide alley on Block 21 and Block 22, in the City of Austin as recorded under Document No. 2004040650 of the Official Public Records of Travis County, Texas.

FOURTH MODIFICATION AND EXTENSION AGREEMENT

This FOURTH MODIFICATION AND EXTENSION AGREEMENT (this "Agreement") dated effective as of March 31, 2010 (the "Effective Date") by and between STRATUS PROPERTIES INC., a Delaware corporation ("Stratus"), STRATUS PROPERTIES OPERATING CO., L.P., a Delaware limited partnership ("SPOC"), CIRCLE C LAND, L.P., a Texas limited partnership ("Circle C"), and AUSTIN 290 PROPERTIES, INC., a Texas corporation ("Austin") (Stratus, SPOC, Circle C and Austin are sometimes referred to in this Agreement severally as "Borrower"), CALERA COURT, L.P., a Texas limited partnership ("Calera Court"), and COMERICA BANK ("Lender");

WITNESSETH:

WHEREAS, Borrower has executed and delivered to Lender, inter alia, (i) that certain Revolving Promissory Note dated as of September 30, 2005, payable to the order of Lender in the original principal sum of \$45,000,000.00, with interest and principal payable as therein provided, which note was amended by that certain (i) Modification and Extension Agreement (the "First Modification") dated as of May 30, 2006, executed by and among Borrower, Calera Court, Stratus JV and Lender, and recorded under Clerk's File No. 2006140557 of the Real Property Records of Travis County, Texas, (ii) Second Modification and Extension Agreement (the "Second Modification") dated as of May 30, 2007, executed by and among Borrower, Calera Court, Stratus JV and Lender, and recorded under Clerk's File No. 2007139303 of the Real Property Records of Travis County, Texas and (iii) Third Modification and Extension Agreement (the "Third Modification") dated as of May 30, 2008, executed by and among Borrower, Calera Court, Stratus JV and Lender, and recorded under Clerk's File No. 2008122886 of the Real Property Records of Travis County, Texas (said note, as amended by the First Modification, Second Modification and Third Modification, is herein called the "Note"); (ii) that certain Loan Agreement dated of even date with the Note between Borrower, Calera Court and Lender, which loan agreement was amended by the First Modification, Second Modification and Third Modification (said loan agreement, as amended by the First Modification, Second Modification and Third Modification, is herein called the "Loan Agreement"); (iii) that certain Deed of Trust, Security Agreement and Assignment of Rents dated of even date with the Note from Stratus to Melinda Chausse, Trustee, securing the payment of the Note, covering certain real and personal property described therein, recorded under Clerk's File No. 2005183345 of the Real Property Records of Travis County, Texas, which deed of trust was amended by the First Modification, Second Modification and Third Modification (said deed of trust, as amended by the First Modification, Second Modification and Third Modification, is herein called the "Stratus Deed of Trust"); (iv) that certain Deed of Trust, Security Agreement and Assignment of Rents dated of even date with the Note from Circle C to Melinda Chausse, Trustee, securing the payment of the Note, covering certain real and personal property described therein, recorded under Clerk's File No. 2005183344 of the Real Property Records of Travis County, Texas, which deed of trust was amended by the First Modification, Second Modification and Third Modification (said deed of trust, as amended by the First Modification, Second Modification and Third Modification, is herein called the "Circle C Deed of Trust"); (v) that certain Deed of Trust, Security Agreement and Assignment of Rents dated of even date with the Note from SPOC to Melinda Chausse, Trustee, securing the payment of the Note, covering certain real and

personal property described therein, recorded under Clerk's File No. 2005183343 of the Real Property Records of Travis County, Texas, which deed of trust was amended by the First Modification, Second Modification and Third Modification (said deed of trust, as amended by the First Modification, Second Modification and Third Modification, is herein called the "SPOC Deed of Trust"); (vi) that certain Deed of Trust, Security Agreement and Assignment of Rents dated of even date with the Note from Austin to Melinda Chausse, Trustee, securing the payment of the Note, covering certain real and personal property described therein, recorded under Clerk's File No. 2005183347 of the Real Property Records of Travis County, Texas, which deed of trust was amended by the First Modification, Second Modification and Third Modification (said deed of trust, as amended by the First Modification, Second Modification and Third Modification, is herein called the "Austin Deed of Trust"); (vii) that certain Deed of Trust, Security Agreement and Assignment of Rents dated of even date with the Note from Calera Court to Melinda Chausse, Trustee, securing the payment of the Note, covering certain real and personal property described therein, recorded under Clerk's File No. 2005183346 of the Real Property Records of Travis County, Texas, which deed of trust was amended by the First Modification, Second Modification and Third Modification (said deed of trust, as amended by the First Modification, Second Modification and Third Modification, is herein called the "Calera Court Deed of Trust"); (viii) that certain Deed of Trust, Security Agreement and Assignment of Rents dated of even date with the Note from Stratus JV to Melinda Chausse. Trustee, securing the payment of the Note, covering certain real and personal property described therein, recorded under Clerk's File No. 2005183348 of the Real Property Records of Travis County, Texas, which deed of trust was amended by the First Modification, Second Modification and Third Modification (said deed of trust, as amended by the First Modification, Second Modification and Third Modification, is herein called the "Stratus JV Deed of Trust") (the Stratus Deed of Trust, Circle C Deed of Trust, SPOC Deed of Trust, Austin Deed of Trust, Calera Court and Stratus JV Deed of Trust are herein collectively called the "Deed of Trust", and all of the property covered by the Deed of Trust is herein collectively called the "Mortgaged Property") (the Note, Loan Agreement, Deed of Trust, First Modification, Second Modification, Third Modification and all other documents executed by Borrower and/or any other party or parties evidencing or securing or otherwise in connection with the loans evidenced by the Note (the "Loan") being herein collectively called the "Loan Documents");

WHEREAS, the Note is due and payable on May 30, 2010, and Borrower has requested that Lender extend the term of the Note to May 30, 2012 and make certain other modifications to the Loan Documents, and Lender is willing to do so on the terms and conditions set forth below; and

WHEREAS, Lender is the owner and holder of the Note and Borrower is the owner of the legal and equitable title to the Mortgaged Property;

NOW, THEREFORE, for and in consideration of the mutual covenants contained herein and for other valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. <u>Defined Terms</u>. Capitalized terms used but not defined in this Agreement shall have the meaning given to such capitalized terms in the Loan Agreement.

- 2. Extension of Maturity Date. The maturity date of the Note is hereby extended to May 30, 2012 (the "Maturity Date"), and the liens, security interests, assignments and other rights evidenced by the Loan Documents are hereby renewed and extended to secure payment of the Note as extended hereby. Without limiting the foregoing, the term "Maturity Date" as used in the Note, Loan Agreement and other Loan Documents are likewise amended to mean and refer to "May 30, 2012".
- 3. Extension Fee. As consideration for the extension of the Maturity Date, and as a condition to the effectiveness of this Agreement and the extension of the Maturity Date, Borrower shall pay to Lender an extension fee in the amount of \$275,000.00 contemporaneously with the execution of this Agreement.
- 4. <u>Bifurcation of Loan</u>. Prior to the Effective Date, the Loan Documents provided that the Loan is a \$45,000,000 revolving credit facility. Notwithstanding anything to the contrary contained in the Loan Documents, from and after the Effective Date, the Loan shall be bifurcated into two (2) tranches, with the first being a \$35,000,000 revolving credit tranche (the "<u>Revolving Loan Tranche</u>"), and the second being a \$10,000,000 term loan tranche (the "<u>Term Loan Tranche</u>"). Borrower may borrow the proceeds of the Term Loan Tranche for the same purposes as provided under the Loan Documents for the Revolving Loan Tranche; however, and notwithstanding anything to the contrary contained in the Loan Documents, any amounts borrowed and repaid under the Term Loan Tranche may not be reborrowed by Borrower. In connection with each request for an Advance under the Loan Agreement, Borrower shall designate in such request whether such Advance is to be under the Revolving Loan Tranche or the Term Loan Tranche; provided, further, at such time as Borrower has borrowed an aggregate of \$10,000,000 under the Term Loan Tranche, Borrower shall not have any further right to borrow under the Term Loan Tranche. As of the Effective Date, (i) \$16,325,872.02 of principal has been advanced and is outstanding under the Revolving Loan Tranche, (ii) \$0 has been advanced and is outstanding under the Term Loan Tranche and (iii) a Letter of Credit in the face amount of \$2,933,077 has been issued and is outstanding under the Revolving Loan Tranche" and "Term Loan Tranche" and "Term Loan Tranche" and Term Loan Tranche are more specifically set forth in this Agreement.
 - 5. Modifications to the Note. From and after the Effective Date, the Note is hereby modified as follows:
 - (a) The definition of "Applicable Base Rate" in the Note is hereby amended and restated in its entirety to read as follows:
 - "Applicable Base Rate' shall mean, (1) with respect to the Revolving Loan Tranche, the lesser of (a) the Base Rate from time to time in effect <u>plus</u> two percent (2.0%) per annum, or (b) the Maximum Lawful Rate, but in no event shall the Applicable Base Rate ever be less than the Floor Rate applicable to the Revolving Loan Tranche, and (2) with respect to the Term Loan Tranche, the lesser of (x) the Base Rate from time to time in effect <u>plus</u> three percent (3.0%)

per annum, or (y) the Maximum Lawful Rate, but in no event shall the Applicable Base Rate ever be less than the Floor Rate applicable to the Term Loan Tranche. Fluctuations in the Applicable Base Rate shall become effective immediately, without necessity for any notice whatsoever."

(b) The definition of "Applicable LIBOR Rate" in the Note is hereby amended and restated in its entirety to read as follows:

"Applicable LIBOR Rate' shall mean, (1) with respect to the Revolving Loan Tranche, the lesser of (a) the rate of interest equal to the Adjusted LIBOR Rate in effect for the subject Interest Period plus four percent (4.0%) or (b) the Maximum Lawful Rate, but in no event shall the Applicable LIBOR Rate ever be less than the Floor Rate applicable to the Revolving Loan Tranche, and (2) with respect to the Term Loan Tranche, the lesser of (x) the rate of interest equal to the Adjusted LIBOR Rate in effect for the subject Interest Period plus five percent (5.0%) or (y) the Maximum Lawful Rate, but in no event shall the Applicable LIBOR Rate ever be less than the Floor Rate applicable to the Term Loan Tranche."

(c) The definition of "Floor Rate" in the Note is hereby amended and restated in its entirety to read as follows:

"Floor Rate' shall mean, (1) with respect to the Revolving Loan Tranche, six percent (6.0%) per annum, and (2) with respect to the Term Loan Tranche, seven percent (7.0%) per annum."

(d) The definition of "Interest Period" in the Note is hereby amended and restated in its entirety to read as follows:

"Interest Period" shall mean the period of time commencing on the Effective Date of any LIBOR Rate Tranche and ending on the numerically corresponding day in the first, second or third calendar month thereafter (as designated by written notice by Maker to Payee given consistent with the requirements of Section 2.6 or Section 2.7 of this Note). With respect to any Interest Period which commences on the last Business Day of a particular calendar month (or on any day for which there is no numerically corresponding day in the appropriate subsequent calendar month), such Interest Period shall end on the last Business Day of the appropriate subsequent calendar month. Any Interest Period which would otherwise extend beyond the Maturity Date shall expire as of the Maturity Date."

- (e) <u>Section 2.1</u> of the Note is hereby amended and restated as follows:
- "2.1 Interest Rate. Pursuant to the terms of this Note, the indebtedness evidenced hereby may collectively consist of either zero (0) or one (1) Base Rate Tranches under each of the Revolving Loan Tranche and the Term Loan Tranche, and any of zero (0), one (1), two (2), or three (3) LIBOR Rate Tranches under

each of the Revolving Loan Tranche and the Term Loan Tranche. Under no circumstances shall any Tranche at any point in time accrue interest at a rate in excess of the Maximum Lawful Rate."

- (f) Section 3.1 of the Note is hereby amended and restated as follows:
- "3.1 <u>Payment Schedule.</u> The amounts advanced by Payee under the Revolving Loan Tranche and repaid by Maker may be re-borrowed as provided in the Loan Agreement. The amounts advanced by Payee under the Term Loan Tranche and repaid by Maker may not be re-borrowed, and shall permanently reduce the amount of the loan evidenced by this Note and the other Loan Documents. The amounts advanced under this Note shall be due and payable as follows:
 - (a) With respect to amounts advanced under the Revolving Loan Tranche, Maker shall pay Payee all then accrued but unpaid interest (including without limitation, all interest accruing under any Base Rate Tranche and under any LIBOR Rate Tranche) on the fifth (5th) day of each calendar month until the Maturity Date.
 - (b) With respect to amounts advanced under the Term Loan Tranche, Maker shall pay Payee all then accrued but unpaid interest (including without limitation, all interest accruing under any Base Rate Tranche and under any LIBOR Rate Tranche) on the fifth (5th) day of each calendar month until the Maturity Date. In addition to the foregoing monthly interest payments, commencing on June 5, 2011 and continuing regularly on the fifth (5th) day of each calendar quarter thereafter until the Maturity Date (i.e., each September 5, December 5, March 5 and June 5), Maker shall make a principal payment of \$500,000 to Payee under the Term Loan Tranche, but only to the extent of any principal then outstanding under the Term Loan Tranche; provided, further, if there is not sufficient principal then outstanding under the Term Loan Tranche on account of Maker not previously requesting disbursements of the Term Loan Tranche, then the amount of the Term Loan Tranche shall be reduced by the amount of principal that Maker would have been required to pay to Payee had the Loan proceeds been previously advanced under the Term Loan Tranche.
 - (c) In addition to the foregoing principal payments due under the Term Loan Tranche, any distributions received by Maker from its investment in CJUF II Stratus Block 21 LLC (the "Block 21 Entity") shall, after repayment of any amounts due to the senior lender and mezzanine lender having a lien or security interest on the Block 21 project (being the W Austin Hotel and Condominium Residences) and/or the Block 21 Entity, be paid to Payee and applied against the Term Loan Tranche to the extent of any outstanding principal thereunder; provided, further, if there is not sufficient principal then outstanding under the Term

Loan Tranche on account of Maker not previously requesting a disbursement of the Term Loan Tranche, then the amount of the Term Loan Tranche shall be reduced by the amount of excess distributions received by Maker (after repayment of the outstanding principal under the Term Loan Tranche) and the amount of the Term Loan Tranche available to be disbursed to Maker shall be reduced by the amount of excess distributions so received by Maker.

- (d) The outstanding principal balance hereof and any and all accrued but unpaid interest thereon, including without limitation, all amounts outstanding under the Revolving Loan Tranche and the Term Loan Tranche, shall be due and payable in full on the Maturity Date or upon earlier maturity hereof, whether by acceleration or otherwise."
- (g) The provisions of the Note regarding Advances, interest rate options, borrowings and similar issues shall apply both to the Revolving Loan Tranche and the Term Loan Tranche. For example, Borrower may elect to have portions of both the Revolving Loan Tranche and the Term Loan Tranche bear interest at the Applicable Base Rate and the Applicable LIBOR Rate.
- 6. Modifications to the Loan Agreement. From and after the Effective Date, the Loan Agreement is hereby modified as follows:
- (a) Notwithstanding anything to the contrary contained in <u>Section 4.19</u> of the Loan Agreement, Lender shall have the right to update the Appraisals for the Primary Collateral or any other property included in the borrowing base on an annual basis (i.e., once per year), but in any event, Lender shall obtain updated Appraisals at least one (1) time every two (2) years for the Primary Collateral.
 - (b) <u>Section 4.20(a)</u> of the Loan Agreement is hereby amended and restated in its entirety to read as follows:
 - "(a) Maintain a Tangible Net Worth at all times of not less than \$120,000,000."
 - (c) Section 5.3 of the Loan Agreement is hereby amended by adding subparagraph (d) thereto, which reads as follows:
 - "(d) guaranties executed by Stratus in connection with (i) the senior loan made by Beal Bank Nevada to CJUF II Stratus Block 21 LLC (the "Block 21 Entity") in connection with the development of the W Austin Hotel and Condominium Residences, herein referred to as the "Block 21 Project") and (ii) the mezzanine loan made by Hunter's Glen/Ford Investments 1 LLC or another affiliate of Gerald Ford to the Block 21 Entity in connection with the Block 21 Project (collectively, the "Block 21 Stratus Guaranties")."
 - (d) Section 5.4(c) of the Loan Agreement is hereby amended and restated in

its entirety to read as follows:

"except for the Guaranties of Non-Recourse Carve-Out Liabilities, the Block 21 Stratus Guaranties and any guaranties for the benefit of Bank with regard to other loans to Subsidiaries of Borrowers or any other Loan Party (except as expressly provided below), contingent liabilities of Borrowers on a Consolidated basis at any one time not to exceed \$20,000,000.00, which \$20,000,000 limitation shall be inclusive of the guaranties previously executed by Stratus (or its affiliates) in connection with (x) the loan made by Bank to Crestview Station RTB Land, LP, Crestview Station LLR Land, LP and Crestview Station DS Land, LP on the Crestview Station project and (y) the loan made by United Heritage Credit Union to 5700 Slaughter Lane, L.L.C. on the 5700 Slaughter Lane retail project;"

(e) The definition of "Borrowing Base Limitation" in Addendum 1 of the Loan Agreement is hereby amended and restated in its entirety to read as follows:

"'Borrowing Base Limitation' shall mean the sum of:

- (a) thirty-five percent (35%) of the fair market value of the Primary Collateral which is unimproved real property (except for any portions which are covered by other subsections in the definition of Borrowing Base Limitation set forth below), as indicated by Appraisals delivered to and accepted by Bank pursuant to Section 4.19 hereof;
- (b) sixty percent (60%) of the fair market value of the Developed Lots, as indicated by Appraisals delivered to and accepted by Bank pursuant to Section 4.19 hereof;
- (c) with respect to portions of the Land which are currently being developed into single-family residential lots (but which are not yet fully Developed Lots), the lesser of (x) sixty percent (60%) of the fair market value of such Land (as if improved and developed), as indicated by Appraisals delivered to and accepted by Bank pursuant to Section 4.19 hereof or (y) an amount equal to the sum of (1) the discounted fair market value of the Land plus (2) the hard and soft cost of all improvements made to the Land as of the date of determination of the Borrowing Base Limitation;
 - (d) fifty percent (50%) of the Credit Bank Value;
 - (e) forty-five percent (45%) of the MUD Reimbursables Value;
- (f) sixty percent (60%) of the appraised value of the Calera Court Spec Houses (it being acknowledged that there are no further Calera Court Model Houses or Calera Court Pre-Sold Houses)."
- (f) The definition of "Loan" in Addendum 1 of the Loan Agreement is hereby amended and restated in its entirety to read as follows:

"Loan" shall mean, collectively, the Revolving Loan Tranche and the Term Loan Tranche made, or to be made, by Bank to or for the credit of Borrowers in one or more Advances not to exceed at any one time the aggregate Maximum Loan Amount applicable to the Revolving Loan Tranche and the Term Loan Tranche, as the case may be, pursuant to the Loan Terms, Conditions and Procedures Addendum.

- (g) The definition of "Maximum Loan Amount" in Addendum 1 of the Loan Agreement is hereby amended and restated in its entirety to read as follows:
 - "Maximum Loan Amount' shall mean, (1) with respect to the Revolving Loan Tranche, the lesser of (a) \$35,000,000.00 or (b) the Borrowing Base Limitation, (2) with respect to the Term Loan Tranche, the lesser of (c) \$10,000,000.00, less any principal payments made by Borrower to Bank under the Term Loan Tranche or (d) the Borrowing Base Limitation and (3) with respect to the entire Loan (including the Revolving Loan Tranche and the Term Loan Tranche), the lesser of (e) \$45,000,000, less any principal payments made by Borrower to Bank under the Term Loan Tranche or (f) the Borrowing Base Limitation."
- (h) All Advances under the Revolving Loan Tranche and the Term Loan Tranche shall be in accordance with the terms and condition set forth in Addendum 2 of the Loan Agreement; provided, however, (i) in no event shall the outstanding principal balance of the Loan under the Term Loan Tranche at any one time exceed the Maximum Loan Amount applicable to the Term Loan Tranche, (ii) in no event shall the outstanding principal balance of the Loan under the Revolving Loan Tranche plus any Letter of Credit Liabilities at any one time exceed the Maximum Loan Amount applicable to the Revolving Loan Tranche, (iii) in no event shall the aggregate outstanding principal balance of the Loan under the Term Loan Tranche, the Revolving Loan Tranche and any Letter of Credit Liabilities at any time exceed the lesser of (x) the Borrowing Base Limitation or (y) \$45,000,000, reduced by any principal payments made by Borrower to Lender under the Term Loan Tranche, (iv) any outstanding Letter of Credit Liabilities shall reduce the amount available to be borrowed by Borrower under the Revolving Loan Tranche by a like amount and (v) any principal payments made by Borrower to Lender under the Term Loan Tranche may not under any circumstances be reborrowed. Notwithstanding anything to the contrary contained in the Loan Agreement, Letters of Credit may only be issued under the Revolving Loan Tranche, and in no event shall the Letter of Credit Liabilities and the sum of the outstanding principal balance of the Revolving Loan Tranche ever exceed the Maximum Loan Amount applicable to the Revolving Loan Tranche.
- (i) Borrower shall continue pay to Lender an unused commitment fee on the entire amount of the Loan (i.e., a fee equal to .125% per annum multiplied by the difference between (i) the aggregate Maximum Loan Amount of the Revolving Loan Tranche and the Term Loan Tranche and (ii) the aggregate outstanding principal balance of the Revolving Loan Tranche and the Term Loan Tranche, in accordance with the terms of Section 1.6(b) of Addendum 2 of the Loan Agreement.

- (j) Notwithstanding anything to the contrary contained in <u>Section 2.4</u> of <u>Addendum 2</u> of the Loan Agreement, Borrower may borrow up to, but not in excess of, \$9,000,000 of Loan proceeds under the Revolving Loan Tranche in the aggregate to make a principal payment due to FAAM on December 31, 2011 under the FAAM Loan.
- (k) In connection with and as consideration for the extension of the Maturity Date, Borrower has agreed to pledge to Lender, and a grant a security interest to Lender in, any Borrower's right to receive any distributions from CJUF II Stratus Block 21 LLC (the "Block 21 Entity", and being the owner of the W Austin Hotel and Condominium Residences), pursuant to a separate Security Agreement dated of even date with this Agreement, which Security Agreement shall be deemed included in the term "Security Agreements" in Addendum 1 of the Loan Agreement. Borrower further covenants and agrees that any distributions received by any Borrower from its investment in the Block 21 Entity shall, after repayment of any amounts due to the senior lender and mezzanine lender having a lien or security interest on the Block 21 project and/or the Block 21 Entity, be paid to Lender and applied against the Term Loan Tranche to the extent of any outstanding indebtedness thereunder; provided, further, if there is not sufficient principal then outstanding under the Term Loan Tranche on account of Borrower not previously requesting a disbursement of the Term Loan Tranche, then the amount of the Term Loan Tranche available to be disbursed to Borrower shall be reduced by the amount of excess distributions received by Borrower (after repayment of the outstanding principal under the Term Loan Tranche) and the amount of the Term Loan Tranche shall be reduced by the amount of excess distributions so received by Borrower.
- 7. Representations and Warranties. Borrower hereby represents and warrants that (a) Borrower is the sole legal and beneficial owner of the Mortgaged Property; (b) Borrower is duly organized and legally existing under the laws of the State of Texas; (c) the execution and delivery of, and performance under this Agreement are within Borrower's power and authority without the joinder or consent of any other party and have been duly authorized by all requisite action and are not in contravention of law or the powers of Borrower's articles of incorporation and bylaws; (d) this Agreement constitutes the legal, valid and binding obligations of Borrower enforceable in accordance with its terms; (e) the execution and delivery of this Agreement by Borrower do not contravene, result in a breach of or constitute a default under any deed of trust, loan agreement, indenture or other contract, agreement or undertaking to which Borrower is a party or by which Borrower or any of its properties may be bound (nor would such execution and delivery constitute such a default with the passage of time or the giving of notice or both) and do not violate or contravene any law, order, decree, rule or regulation to which Borrower is subject; and (f) to the best of Borrower's knowledge there exists no uncured default under any of the Loan Documents. Borrower agrees to indemnify and hold Lender harmless against any loss, claim, damage, liability or expense (including without limitation reasonable attorneys' fees) incurred as a result of any representation or warranty made by it herein proving to be untrue in any respect.
- 8. Further Assurances. Borrower, upon request from Lender, agrees to execute such other and further documents as may be reasonably necessary or appropriate to consummate the transactions contemplated herein or to perfect the liens and security interests intended to

secure the payment of the loan evidenced by the Note.

- 9. <u>Default; Remedies</u>. If Borrower shall fail to keep or perform any of the covenants or agreements contained herein or if any statement, representation or warranty contained herein is false, misleading or erroneous in any material respect, Borrower shall be deemed to be in default under the Deed of Trust and Lender shall be entitled at its option to exercise any and all of the rights and remedies granted pursuant to the any of the Loan Documents or to which Lender may otherwise be entitled, whether at law or in equity.
- 10. Endorsement to Mortgagee Title Policy. Contemporaneously with the execution and delivery hereof, Borrower shall, at its sole cost and expense, obtain and deliver to Lender an Endorsement of the Mortgagee Title Policy insuring the lien of the Deed of Trust, under Procedural Rule P-9b(3) of the applicable title insurance rules and regulations, in form and content acceptable to Lender, stating that the company issuing said Mortgagee Title Policy will not claim that policy coverage has terminated or that policy coverage has been reduced, solely by reason of the execution of this Agreement.
- 11. Ratification of Loan Documents. Except as provided herein, the terms and provisions of the Loan Documents shall remain unchanged and shall remain in full force and effect. Any modification herein of any of the Loan Documents shall in no way adversely affect the security of the Deed of Trust and the other Loan Documents for the payment of the Note. The Loan Documents as modified and amended hereby are hereby ratified and confirmed in all respects. All liens, security interests, mortgages and assignments granted or created by or existing under the Loan Documents remain unchanged and continue, unabated, in full force and effect, to secure Borrower's obligation to repay the Note.
- 12. Liens Valid; No Offsets or Defenses. Borrower hereby acknowledges that the liens, security interests and assignments created and evidenced by the Loan Documents are valid and subsisting and further acknowledges and agrees that there are no offsets, claims or defenses to any of the Loan Documents.
- 13. Merger; No Prior Oral Agreements. This Agreement supersedes and merges all prior and contemporaneous promises, representations and agreements. No modification of this Agreement or any of the Loan Documents, or any waiver of rights under any of the foregoing, shall be effective unless made by supplemental agreement, in writing, executed by Lender and Borrower. Lender and Borrower further agree that this Agreement may not in any way be explained or supplemented by a prior, existing or future course of dealings between the parties or by any prior, existing, or future performance between the parties pursuant to this Agreement or otherwise.
- 14. Notices. Any notice or communication required or permitted hereunder or under any of the Loan Documents shall be given in writing and sent in the manner required under the Loan Agreement. Notwithstanding the foregoing, the address for notices to Lender under the Loan Documents is hereby amended to the following:

to Lender: Comerica Bank

300 W. Sixth Street, Suite 1300

Austin, Texas 78701

Attention: Commercial Real Estate, Sterling J. Silver

With a copy to: Thompson & Knight LLP

1722 Routh Street, Suite 1500 Dallas, Texas 75201-2533 Attention: Mark M. Sloan

- 15. <u>Costs and Expenses</u>. Contemporaneously with the execution and delivery hereof, Borrower shall pay, or cause to be paid, all costs and expenses incident to the preparation hereof and the consummation of the transactions specified herein, including without limitation title insurance policy endorsement charges, recording fees and fees and expenses of legal counsel to Lender.
- Release of Lender. Borrower hereby releases, remises, acquits and forever discharges Lender, together with its employees, agents, representatives, consultants, attorneys, fiduciaries, servants, officers, directors, partners, predecessors, successors and assigns, subsidiary corporations, parent corporations, and related corporate divisions (all of the foregoing hereinafter called the "Released Parties"), from any and all actions and causes of action, judgments, executions, suits, debts, claims, demands, liabilities, obligations, damages and expenses of any and every character, known or unknown, direct and/or indirect, at law or in equity, of whatsoever kind or nature, whether heretofore or hereafter accruing, for or because of any matter or things done, omitted or suffered to be done by any of the Released Parties prior to and including the Effective Date, and in any way directly or indirectly arising out of or in any way connected to this Agreement or any of the Loan Documents, or any of the transactions associated therewith, or the Mortgaged Property, including specifically but not limited to claims of usury.
- 17. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts with the same effect as if all parties hereto had signed the same document. All such counterparts shall be construed together and shall constitute one instrument, but in making proof hereof it shall only be necessary to produce one such counterpart.
- 18. Severability. If any covenant, condition, or provision herein contained is held to be invalid by final judgment of any court of competent jurisdiction, the invalidity of such covenant, condition, or provision shall not in any way affect any other covenant, condition or provision herein contained.
 - 19. <u>Time of the Essence</u>. It is expressly agreed by the parties hereto that time is of the essence with respect to this Agreement.
- 20. Representation by Counsel. The parties acknowledge and confirm that each of their respective attorneys have participated jointly in the review and revision of this Agreement and that it has not been written solely by counsel for one party. The parties hereto therefore

stipulate and agree that the rule of construction to the effect that any ambiguities are to or may be resolved against the drafting party shall not be employed in the interpretation of this Agreement to favor either party against the other.

- 21. Governing Law. This Agreement and the rights and duties of the parties hereunder shall be governed for all purposes by the law of the State of Texas and the law of the United States applicable to transactions within said State.
- 22. <u>Successors and Assigns.</u> The terms and provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns.
 - 23. Notice of No Oral Agreements. Borrower and Lender hereby take notice of and agree to the following:
- A. PURSUANT TO SUBSECTION 26.02(b) OF THE TEXAS BUSINESS AND COMMERCE CODE, A LOAN AGREEMENT IN WHICH THE AMOUNT INVOLVED THEREIN EXCEEDS \$50,000 IN VALUE IS NOT ENFORCEABLE UNLESS THE AGREEMENT IS IN WRITING AND SIGNED BY THE PARTY TO BE BOUND OR BY THAT PARTY'S AUTHORIZED REPRESENTATIVE.
- B. PURSUANT TO SUBSECTION 26.02(c) OF THE TEXAS BUSINESS AND COMMERCE CODE, THE RIGHTS AND OBLIGATIONS OF THE PARTIES TO THE LOAN DOCUMENTS SHALL BE DETERMINED SOLELY FROM THE LOAN DOCUMENTS, AND ANY PRIOR ORAL AGREEMENTS BETWEEN THE PARTIES ARE SUPERSEDED BY AND MERGED INTO THE LOAN DOCUMENTS.
- C. THE LOAN DOCUMENTS AND THIS AGREEMENT REPRESENT THE FINAL AGREEMENT BETWEEN THE PARTIES THERETO AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES THERETO. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, this Agreement is executed on the respective dates of acknowledgement below but is effective as of the date first above written.

BORROWER:

STRATUS PROPERTIES INC.,

a Delaware corporation

By: /s/ Erin D. Pickens

Erin D. Pickens, Sr. Vice President

STRATUS PROPERTIES OPERATING CO., L.P., a Delaware limited partnership

By: STRS L.L.C., a Delaware limited liability company, General Partner

By Stratus Properties Inc., a Delaware corporation, Sole Member

By: /s/ Erin D. Pickens
Erin D. Pickens,
Sr. Vice President

CIRCLE C LAND, L.P.,

a Texas limited partnership

By: Circle C GP, L.L.C., a Delaware limited liability company, General Partner

By Stratus Properties Inc., a Delaware corporation, Sole Member

By: /s/ Erin D. Pickens
Erin D. Pickens,
Sr. Vice President

AUSTIN 290 PROPERTIES, INC.,

a Texas corporation

By: /s/ Erin D. Pickens Erin D. Pickens, Sr. Vice President

[Signature Page - Fourth Modification and Extension Agreement]

COMERICA BANK

By: _/s/ Sterling J. Silver	Sterling J. Silver, Senior Vice President
	[Signature Page - Fourth Modification and Extension Agreement]

STATE OF TEXAS	§ §	
COUNTY OF TRAVIS	§	
This instrument was Delaware corporation, on beh	acknowledged be alf of said corpor	efore me on the day of March, 2010, by Erin D. Pickens, Sr. Vice President of Stratus Properties Inc., a ration.
		Notary Public, State of Texas My Commission Expires: Printed Name of Notary:
STATE OF TEXAS	§	
COUNTY OF TRAVIS	§ §	
Delaware corporation, Sole M	ember of STRS I	efore me on the day of March, 2010, by Erin D. Pickens, Sr. Vice President of Stratus Properties Inc., a L.L.C., a Delaware limited liability company, General Partner of Stratus Properties Operating Co., L.P., a I corporation, limited liability company and limited partnership. Notary Public, State of Texas My Commission Expires:
		Printed Name of Notary:
STATE OF TEXAS	§ §	
COUNTY OF TRAVIS	8 §	
Delaware corporation, Sole M	ember of Circle	efore me on the day of March, 2010, by Erin D. Pickens, Sr. Vice President of Stratus Properties Inc., a C GP, L.L.C., a Delaware limited liability company, General Partner of Circle C Land, L.P., a Delaware limited ited liability company limited partnership.
		Notary Public, State of Texas My Commission Expires: Printed Name of Notary:
		[Signature Page - Fourth Modification and Extension Agreement]

STATE OF TEXAS	§ §	
COUNTY OF TRAVIS	§ §	
This instrument was acknobelaware corporation, on behalf of		ne on the day of March, 2010, by Erin D. Pickens, Sr. Vice President of Austin 290 Properties Inc., a
		Notary Public, State of Texas My Commission Expires: Printed Name of Notary:
STATE OF TEXAS COUNTY OF DALLAS	\$ \$ \$	
This instrument was acknobehalf of said bank.	owledged before r	ne on the day of March, 2010, by Sterling J. Silver, Senior Vice President of Comerica Bank, on
		Notary Public, State of Texas My Commission Expires:
		Printed Name of Notary:
	[Signa	ture Page - Fourth Modification and Extension Agreement]

CONSENT OF CALERA COURT

The undersigned, Calera Court, L.P., hereby executes the Agreement to evidence its agreement to be bound by the extension of the Maturity Date, and confirms and agrees that all of its obligations under the Loan Documents applicable to it remain in full force and effect.

Executed on the date of acknowledgement below but effective as of March _____, 2010.

CALERA COURT, L.P. a Texas limited partnership

By:	Calera	Court M	anageme	ent, L.L.C., a Texas limited liability company, its general partner
	By:	Stratus	Properti	es Operating Co., L.P., a Delaware limited partnership, its Manager
		By:	STRS	L.L.C., a Delaware limited liability company, its general partner
			By:	Stratus Properties Inc., a Delaware corporation, its Sole Member
]	By: <u>/s/</u>	<u>Erin D. F</u>	<u>Pickens</u> Erin D. Pickens, Sr. Vice President

[Signature Page - Fourth Modification and Extension Agreement]

COUNTY OF TRAVIS §		
Delaware corporation, Sole Member of STRS L	efore me on the day of March, 2010, by Erin D. Pickens, Sr. V. L.C., a Delaware limited liability company, General Partner of St ra Court Management, L.L.C., a Texas limited liability company, ty and said limited partnership.	ratus Properties Operating Co., L.P., a
	Notary Public in and for the State of Texas	
My Commission Expires:	Printed/Typed Name of Notary	
	Signature Page - Fourth Modification and Extension Agreement]	

STATE OF TEXAS

§

CONSENT OF TRACT 107, L.L.C.

The undersigned, Tract 107, L.L.C., a Texas limited liability company, hereby executes the Agreement to evidence its agreement to be bound by the extension of the Maturity Date, and confirms and agrees that all of its obligations under the Loan Documents applicable to it remain in full force and effect.

Executed on the date of acknowledgement below but effective as of March _____, 2010.

	FRACT 107, L.L.C., Texas limited liability company
By:	/s/ Erin D. Pickens

2007 ASP \$3.5M Loan

LOAN MODIFICATION AGREEMENT

THIS LOAN MODIFICATION AGREEMENT (this "<u>Agreement</u>") is made as of March___, 2010, by and between STRATUS PROPERTIES INC., a Delaware corporation having an address of 98 San Jacinto Boulevard, Suite 220, Austin, TX 78791 ("<u>Borrower</u>"), and AMERICAN STRATEGIC INCOME PORTFOLIO INC., a Minnesota corporation having an address at c/o FAF Advisors, Inc., 800 Nicollet Mall, Suite 500, BC-MN-H05W, Minneapolis, MN 55402 ("<u>Investor</u>").

RECITALS:

WHEREAS, Investor holds the lender's interest in that certain "Loan" and other "Loan Documents" as defined and described in that certain Loan Agreement dated as of June 1, 2007 between Borrower and Holliday Fenoglio Fowler, L.P., a Texas limited partnership ("Lender") (the "Loan Agreement").

WHEREAS, Borrower is liable for the payment and performance of all of Borrower's obligations under the "Note" (as defined in the Loan Agreement) in the original principal amount of Three Million Five Hundred Thousand Dollars (\$3,500,000) and the other Loan Documents.

WHEREAS, Borrower and Investor desire to modify certain terms contained in the Note and Loan Agreement, and to reaffirm the Loan, as modified by this Agreement.

AGREEMENT:

In consideration of the foregoing premises and the mutual covenants set forth in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Investor and Borrower agree as follows.

ARTICLE I

1.1 <u>Definitions.</u> Capitalized terms not otherwise defined herein have the meanings ascribed to them in the Loan Agreement.

ARTICLE II

MODIFICATION OF THE LOAN AGREEMENT

- 2 . 1 <u>Financial Reporting of Total Stockholder Equity.</u> Section 6.A. (2) is hereby revised to add a reporting requirement with respect to Total Shareholder Equity; accordingly said section is amended and restated in its entirety to read as follows:
 - On or before the first Business Day which occurs forty-five (45) calendar days after the close of each Accounting Period, (i) Borrower's balance sheet as of the close of such Accounting Period and its income statement for that portion of the then current fiscal year through the end of such Accounting Period prepared in accordance with GAAP and certified as being complete, correct, and fairly representing its financial condition and results of operations by the chief financial officer of Borrower, subject to the absence of footnotes and year-end adjustments, (ii) a statement of changes in equity and cash flows for the period ended on such date, certified by the chief financial officer of Borrower, (iii) the calculation of the Debt Service Coverage Ratio and the calculation of Total Stockholder Equity demonstrating that Borrower is in compliance with Subsection 8.G. of this Agreement, together with any supporting calculations used to arrive at such calculations, certified by the chief financial officer of Borrower, and (iv) a completed Borrower's Officer's Compliance Certificate:
- 2 . 2 <u>Redemption.</u> Section 7.F. (2) is hereby revised to delete the limited right of Borrower to redeem certain shares of common stock of Borrower; accordingly said section is amended and restated in its entirety to read as follows:
 - (2) Borrower will not, except as allowed below, directly or indirectly redeem, retire, purchase, or otherwise acquire beneficially any shares of any class of its own stock now or hereafter outstanding or set apart any sum for any such purpose.
- 2.3 <u>Senior Management.</u> Section 7.H. (a)(iii) is hereby revised to replace John E. Baker with Erin D. Pickens as one of the three individuals currently comprising senior management; accordingly said section is amended and restated in its entirety to read as follows:
 - (iii) a majority of those three individuals currently comprising senior management, William H. Armstrong, President, Erin D. Pickens, Senior Vice President, and Kenneth N. Jones, General Counsel, cease to serve in their current positions; or
- 2.4 <u>Debt Service Default Exception</u>. Section 8.G. is hereby revised to add an exception to the Debt Service Coverage Ration default provision, upon the Borrower's satisfaction

of certain other financial conditions; accordingly said section is amended and restated in its entirety to read as follows

G. The Debt Service Coverage Ratio measured on a quarterly basis for the previous twelve (12) months shall be less than (1) (a) 5.0 minus (b) the product of 5.0 multiplied by the Cash Collateral Factor, to (2) 1.0, unless Borrower has Total Stockholder Equity equal to or greater than One Hundred Twenty Million Dollars (\$120,000,000).

2.5 New Definitions. Schedule 1 to Loan Agreement is hereby revised to add the following new definitions:

"Beal Guaranty" means that certain Amended and Restated Guaranty Agreement dated as of October 21, 2009 executed by Borrower and Canyon-Johnson Urban Fund II, L.P., as guarantors, for the benefit of Beal Bank Nevada in connection with a \$120,000,000.00 construction financing facility from Beal Bank Nevada to CJUF II Stratus Block 21, LLC evidenced by an Amended and Restated Construction Loan Agreement of even date therewith.

"Ford Guaranty" means that certain Guaranty Agreement dated as of the date hereof executed by Borrower, as guarantor, for the benefit of Hunter's Glen/Ford Investments 1 LLC ("Ford Lender") in connection with a Thirty Million Dollar (\$30,000,000) mezzanine construction financing facility from Ford Lender to CJUF II Stratus Block 21, LLC evidenced by a Loan Agreement of even date therewith.

"Total Stockholder Equity" means the amount shown on the line item entry set forth in Borrower's balance sheet (to be submitted to Lender in accordance with Section 6.A.(2) of the Agreement) for "Total Stratus stockholders' equity" (for illustrative purposes only, Total Stockholder Equity as of September 30, 2009 is listed on the Borrower's balance sheet as Total Stratus stockholders' equity in the amount of \$144,598,000); provided, if in the future Borrower deconsolidates financial statements of joint ventures from Borrower's financial statements, then the reference to "Total Stratus stockholders' equity" above will be deemed to refer to "Total stockholders' equity." Total Stockholder Equity shall be determined in accordance with GAAP, as in effect as of the date hereof. In the event the applicable accounting standard or any other aspect of GAAP are materially revised so as to alter the determination of Total Stockholder Equity, Total Stockholder Equity shall nevertheless be calculated in accordance with GAAP in effect as of the date hereof (so that the alternative minimum Total Stockholder Equity requirement set forth in Section 8.G. remains a fixed benchmark) and Borrower shall submit and certify the appropriate calculations separate and apart from the balance sheet it submits to Lender in accordance with Section 6.A.(2).

2.6 **Revised Definitions.** Schedule 1 to Loan Agreement is hereby revised to amend certain definitions; accordingly the following definitions are amended and restated in their entirety to read as follows:

"Comerica Loan Agreement" means that certain Loan Agreement dated as of September 30, 2005, among Borrower and certain Affiliates of Borrower and Comerica Bank-Texas- as modified by Modification and Extension Agreement dated as of May 30, 2006, Second Modification and Extension Agreement dated as of May 30, 2008, and Fourth Modification and Extension Agreement dated as of the date hereof.

"Maturity Date" means the Maturity Date set forth in the Note.

"Permitted Debt" means (i) the Loan and other Indebtedness to Lender or Related Lenders, (ii) the Comerica Debt (as of the date hereof), (iii) any other Indebtedness of Borrower for fair value received that is secured by assets owned by Borrower having an appraised value equal to or greater than the indebtedness secured thereby (and which assets do not secure other indebtedness), (iv) debt outstanding as of the date of the Loan Agreement, (v) unsecured trade, utility or non-extraordinary accounts payable in the ordinary course of business and other unsecured debt of Borrower at any one time not to exceed Five Hundred Thousand Dollars (\$500,000), and (vi) the Beal Guaranty (as of the date hereof), the Ford Guaranty (as of the date hereof), and guaranties of Borrower guaranteeing project development and/or construction costs and related costs, provided that Borrower has a direct or indirect interest in such projects and that the aggregate amount, at any one time, of such guaranties does not exceed the sum of Fifteen Million Dollars (\$15,000,000).

ARTICLE III

MODIFICATION OF THE NOTE

- 3.1 Extended Maturity Date. The Maturity Date (as defined in Section 5.(a) of the Note) shall be extended; accordingly the stated Maturity Date in the Note of December 31, 2011 is hereby amended to so that the Maturity Date for the Note shall be December 31, 2014.
- 3.2 <u>Interest Rate</u>. The Regular Rate (as defined in **Section 2 of the Note**) shall be increased; accordingly the stated Regular Rate in the Note of 6.915% is hereby amended to so that the Regular Rate of interest shall be 8.75%.
- 3.3 <u>Prepayment Adjustments.</u> Sections 8.(a) and (b) of the Note is hereby revised to adjust the applicable time frames and corresponding amounts of any Reinvestment Charge due under the terms of the Loan Documents; accordingly said sections are amended and restated in their entirety to read as follows:

- (a) Prior to December 31, 2011, this Note may not be prepaid, except as may be expressly provided for in the Loan Agreement.
- (b) Beginning on January 1, 2012, the Borrower may prepay this Note in whole or in part, subject to subsection 8(c) below, provided such prepayment is accompanied by a reinvestment charge (hereinafter referred to as the "Reinvestment Charge"). The Reinvestment Charge shall be an amount equal to (i) two percent (2.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2012 through December 31, 2012, (ii) one percent (1.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2013 through December 31, 2013, and (iii) zero percent (0%) of the principal amount prepaid with respect to prepayments made after December 31, 2013.

ARTICLE IV

MISCELLANEOUS

- 4.1 No Other Amendments Intended. Except as specifically provided herein, no other amendment of the Loan Documents is intended and all other terms and conditions of the Note, the Loan Agreement and any other Loan Documents shall remain in full force and effect and shall not be modified or released in any way by this Agreement. This Agreement amends the Note and is not in payment or substitution thereof. Borrower hereby ratifies and reaffirms all of Borrower's obligations under the Note, the Loan Agreement and all of the other Loan Documents as amended hereby.
- 4.2 <u>No Impairment of Lien.</u> Nothing in this Agreement shall affect the lien of any of the Loan Documents or the priority of any such liens, nor release or change the liability of any party who may now be or after the date of this Agreement, become liable, primarily or secondarily, under the Loan Documents.
 - 4.3 Representations and Warranties of Borrower.
- (a) <u>Representations and Warranties in Loan Documents</u>. The representations and warranties of Borrower contained in the other Loan Documents, as amended hereby, are true and correct in all material respects as of the date first written above (as if such representations and warranties were made effective as of the date first written above).
- (b) <u>Power to Perform.</u> Borrower has the power, under its organizational documents, to enter into this Agreement and to perform the obligations required to be performed by Borrower under the terms hereunder.
- (c) <u>Due Authorization</u>. The execution, delivery and performance by Borrower of this Agreement have been duly authorized by all necessary action on the part of Borrower. This Agreement has been duly executed and delivered by Borrower and, assuming the due execution

and delivery of this Agreement by Investor, constitute the legal, valid and binding obligations of Borrower enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles affecting the enforcement of creditors' rights generally. No registration with, or consent or approval of, or notice to, or other action by, (i) any trustee or holder of any indebtedness or obligation of Borrower or (ii) any other person for the execution, delivery and performance of this Agreement by Borrower is required or, if required, such registration has been made, such consent, approval or notice given or such other appropriate action taken.

4.4 Miscellaneous.

- (a) **Jurisdiction.** This Agreement shall be construed according to and governed by the laws of the state of Minnesota.
- (b) <u>Severability; Counterparts.</u> If any provision of this Agreement is adjudicated to be invalid, illegal or enforceable, in whole or in part, it will be deemed omitted to that extent and all other provisions of this Agreement will remain in full force and effect. This Agreement may be executed in one or more counterparts and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed to be an original; such counterparts, together, shall constitute one and the same agreement.
- (c) <u>Notices.</u> All notices given pursuant to this Agreement must be in writing and will be effectively given if personally delivered or, if mailed, postage prepaid, certified or registered mail, return receipt requested, to the addresses of Investor and Borrower first set forth above or to such other address as any party subsequently may designate in writing.
- (d) Assignment. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, legal representatives, successors and assigns. This Agreement may not be assigned by Borrower without the prior written consent of Investor. This Agreement may be assigned by Investor in connection with an assignment of the Loan without any required notice to Borrower.
- (e) <u>Costs; Fee; Further Assurances</u>. Borrower agrees to pay Investor's out-of-pocket expenses in connection with the preparation of this Agreement and any related expenses, including without limitation, reasonable attorneys' fees. In addition, Borrower agrees to execute such other instruments as may be reasonably required by Investor to evidence or facilitate the agreements set forth herein.

[The remainder of this page was intentionally left blank.]

LOAN MODIFICATION AGREEMENT

BORROWER SIGNATURE PAGE

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be execut	ed as of the day and year first above written.
	BORROWER:
	STRATUS PROPERTIES INC., a Delaware corporation
	B y :/s/ Erin I Pickens Name: Erin D. Pickens Title: Senior Vice President

7

LOAN MODIFICATION AGREEMENT

INVESTOR SIGNATURE PAGE

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above written.	
	INVESTOR:
	AMERICAN STRATEGIC INCOME PORTFOLIO INC., a Minnesota corporation
	B y : Name: Its:
8	

2006 BSP \$8M Loan

LOAN MODIFICATION AGREEMENT

THIS LOAN MODIFICATION AGREEMENT (this "<u>Agreement</u>") is made as of March___, 2010, by and between STRATUS PROPERTIES INC., a Delaware corporation having an address of 98 San Jacinto Boulevard, Suite 220, Austin, TX 78791 ("<u>Borrower</u>"), and AMERICAN STRATEGIC INCOME PORTFOLIO INC.—II, a Minnesota corporation having an address at c/o FAF Advisors, Inc., 800 Nicollet Mall, Suite 500, BC-MN-H05W, Minneapolis, MN 55402 ("<u>Investor</u>").

RECITALS:

WHEREAS, Investor holds the lender's interest in that certain "Loan" and other "Loan Documents" as defined and described in that certain Loan Agreement dated as of December 12, 2006 between Borrower and Holliday Fenoglio Fowler, L.P., a Texas limited partnership ("Lender") (the "Loan Agreement").

WHEREAS, Borrower is liable for the payment and performance of all of Borrower's obligations under the "Note" (as defined in the Loan Agreement) in the original principal amount of Eight Million Dollars (\$8,000,000) and the other Loan Documents.

WHEREAS, Borrower and Investor desire to modify certain terms contained in the Note and Loan Agreement, and to reaffirm the Loan, as modified by this Agreement.

AGREEMENT:

In consideration of the foregoing premises and the mutual covenants set forth in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Investor and Borrower agree as follows.

ARTICLE I

1.1 <u>Definitions</u>. Capitalized terms not otherwise defined herein have the meanings ascribed to them in the Loan Agreement.

ARTICLE II

MODIFICATION OF THE LOAN AGREEMENT

- 2 . 1 <u>Financial Reporting of Total Stockholder Equity.</u> Section 6.A. (2) is hereby revised to add a reporting requirement with respect to Total Shareholder Equity; accordingly said section is amended and restated in its entirety to read as follows:
 - On or before the first Business Day which occurs forty-five (45) calendar days after the close of each Accounting Period, (i) Borrower's balance sheet as of the close of such Accounting Period and its income statement for that portion of the then current fiscal year through the end of such Accounting Period prepared in accordance with GAAP and certified as being complete, correct, and fairly representing its financial condition and results of operations by the chief financial officer of Borrower, subject to the absence of footnotes and year-end adjustments, (ii) a statement of changes in equity and cash flows for the period ended on such date, certified by the chief financial officer of Borrower, (iii) the calculation of the Debt Service Coverage Ratio and the calculation of Total Stockholder Equity demonstrating that Borrower is in compliance with Subsection 8.G. of this Agreement, together with any supporting calculations used to arrive at such calculations, certified by the chief financial officer of Borrower, and (iv) a completed Borrower's Officer's Compliance Certificate:
- 2 . 2 <u>Redemption.</u> Section 7.F. (2) is hereby revised to delete the limited right of Borrower to redeem certain shares of common stock of Borrower; accordingly said section is amended and restated in its entirety to read as follows:
 - (2) Borrower will not, except as allowed below, directly or indirectly redeem, retire, purchase, or otherwise acquire beneficially any shares of any class of its own stock now or hereafter outstanding or set apart any sum for any such purpose.
- 2.3 <u>Senior Management.</u> Section 7.H. (a)(iii) is hereby revised to replace John E. Baker with Erin D. Pickens as one of the three individuals currently comprising senior management; accordingly said section is amended and restated in its entirety to read as follows:
 - (iii) a majority of those three individuals currently comprising senior management, William H. Armstrong, President, <u>Erin D. Pickens</u>, Senior Vice President, and Kenneth N. Jones, General Counsel, cease to serve in their current positions; or

- 2.4 <u>Debt Service Default Exception</u>. Section 8.G. is hereby revised to add an exception to the Debt Service Coverage Ration default provision, upon the Borrower's satisfaction of certain other financial conditions; accordingly said section is amended and restated in its entirety to read as follows:
 - G. The Debt Service Coverage Ratio measured on a quarterly basis for the previous twelve (12) months shall be less than (1) (a) 5.0 minus (b) the product of 5.0 multiplied by the Cash Collateral Factor, to (2) 1.0, unless Borrower has Total Stockholder Equity equal to or greater than One Hundred Twenty Million Dollars (\$120,000,000).
- 2.5 New Definitions. Schedule 1 to Loan Agreement is hereby revised to add the following new definitions:

"Beal Guaranty" means that certain Amended and Restated Guaranty Agreement dated as of October 21, 2009 executed by Borrower and Canyon-Johnson Urban Fund II, L.P., as guarantors, for the benefit of Beal Bank Nevada in connection with a \$120,000,000.00 construction financing facility from Beal Bank Nevada to CJUF II Stratus Block 21, LLC evidenced by an Amended and Restated Construction Loan Agreement of even date therewith.

"Ford Guaranty" means that certain Guaranty Agreement dated as of the date hereof executed by Borrower, as guarantor, for the benefit of Hunter's Glen/Ford Investments 1 LLC ("Ford Lender") in connection with a Thirty Million Dollar (\$30,000,000) mezzanine construction financing facility from Ford Lender to CJUF II Stratus Block 21, LLC evidenced by a Loan Agreement of even date therewith.

"Total Stockholder Equity" means the amount shown on the line item entry set forth in Borrower's balance sheet (to be submitted to Lender in accordance with Section 6.A.(2) of the Agreement) for "Total Stratus stockholders' equity" (for illustrative purposes only, Total Stockholder Equity as of September 30, 2009 is listed on the Borrower's balance sheet as Total Stratus stockholders' equity in the amount of \$144,598,000); provided, if in the future Borrower deconsolidates financial statements of joint ventures from Borrower's financial statements, then the reference to "Total Stratus stockholders' equity" above will be deemed to refer to "Total stockholders' equity." Total Stockholder Equity shall be determined in accordance with GAAP, as in effect as of the date hereof. In the event the applicable accounting standard or any other aspect of GAAP are materially revised so as to alter the determination of Total Stockholder Equity, Total Stockholder Equity shall nevertheless be calculated in accordance with GAAP in effect as of the date hereof (so that the alternative minimum Total Stockholder Equity requirement set forth in Section 8.G. remains a fixed benchmark) and Borrower shall

submit and certify the appropriate calculations separate and apart from the balance sheet it submits to Lender in accordance with Section 6.A.(2).

2 . 6 Revised Definitions. Schedule 1 to Loan Agreement is hereby revised to amend certain definitions; accordingly the following definitions are amended and restated in their entirety to read as follows:

"Comerica Loan Agreement" means that certain Loan Agreement dated as of September 30, 2005, among Borrower and certain Affiliates of Borrower and Comerica Bank-Texas- as modified by Modification and Extension Agreement dated as of May 30, 2006, Second Modification and Extension Agreement dated as of May 30, 2008, and Fourth Modification and Extension Agreement dated as of the date hereof.

"Maturity Date" means the Maturity Date set forth in the Note.

"Permitted Debt" means (i) the Loan and other Indebtedness to Lender or Related Lenders, (ii) the Comerica Debt (as of the date hereof), (iii) any other Indebtedness of Borrower for fair value received that is secured by assets owned by Borrower having an appraised value equal to or greater than the indebtedness secured thereby (and which assets do not secure other indebtedness), (iv) debt outstanding as of the date of the Loan Agreement, (v) unsecured trade, utility or non-extraordinary accounts payable in the ordinary course of business and other unsecured debt of Borrower at any one time not to exceed Five Hundred Thousand Dollars (\$500,000), and (vi) the Beal Guaranty (as of the date hereof), the Ford Guaranty (as of the date hereof), and guaranties of Borrower guaranteeing project development and/or construction costs and related costs, provided that Borrower has a direct or indirect interest in such projects and that the aggregate amount, at any one time, of such guaranties does not exceed the sum of Fifteen Million Dollars (\$15,000,000).

ARTICLE III

MODIFICATION OF THE NOTE

- 3.1 Extended Maturity Date. The Maturity Date (as defined in Section 5.(a) of the Note) shall be extended; accordingly the stated Maturity Date in the Note of December 31, 2011 is hereby amended to so that the Maturity Date for the Note shall be December 31, 2013.
- 3.2 <u>Interest Rate</u>. The Regular Rate (as defined in **Section 2 of the Note**) shall be increased; accordingly the stated Regular Rate in the Note of 6.560% is hereby amended to so that the Regular Rate of interest shall be 8.75%.

- 3 . 3 Prepayment Adjustments. Sections 8.(a) and (b) of the Note are hereby revised to adjust the applicable time frames and corresponding amounts of any Reinvestment Charge due under the terms of the Loan Documents; accordingly said sections are amended and restated in their entirety to read as follows:
 - (a) Prior to December 31, 2010, this Note may not be prepaid, except as may be expressly provided for in the Loan Agreement.
 - (b) Beginning on January 1, 2011, the Borrower may prepay this Note in whole or in part, subject to subsection 8(c) below, provided such prepayment is accompanied by a reinvestment charge (hereinafter referred to as the "Reinvestment Charge"). The Reinvestment Charge shall be an amount equal to (i) two percent (2.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2011 through December 31, 2011, (ii) one percent (1.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2012 through December 31, 2012, and (iii) zero percent (0%) of the principal amount prepaid with respect to prepayments made after December 31, 2012.

ARTICLE IV

MISCELLANEOUS

- 4.1 **No Other Amendments Intended.** Except as specifically provided herein, no other amendment of the Loan Documents is intended and all other terms and conditions of the Note, the Loan Agreement and any other Loan Documents shall remain in full force and effect and shall not be modified or released in any way by this Agreement. This Agreement amends the Note and is not in payment or substitution thereof. Borrower hereby ratifies and reaffirms all of Borrower's obligations under the Note, the Loan Agreement and all of the other Loan Documents as amended hereby.
- 4.2 <u>No Impairment of Lien.</u> Nothing in this Agreement shall affect the lien of any of the Loan Documents or the priority of any such liens, nor release or change the liability of any party who may now be or after the date of this Agreement, become liable, primarily or secondarily, under the Loan Documents.
 - 4.3 Representations and Warranties of Borrower.
- (a) Representations and Warranties in Loan Documents. The representations and warranties of Borrower contained in the other Loan Documents, as amended hereby, are true and correct in all material respects as of the date first written above (as if such representations and warranties were made effective as of the date first written above).

- (b) <u>Power to Perform.</u> Borrower has the power, under its organizational documents, to enter into this Agreement and to perform the obligations required to be performed by Borrower under the terms hereunder.
- Due Authorization. The execution, delivery and performance by Borrower of this Agreement have been duly authorized by all necessary action on the part of Borrower. This Agreement has been duly executed and delivered by Borrower and, assuming the due execution and delivery of this Agreement by Investor, constitute the legal, valid and binding obligations of Borrower enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles affecting the enforcement of creditors' rights generally. No registration with, or consent or approval of, or notice to, or other action by, (i) any trustee or holder of any indebtedness or obligation of Borrower or (ii) any other person for the execution, delivery and performance of this Agreement by Borrower is required or, if required, such registration has been made, such consent, approval or notice given or such other appropriate action taken.

4.4 Miscellaneous.

- (a) Jurisdiction. This Agreement shall be construed according to and governed by the laws of the state of Minnesota.
- (b) <u>Severability; Counterparts.</u> If any provision of this Agreement is adjudicated to be invalid, illegal or enforceable, in whole or in part, it will be deemed omitted to that extent and all other provisions of this Agreement will remain in full force and effect. This Agreement may be executed in one or more counterparts and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed to be an original; such counterparts, together, shall constitute one and the same agreement.
- (c) <u>Notices</u>. All notices given pursuant to this Agreement must be in writing and will be effectively given if personally delivered or, if mailed, postage prepaid, certified or registered mail, return receipt requested, to the addresses of Investor and Borrower first set forth above or to such other address as any party subsequently may designate in writing.
- (d) Assignment. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, legal representatives, successors and assigns. This Agreement may not be assigned by Borrower without the prior written consent of Investor. This Agreement may be assigned by Investor in connection with an assignment of the Loan without any required notice to Borrower.
- (e) <u>Costs; Fee; Further Assurances.</u> Borrower agrees to pay Investor's out-of-pocket expenses in connection with the preparation of this Agreement and any related expenses, including without limitation, reasonable attorneys' fees. In addition, Borrower agrees to execute

such other instruments as may be reasonably required by Investor to evidence or facilitate the agreements set forth herein.

[The remainder of this page was intentionally left blank.]

LOAN MODIFICATION AGREEMENT

BORROWER SIGNATURE PAGE

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the day and year first above written.

BORROWER:

STRATUS PROPERTIES INC.,

a Delaware corporation

By: /s/ Erin D. Pickens Name: Erin D. Pickens Title: Senior Vice President

LOAN MODIFICATION AGREEMENT

INVESTOR SIGNATURE PAGE

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above written.

INVESTOR:

AMERICAN STRATEGIC INCOME
PORTFOLIO INC.—II,
a Minnesota corporation

By: ______
Name: ______
Its: _______
Its: _______

2006 BSP \$5M Loan

LOAN MODIFICATION AGREEMENT

THIS LOAN MODIFICATION AGREEMENT (this "<u>Agreement</u>") is made as of March___, 2010, by and between STRATUS PROPERTIES INC., a Delaware corporation having an address of 98 San Jacinto Boulevard, Suite 220, Austin, TX 78791 ("<u>Borrower</u>"), and AMERICAN STRATEGIC INCOME PORTFOLIO INC.—II, a Minnesota corporation having an address at c/o FAF Advisors, Inc., 800 Nicollet Mall, Suite 500, BC-MN-H05W, Minneapolis, MN 55402 ("<u>Investor</u>").

RECITALS:

WHEREAS, Investor holds the lender's interest in that certain "Loan" and other "Loan Documents" as defined and described in that certain Amended and Restated Loan Agreement dated as of December 12, 2006 between Borrower and Holliday Fenoglio Fowler, L.P., a Texas limited partnership ("Lender") (the "Loan Agreement").

WHEREAS, Borrower is liable for the payment and performance of all of Borrower's obligations under the "Note" (as defined in the Loan Agreement) in the original principal amount of Five Million Dollars (\$5,000,000) and the other Loan Documents.

WHEREAS, Borrower and Investor desire to modify certain terms contained in the Note and Loan Agreement, and to reaffirm the Loan, as modified by this Agreement.

AGREEMENT:

In consideration of the foregoing premises and the mutual covenants set forth in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Investor and Borrower agree as follows.

ARTICLE I

1.1 <u>Definitions</u>. Capitalized terms not otherwise defined herein have the meanings ascribed to them in the Loan Agreement.

ARTICLE II

MODIFICATION OF THE LOAN AGREEMENT

- 2 . 1 <u>Financial Reporting of Total Stockholder Equity.</u> Section 6.A. (2) is hereby revised to add a reporting requirement with respect to Total Shareholder Equity; accordingly said section is amended and restated in its entirety to read as follows:
 - On or before the first Business Day which occurs forty-five (45) calendar days after the close of each Accounting Period, (i) Borrower's balance sheet as of the close of such Accounting Period and its income statement for that portion of the then current fiscal year through the end of such Accounting Period prepared in accordance with GAAP and certified as being complete, correct, and fairly representing its financial condition and results of operations by the chief financial officer of Borrower, subject to the absence of footnotes and year-end adjustments, (ii) a statement of changes in equity and cash flows for the period ended on such date, certified by the chief financial officer of Borrower, (iii) the calculation of the Debt Service Coverage Ratio and the calculation of Total Stockholder Equity demonstrating that Borrower is in compliance with Subsection 8.G. of this Agreement, together with any supporting calculations used to arrive at such calculations, certified by the chief financial officer of Borrower, and (iv) a completed Borrower's Officer's Compliance Certificate:
- 2 . 2 <u>Redemption.</u> Section 7.F. (2) is hereby revised to delete the limited right of Borrower to redeem certain shares of common stock of Borrower; accordingly said section is amended and restated in its entirety to read as follows:
 - (2) Borrower will not, except as allowed below, directly or indirectly redeem, retire, purchase, or otherwise acquire beneficially any shares of any class of its own stock now or hereafter outstanding or set apart any sum for any such purpose.
- 2.3 <u>Senior Management.</u> Section 7.H. (a)(iii) is hereby revised to replace John E. Baker with Erin D. Pickens as one of the three individuals currently comprising senior management; accordingly said section is amended and restated in its entirety to read as follows:
 - (iii) a majority of those three individuals currently comprising senior management, William H. Armstrong, President, Erin D. Pickens, Senior Vice President, and Kenneth N. Jones, General Counsel, cease to serve in their current positions; or

- 2.4 <u>Debt Service Default Exception</u>. Section 8.G. is hereby revised to add an exception to the Debt Service Coverage Ration default provision, upon the Borrower's satisfaction of certain other financial conditions; accordingly said section is amended and restated in its entirety to read as follows:
 - G. The Debt Service Coverage Ratio measured on a quarterly basis for the previous twelve (12) months shall be less than (1) (a) 5.0 minus (b) the product of 5.0 multiplied by the Cash Collateral Factor, to (2) 1.0, unless Borrower has Total Stockholder Equity equal to or greater than One Hundred Twenty Million Dollars (\$120,000,000).
- 2.5 New Definitions. Schedule 1 to Loan Agreement is hereby revised to add the following new definitions:

"Beal Guaranty" means that certain Amended and Restated Guaranty Agreement dated as of October 21, 2009 executed by Borrower and Canyon-Johnson Urban Fund II, L.P., as guarantors, for the benefit of Beal Bank Nevada in connection with a \$120,000,000.00 construction financing facility from Beal Bank Nevada to CJUF II Stratus Block 21, LLC evidenced by an Amended and Restated Construction Loan Agreement of even date therewith.

"Ford Guaranty" means that certain Guaranty Agreement dated as of the date hereof executed by Borrower, as guarantor, for the benefit of Hunter's Glen/Ford Investments 1 LLC ("Ford Lender") in connection with a Thirty Million Dollar (\$30,000,000) mezzanine construction financing facility from Ford Lender to CJUF II Stratus Block 21, LLC evidenced by a Loan Agreement of even date therewith.

"Total Stockholder Equity" means the amount shown on the line item entry set forth in Borrower's balance sheet (to be submitted to Lender in accordance with Section 6.A.(2) of the Agreement) for "Total Stratus stockholders' equity" (for illustrative purposes only, Total Stockholder Equity as of September 30, 2009 is listed on the Borrower's balance sheet as Total Stratus stockholders' equity in the amount of \$144,598,000); provided, if in the future Borrower deconsolidates financial statements of joint ventures from Borrower's financial statements, then the reference to "Total Stratus stockholders' equity" above will be deemed to refer to "Total stockholders' equity." Total Stockholder Equity shall be determined in accordance with GAAP, as in effect as of the date hereof. In the event the applicable accounting standard or any other aspect of GAAP are materially revised so as to alter the determination of Total Stockholder Equity, Total Stockholder Equity shall nevertheless be calculated in accordance with GAAP in effect as of the date hereof (so that the alternative minimum Total Stockholder Equity requirement set forth in Section 8.G. remains a fixed benchmark) and Borrower shall submit and certify the appropriate calculations separate and apart from the balance sheet it submits to Lender in accordance with Section 6.A.(2).

2 . 6 Revised Definitions. Schedule 1 to Loan Agreement is hereby revised to amend certain definitions; accordingly the following definitions are amended and restated in their entirety to read as follows:

"Comerica Loan Agreement" means that certain Loan Agreement dated as of September 30, 2005, among Borrower and certain Affiliates of Borrower and Comerica Bank-Texas- as modified by Modification and Extension Agreement dated as of May 30, 2006, Second Modification and Extension Agreement dated as of May 30, 2008, and Fourth Modification and Extension Agreement dated as of the date hereof.

"Maturity Date" means the Maturity Date set forth in the Note.

"Permitted Debt" means (i) the Loan and other Indebtedness to Lender or Related Lenders, (ii) the Comerica Debt (as of the date hereof), (iii) any other Indebtedness of Borrower for fair value received that is secured by assets owned by Borrower having an appraised value equal to or greater than the indebtedness secured thereby (and which assets do not secure other indebtedness), (iv) debt outstanding as of the date of the Loan Agreement, (v) unsecured trade, utility or non-extraordinary accounts payable in the ordinary course of business and other unsecured debt of Borrower at any one time not to exceed Five Hundred Thousand Dollars (\$500,000), and (vi) the Beal Guaranty (as of the date hereof), the Ford Guaranty (as of the date hereof), and guaranties of Borrower guaranteeing project development and/or construction costs and related costs, provided that Borrower has a direct or indirect interest in such projects and that the aggregate amount, at any one time, of such guaranties does not exceed the sum of Fifteen Million Dollars (\$15,000,000).

ARTICLE III

MODIFICATION OF THE NOTE

- 3.1 <u>Interest Rate</u>. The Regular Rate (as defined in **Section 2 of the Note**) shall be increased; accordingly the stated Regular Rate in the Note of 6.560% is hereby amended to so that the Regular Rate of interest shall be 8.75%.
- 3.2 <u>Prepayment Adjustments.</u> Sections 8.(a) and (b) of the Note are hereby revised to allow for prepayment of the Note, which would not be subject to any Reinvestment Charge (as defined in the original Loan Agreement); accordingly said sections are amended and restated in their entirety to read as follows:

- (a) Borrower may prepay this Note in whole or in part at any time, subject to subsection 8(c) below.
- (b) Any prepayment so made shall not be subject to any reinvestment charge or other prepayment premium.

ARTICLE IV

MISCELLANEOUS

- 4.1 No Other Amendments Intended. Except as specifically provided herein, no other amendment of the Loan Documents is intended and all other terms and conditions of the Note, the Loan Agreement and any other Loan Documents shall remain in full force and effect and shall not be modified or released in any way by this Agreement. This Agreement amends the Note and is not in payment or substitution thereof. Borrower hereby ratifies and reaffirms all of Borrower's obligations under the Note, the Loan Agreement and all of the other Loan Documents as amended hereby.
- 4.2 <u>No Impairment of Lien.</u> Nothing in this Agreement shall affect the lien of any of the Loan Documents or the priority of any such liens, nor release or change the liability of any party who may now be or after the date of this Agreement, become liable, primarily or secondarily, under the Loan Documents.
 - 4.3 Representations and Warranties of Borrower.
- (a) <u>Representations and Warranties in Loan Documents</u>. The representations and warranties of Borrower contained in the other Loan Documents, as amended hereby, are true and correct in all material respects as of the date first written above (as if such representations and warranties were made effective as of the date first written above).
- (b) <u>Power to Perform.</u> Borrower has the power, under its organizational documents, to enter into this Agreement and to perform the obligations required to be performed by Borrower under the terms hereunder.
- (c) <u>Due Authorization</u>. The execution, delivery and performance by Borrower of this Agreement have been duly authorized by all necessary action on the part of Borrower. This Agreement has been duly executed and delivered by Borrower and, assuming the due execution and delivery of this Agreement by Investor, constitute the legal, valid and binding obligations of Borrower enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles affecting the enforcement of creditors' rights generally. No registration with, or consent or approval of, or notice to, or other action by, (i) any trustee or holder of any

indebtedness or obligation of Borrower or (ii) any other person for the execution, delivery and performance of this Agreement by Borrower is required or, if required, such registration has been made, such consent, approval or notice given or such other appropriate action taken.

4.4 Miscellaneous.

- (a) <u>Jurisdiction</u>. This Agreement shall be construed according to and governed by the laws of the state of Minnesota.
- (b) <u>Severability; Counterparts.</u> If any provision of this Agreement is adjudicated to be invalid, illegal or enforceable, in whole or in part, it will be deemed omitted to that extent and all other provisions of this Agreement will remain in full force and effect. This Agreement may be executed in one or more counterparts and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed to be an original; such counterparts, together, shall constitute one and the same agreement.
- (c) <u>Notices.</u> All notices given pursuant to this Agreement must be in writing and will be effectively given if personally delivered or, if mailed, postage prepaid, certified or registered mail, return receipt requested, to the addresses of Investor and Borrower first set forth above or to such other address as any party subsequently may designate in writing.
- (d) <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, legal representatives, successors and assigns. This Agreement may not be assigned by Borrower without the prior written consent of Investor. This Agreement may be assigned by Investor in connection with an assignment of the Loan without any required notice to Borrower.
- (e) <u>Costs; Fee; Further Assurances</u>. Borrower agrees to pay Investor's out-of-pocket expenses in connection with the preparation of this Agreement and any related expenses, including without limitation, reasonable attorneys' fees. In addition, Borrower agrees to execute such other instruments as may be reasonably required by Investor to evidence or facilitate the agreements set forth herein.

[The remainder of this page was intentionally left blank.]

LOAN MODIFICATION AGREEMENT

BORROWER SIGNATURE PAGE

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the day and year first above written.

BORROWER:

STRATUS PROPERTIES INC.,

a Delaware corporation
By: /s/ Erin D. Pickens
Name: Erin D. Pickens
Title: Senior Vice President

LOAN MODIFICATION AGREEMENT

INVESTOR SIGNATURE PAGE

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above written.

INVESTOR:
AMERICAN STRATEGIC INCOME PORTFOLIO INC.—II,
a Minnesota corporation

By: ______
Name: _____
Its: _____

2007 CSP \$8M Loan

LOAN MODIFICATION AGREEMENT

THIS LOAN MODIFICATION AGREEMENT (this "<u>Agreement</u>") is made as of March___, 2010, by and between STRATUS PROPERTIES INC., a Delaware corporation having an address of 98 San Jacinto Boulevard, Suite 220, Austin, TX 78791 ("<u>Borrower</u>"), and AMERICAN STRATEGIC INCOME PORTFOLIO INC.—III, a Minnesota corporation having an address at c/o FAF Advisors, Inc., 800 Nicollet Mall, Suite 500, BC-MN-H05W, Minneapolis, MN 55402 ("<u>Investor</u>").

RECITALS:

WHEREAS, Investor holds the lender's interest in that certain "Loan" and other "Loan Documents" as defined and described in that certain Loan Agreement dated as of June 1, 2007 between Borrower and Holliday Fenoglio Fowler, L.P., a Texas limited partnership ("Lender") (the "Loan Agreement").

WHEREAS, Borrower is liable for the payment and performance of all of Borrower's obligations under the "Note" (as defined in the Loan Agreement) in the original principal amount of Eight Million Dollars (\$8,000,000) and the other Loan Documents.

WHEREAS, Borrower and Investor desire to modify certain terms contained in the Note and Loan Agreement, and to reaffirm the Loan, as modified by this Agreement.

AGREEMENT:

In consideration of the foregoing premises and the mutual covenants set forth in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Investor and Borrower agree as follows.

ARTICLE I

1.1 <u>Definitions.</u> Capitalized terms not otherwise defined herein have the meanings ascribed to them in the Loan Agreement.

ARTICLE II

MODIFICATION OF THE LOAN AGREEMENT

- 2 . 1 <u>Financial Reporting of Total Stockholder Equity.</u> Section 6.A. (2) is hereby revised to add a reporting requirement with respect to Total Shareholder Equity; accordingly said section is amended and restated in its entirety to read as follows:
 - On or before the first Business Day which occurs forty-five (45) calendar days after the close of each Accounting Period, (i) Borrower's balance sheet as of the close of such Accounting Period and its income statement for that portion of the then current fiscal year through the end of such Accounting Period prepared in accordance with GAAP and certified as being complete, correct, and fairly representing its financial condition and results of operations by the chief financial officer of Borrower, subject to the absence of footnotes and year-end adjustments, (ii) a statement of changes in equity and cash flows for the period ended on such date, certified by the chief financial officer of Borrower, (iii) the calculation of the Debt Service Coverage Ratio and the calculation of Total Stockholder Equity demonstrating that Borrower is in compliance with Subsection 8.G. of this Agreement, together with any supporting calculations used to arrive at such calculations, certified by the chief financial officer of Borrower, and (iv) a completed Borrower's Officer's Compliance Certificate:
- 2 . 2 <u>Redemption.</u> Section 7.F. (2) is hereby revised to delete the limited right of Borrower to redeem certain shares of common stock of Borrower; accordingly said section is amended and restated in its entirety to read as follows:
 - (2) Borrower will not, except as allowed below, directly or indirectly redeem, retire, purchase, or otherwise acquire beneficially any shares of any class of its own stock now or hereafter outstanding or set apart any sum for any such purpose.
- 2.3 <u>Senior Management.</u> Section 7.H. (a)(iii) is hereby revised to replace John E. Baker with Erin D. Pickens as one of the three individuals currently comprising senior management; accordingly said section is amended and restated in its entirety to read as follows:
 - (iii) a majority of those three individuals currently comprising senior management, William H. Armstrong, President, Erin D. Pickens, Senior Vice President, and Kenneth N. Jones, General Counsel, cease to serve in their current positions; or

- 2.4 <u>Debt Service Default Exception</u>. Section 8.G. is hereby revised to add an exception to the Debt Service Coverage Ration default provision, upon the Borrower's satisfaction of certain other financial conditions; accordingly said section is amended and restated in its entirety to read as follows:
 - G. The Debt Service Coverage Ratio measured on a quarterly basis for the previous twelve (12) months shall be less than (1) (a) 5.0 minus (b) the product of 5.0 multiplied by the Cash Collateral Factor, to (2) 1.0, unless Borrower has Total Stockholder Equity equal to or greater than One Hundred Twenty Million Dollars (\$120,000,000).
- 2.5 New Definitions. Schedule 1 to Loan Agreement is hereby revised to add the following new definitions:

"Beal Guaranty" means that certain Amended and Restated Guaranty Agreement dated as of October 21, 2009 executed by Borrower and Canyon-Johnson Urban Fund II, L.P., as guarantors, for the benefit of Beal Bank Nevada in connection with a \$120,000,000.00 construction financing facility from Beal Bank Nevada to CJUF II Stratus Block 21, LLC evidenced by an Amended and Restated Construction Loan Agreement of even date therewith.

"Ford Guaranty" means that certain Guaranty Agreement dated as of the date hereof executed by Borrower, as guarantor, for the benefit of Hunter's Glen/Ford Investments 1 LLC ("Ford Lender") in connection with a Thirty Million Dollar (\$30,000,000) mezzanine construction financing facility from Ford Lender to CJUF II Stratus Block 21, LLC evidenced by a Loan Agreement of even date therewith.

"Total Stockholder Equity" means the amount shown on the line item entry set forth in Borrower's balance sheet (to be submitted to Lender in accordance with Section 6.A.(2) of the Agreement) for "Total Stratus stockholders' equity" (for illustrative purposes only, Total Stockholder Equity as of September 30, 2009 is listed on the Borrower's balance sheet as Total Stratus stockholders' equity in the amount of \$144,598,000); provided, if in the future Borrower deconsolidates financial statements of joint ventures from Borrower's financial statements, then the reference to "Total Stratus stockholders' equity" above will be deemed to refer to "Total stockholders' equity." Total Stockholder Equity shall be determined in accordance with GAAP, as in effect as of the date hereof. In the event the applicable accounting standard or any other aspect of GAAP are materially revised so as to alter the determination of Total Stockholder Equity, Total Stockholder Equity shall nevertheless be calculated in accordance with GAAP in effect as of the date hereof (so that the alternative minimum Total Stockholder Equity requirement set forth in Section 8.G. remains a fixed benchmark) and Borrower shall submit and certify the appropriate calculations separate and apart from the balance sheet it submits to Lender in accordance with Section 6.A.(2).

2.6 **Revised Definitions.** Schedule 1 to Loan Agreement is hereby revised to amend certain definitions; accordingly the following definitions are amended and restated in their entirety to read as follows:

"Comerica Loan Agreement" means that certain Loan Agreement dated as of September 30, 2005, among Borrower and certain Affiliates of Borrower and Comerica Bank-Texas- as modified by Modification and Extension Agreement dated as of May 30, 2006, Second Modification and Extension Agreement dated as of May 30, 2008, and Fourth Modification and Extension Agreement dated as of the date hereof.

"Maturity Date" means the Maturity Date set forth in the Note.

"Permitted Debt" means (i) the Loan and other Indebtedness to Lender or Related Lenders, (ii) the Comerica Debt (as of the date hereof), (iii) any other Indebtedness of Borrower for fair value received that is secured by assets owned by Borrower having an appraised value equal to or greater than the indebtedness secured thereby (and which assets do not secure other indebtedness), (iv) debt outstanding as of the date of the Loan Agreement, (v) unsecured trade, utility or non-extraordinary accounts payable in the ordinary course of business and other unsecured debt of Borrower at any one time not to exceed Five Hundred Thousand Dollars (\$500,000), and (vi) the Beal Guaranty (as of the date hereof), the Ford Guaranty (as of the date hereof), and guaranties of Borrower guaranteeing project development and/or construction costs and related costs, provided that Borrower has a direct or indirect interest in such projects and that the aggregate amount, at any one time, of such guaranties does not exceed the sum of Fifteen Million Dollars (\$15,000,000).

ARTICLE III

MODIFICATION OF THE NOTE

- 3.1 <u>Interest Rate</u>. The Regular Rate (as defined in **Section 2 of the Note**) shall be increased; accordingly the stated Regular Rate in the Note of 6.915% is hereby amended to so that the Regular Rate of interest shall be 8.75%.
- 3.2 <u>Prepayment Adjustments.</u> Sections 8.(a) and (b) of the Note are hereby revised to allow for prepayment of the Note, which would not be subject to any Reinvestment Charge (as defined in the original Loan Agreement); accordingly said sections are amended and restated in their entirety to read as follows:

- (a) Borrower may prepay this Note in whole or in part at any time (and shall be obligated to pay a portion of this Note in accordance with subsection 5.(a) above), subject to subsection 8.(c) below.
- (b) Any prepayment so made shall not be subject to any reinvestment charge or other prepayment premium.
- 3 . 3 Paydown of Principal. Notwithstanding the Maturity Date set forth in the Note (as revised hereby), Borrower shall be required to make two (2) paydowns of principal ("Principal Paydowns") in addition to all other payments due under the Note. Accordingly, Section 5. of the Note is hereby revised to set forth Borrower's obligation to make such payments by adding a new Section 5.(c) as follows:
 - (c) In addition to the all other payments due pursuant to the terms of this Note, including without limitation those set forth in Section 5.(a) above, Borrower shall be required to make two (2) paydowns of principal ("<u>Principal Paydowns</u>"). Borrower shall be obligated to make the first Principal Paydown in the amount of Two Million Dollars (\$2,000,000) on or before March 31, 2010. Borrower shall be obligated to make the second Principal Paydown in the amount of Two Million Dollars (\$2,000,000) on or before June 30, 2010. Neither of the two Principal Paydowns will relieve Borrower from making the monthly payments set forth in Section 5.(a) above, nor shall it affect the amount due pursuant to such section (i.e., the Loan will not be re-amortized based on such Principal Paydown).

ARTICLE IV

MISCELLANEOUS

- 4.1 No Other Amendments Intended. Except as specifically provided herein, no other amendment of the Loan Documents is intended and all other terms and conditions of the Note, the Loan Agreement and any other Loan Documents shall remain in full force and effect and shall not be modified or released in any way by this Agreement. This Agreement amends the Note and is not in payment or substitution thereof. Borrower hereby ratifies and reaffirms all of Borrower's obligations under the Note, the Loan Agreement and all of the other Loan Documents as amended hereby.
- 4.2 <u>No Impairment of Lien.</u> Nothing in this Agreement shall affect the lien of any of the Loan Documents or the priority of any such liens, nor release or change the liability of any party who may now be or after the date of this Agreement, become liable, primarily or secondarily, under the Loan Documents.
 - 4.3 Representations and Warranties of Borrower.

- (a) Representations and Warranties in Loan Documents. The representations and warranties of Borrower contained in the other Loan Documents, as amended hereby, are true and correct in all material respects as of the date first written above (as if such representations and warranties were made effective as of the date first written above).
- (b) <u>Power to Perform.</u> Borrower has the power, under its organizational documents, to enter into this Agreement and to perform the obligations required to be performed by Borrower under the terms hereunder.
- Due Authorization. The execution, delivery and performance by Borrower of this Agreement have been duly authorized by all necessary action on the part of Borrower. This Agreement has been duly executed and delivered by Borrower and, assuming the due execution and delivery of this Agreement by Investor, constitute the legal, valid and binding obligations of Borrower enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles affecting the enforcement of creditors' rights generally. No registration with, or consent or approval of, or notice to, or other action by, (i) any trustee or holder of any indebtedness or obligation of Borrower or (ii) any other person for the execution, delivery and performance of this Agreement by Borrower is required or, if required, such registration has been made, such consent, approval or notice given or such other appropriate action taken.

4.4 Miscellaneous.

- (a) Jurisdiction. This Agreement shall be construed according to and governed by the laws of the state of Minnesota.
- (b) <u>Severability: Counterparts.</u> If any provision of this Agreement is adjudicated to be invalid, illegal or enforceable, in whole or in part, it will be deemed omitted to that extent and all other provisions of this Agreement will remain in full force and effect. This Agreement may be executed in one or more counterparts and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed to be an original; such counterparts, together, shall constitute one and the same agreement.
- (c) <u>Notices.</u> All notices given pursuant to this Agreement must be in writing and will be effectively given if personally delivered or, if mailed, postage prepaid, certified or registered mail, return receipt requested, to the addresses of Investor and Borrower first set forth above or to such other address as any party subsequently may designate in writing.
- (d) Assignment. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, legal representatives, successors and assigns. This Agreement may not be assigned by Borrower without the prior written consent of Investor. This Agreement may be assigned by Investor in connection with an assignment of the Loan without any required notice to Borrower.

((e) <u>C</u>	osts; Fee; Furtl	<u>her Assurances</u> .	Borrower agree	es to pay Invest	tor's out	t-of-pocket ex	penses in conn	ection wi	th the prepa	ration of
this Agreement an	nd any rela	ated expenses,	including without	out limitation,	reasonable atto	rneys' f	fees. In additi	ion, Borrower	agrees to	execute suc	ch other
instruments as may	be reason:	ably required b	y Investor to evi	dence or facilit	tate the agreeme	nts set f	forth herein.				

[The remainder of this page was intentionally left blank.]

LOAN MODIFICATION AGREEMENT

BORROWER SIGNATURE PAGE

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the day and year first above written.

BORROWER:

STRATUS PROPERTIES INC.,

a Delaware corporation

By: <u>/s/ Erin D. Pickens</u>
Name: Erin D. Pickens
Title: Senior Vice President

LOAN MODIFICATION AGREEMENT

INVESTOR SIGNATURE PAGE

VITNESS WHEREOF, the parties have executed this Agreement as of the c	date first above written.
	INVESTOR: AMERICAN STRATEGIC INCOME PORTFOLIO INC.—III, a Minnesota corporation
	By: Name: Its:
	9

2006 CSP \$7M Loan

LOAN MODIFICATION AGREEMENT

THIS LOAN MODIFICATION AGREEMENT (this "**Agreement**") is made as of March ____, 2010, by and between STRATUS PROPERTIES INC., a Delaware corporation having an address of 98 San Jacinto Boulevard, Suite 220, Austin, TX 78791 ("**Borrower**"), and AMERICAN STRATEGIC INCOME PORTFOLIO INC.—III, a Minnesota corporation having an address at c/o FAF Advisors, Inc., 800 Nicollet Mall, Suite 500, BC-MN-H05W, Minneapolis, MN 55402 ("**Investor**").

RECITALS:

WHEREAS, Investor holds the lender's interest in that certain "Loan" and other "Loan Documents" as defined and described in that certain Loan Agreement dated as of December 12, 2006 between Borrower and Holliday Fenoglio Fowler, L.P., a Texas limited partnership ("Lender") (the "Loan Agreement").

WHEREAS, Borrower is liable for the payment and performance of all of Borrower's obligations under the "Note" (as defined in the Loan Agreement) in the original principal amount of Seven Million Dollars (\$7,000,000) and the other Loan Documents.

WHEREAS, Borrower and Investor desire to modify certain terms contained in the Note and Loan Agreement, and to reaffirm the Loan, as modified by this Agreement.

AGREEMENT:

In consideration of the foregoing premises and the mutual covenants set forth in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Investor and Borrower agree as follows.

ARTICLE I

1.1 <u>Definitions</u>. Capitalized terms not otherwise defined herein have the meanings ascribed to them in the Loan Agreement.

ARTICLE II

MODIFICATION OF THE LOAN AGREEMENT

- 2 . 1 <u>Financial Reporting of Total Stockholder Equity.</u> Section 6.A. (2) is hereby revised to add a reporting requirement with respect to Total Shareholder Equity; accordingly said section is amended and restated in its entirety to read as follows:
 - On or before the first Business Day which occurs forty-five (45) calendar days after the close of each Accounting Period, (i) Borrower's balance sheet as of the close of such Accounting Period and its income statement for that portion of the then current fiscal year through the end of such Accounting Period prepared in accordance with GAAP and certified as being complete, correct, and fairly representing its financial condition and results of operations by the chief financial officer of Borrower, subject to the absence of footnotes and year-end adjustments, (ii) a statement of changes in equity and cash flows for the period ended on such date, certified by the chief financial officer of Borrower, (iii) the calculation of the Debt Service Coverage Ratio and the calculation of Total Stockholder Equity demonstrating that Borrower is in compliance with Subsection 8.G. of this Agreement, together with any supporting calculations used to arrive at such calculations, certified by the chief financial officer of Borrower, and (iv) a completed Borrower's Officer's Compliance Certificate:
- 2 . 2 <u>Redemption.</u> Section 7.F. (2) is hereby revised to delete the limited right of Borrower to redeem certain shares of common stock of Borrower; accordingly said section is amended and restated in its entirety to read as follows:
 - (2) Borrower will not, except as allowed below, directly or indirectly redeem, retire, purchase, or otherwise acquire beneficially any shares of any class of its own stock now or hereafter outstanding or set apart any sum for any such purpose.
- 2.3 <u>Senior Management.</u> Section 7.H. (a)(iii) is hereby revised to replace John E. Baker with Erin D. Pickens as one of the three individuals currently comprising senior management; accordingly said section is amended and restated in its entirety to read as follows:
 - (iii) a majority of those three individuals currently comprising senior management, William H. Armstrong, President, Erin D. Pickens, Senior Vice President, and Kenneth N. Jones, General Counsel, cease to serve in their current positions; or

- 2.4 <u>Debt Service Default Exception</u>. Section 8.G. is hereby revised to add an exception to the Debt Service Coverage Ration default provision, upon the Borrower's satisfaction of certain other financial conditions; accordingly said section is amended and restated in its entirety to read as follows:
 - G. The Debt Service Coverage Ratio measured on a quarterly basis for the previous twelve (12) months shall be less than (1) (a) 5.0 minus (b) the product of 5.0 multiplied by the Cash Collateral Factor, to (2) 1.0, unless Borrower has Total Stockholder Equity equal to or greater than One Hundred Twenty Million Dollars (\$120,000,000).
- 2.5 New Definitions. Schedule 1 to Loan Agreement is hereby revised to add the following new definitions:

"Beal Guaranty" means that certain Amended and Restated Guaranty Agreement dated as of October 21, 2009 executed by Borrower and Canyon-Johnson Urban Fund II, L.P., as guarantors, for the benefit of Beal Bank Nevada in connection with a \$120,000,000.00 construction financing facility from Beal Bank Nevada to CJUF II Stratus Block 21, LLC evidenced by an Amended and Restated Construction Loan Agreement of even date therewith.

"Ford Guaranty" means that certain Guaranty Agreement dated as of the date hereof executed by Borrower, as guarantor, for the benefit of Hunter's Glen/Ford Investments 1 LLC ("Ford Lender") in connection with a Thirty Million Dollar (\$30,000,000) mezzanine construction financing facility from Ford Lender to CJUF II Stratus Block 21, LLC evidenced by a Loan Agreement of even date therewith.

"Total Stockholder Equity" means the amount shown on the line item entry set forth in Borrower's balance sheet (to be submitted to Lender in accordance with Section 6.A.(2) of the Agreement) for "Total Stratus stockholders' equity" (for illustrative purposes only, Total Stockholder Equity as of September 30, 2009 is listed on the Borrower's balance sheet as Total Stratus stockholders' equity in the amount of \$144,598,000); provided, if in the future Borrower deconsolidates financial statements of joint ventures from Borrower's financial statements, then the reference to "Total Stratus stockholders' equity" above will be deemed to refer to "Total stockholders' equity." Total Stockholder Equity shall be determined in accordance with GAAP, as in effect as of the date hereof. In the event the applicable accounting standard or any other aspect of GAAP are materially revised so as to alter the determination of Total Stockholder Equity, Total Stockholder Equity shall nevertheless be calculated in accordance with GAAP in effect as of the date hereof (so that the alternative minimum Total Stockholder Equity requirement set forth in Section 8.G. remains a fixed benchmark) and Borrower shall submit and certify the appropriate calculations separate and apart from the balance sheet it submits to Lender in accordance with Section 6.A.(2).

2 . 6 Revised Definitions. Schedule 1 to Loan Agreement is hereby revised to amend certain definitions; accordingly the following definitions are amended and restated in their entirety to read as follows:

"Comerica Loan Agreement" means that certain Loan Agreement dated as of September 30, 2005, among Borrower and certain Affiliates of Borrower and Comerica Bank-Texas- as modified by Modification and Extension Agreement dated as of May 30, 2006, Second Modification and Extension Agreement dated as of May 30, 2008, and Fourth Modification and Extension Agreement dated as of the date hereof.

"Maturity Date" means the Maturity Date set forth in the Note.

"Permitted Debt" means (i) the Loan and other Indebtedness to Lender or Related Lenders, (ii) the Comerica Debt (as of the date hereof), (iii) any other Indebtedness of Borrower for fair value received that is secured by assets owned by Borrower having an appraised value equal to or greater than the indebtedness secured thereby (and which assets do not secure other indebtedness), (iv) debt outstanding as of the date of the Loan Agreement, (v) unsecured trade, utility or non-extraordinary accounts payable in the ordinary course of business and other unsecured debt of Borrower at any one time not to exceed Five Hundred Thousand Dollars (\$500,000), and (vi) the Beal Guaranty (as of the date hereof), the Ford Guaranty (as of the date hereof), and guaranties of Borrower guaranteeing project development and/or construction costs and related costs, provided that Borrower has a direct or indirect interest in such projects and that the aggregate amount, at any one time, of such guaranties does not exceed the sum of Fifteen Million Dollars (\$15,000,000).

ARTICLE III

MODIFICATION OF THE NOTE

- 3.1 Extended Maturity Date. The Maturity Date (as defined in Section 5.(a) of the Note) shall be extended; accordingly the stated Maturity Date in the Note of December 31, 2011 is hereby amended to so that the Maturity Date for the Note shall be December 31, 2013.
- 3.2 <u>Interest Rate</u>. The Regular Rate (as defined in **Section 2 of the Note**) shall be increased; accordingly the stated Regular Rate in the Note of 6.560% is hereby amended to so that the Regular Rate of interest shall be 8.75%.
- 3.3 <u>Prepayment Adjustments</u>. Sections 8.(a) and (b) of the Note are hereby revised to adjust the applicable time frames and corresponding amounts of any Reinvestment Charge due

under the terms of the Loan Documents; accordingly said sections are amended and restated in their entirety to read as follows:

- (a) Prior to December 31, 2010, this Note may not be prepaid, except as may be expressly provided for in the Loan Agreement.
- (b) Beginning on January 1, 2011, the Borrower may prepay this Note in whole or in part, subject to subsection 8(c) below, provided such prepayment is accompanied by a reinvestment charge (hereinafter referred to as the "Reinvestment Charge"). The Reinvestment Charge shall be an amount equal to (i) two percent (2.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2011 through December 31, 2011, (ii) one percent (1.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2012 through December 31, 2012, and (iii) zero percent (0%) of the principal amount prepaid with respect to prepayments made after December 31, 2012.

ARTICLE IV

MISCELLANEOUS

- 4.1 No Other Amendments Intended. Except as specifically provided herein, no other amendment of the Loan Documents is intended and all other terms and conditions of the Note, the Loan Agreement and any other Loan Documents shall remain in full force and effect and shall not be modified or released in any way by this Agreement. This Agreement amends the Note and is not in payment or substitution thereof. Borrower hereby ratifies and reaffirms all of Borrower's obligations under the Note, the Loan Agreement and all of the other Loan Documents as amended hereby.
- 4.2 **No Impairment of Lien.** Nothing in this Agreement shall affect the lien of any of the Loan Documents or the priority of any such liens, nor release or change the liability of any party who may now be or after the date of this Agreement, become liable, primarily or secondarily, under the Loan Documents.
 - 4.3 Representations and Warranties of Borrower.
- (a) <u>Representations and Warranties in Loan Documents</u>. The representations and warranties of Borrower contained in the other Loan Documents, as amended hereby, are true and correct in all material respects as of the date first written above (as if such representations and warranties were made effective as of the date first written above).
- (b) <u>Power to Perform.</u> Borrower has the power, under its organizational documents, to enter into this Agreement and to perform the obligations required to be performed by Borrower under the terms hereunder.

Due Authorization. The execution, delivery and performance by Borrower of this Agreement have been duly authorized by all necessary action on the part of Borrower. This Agreement has been duly executed and delivered by Borrower and, assuming the due execution and delivery of this Agreement by Investor, constitute the legal, valid and binding obligations of Borrower enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles affecting the enforcement of creditors' rights generally. No registration with, or consent or approval of, or notice to, or other action by, (i) any trustee or holder of any indebtedness or obligation of Borrower or (ii) any other person for the execution, delivery and performance of this Agreement by Borrower is required or, if required, such registration has been made, such consent, approval or notice given or such other appropriate action taken.

4.4 Miscellaneous.

- (a) Jurisdiction. This Agreement shall be construed according to and governed by the laws of the state of Minnesota.
- (b) <u>Severability: Counterparts.</u> If any provision of this Agreement is adjudicated to be invalid, illegal or enforceable, in whole or in part, it will be deemed omitted to that extent and all other provisions of this Agreement will remain in full force and effect. This Agreement may be executed in one or more counterparts and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed to be an original; such counterparts, together, shall constitute one and the same agreement.
- (c) <u>Notices.</u> All notices given pursuant to this Agreement must be in writing and will be effectively given if personally delivered or, if mailed, postage prepaid, certified or registered mail, return receipt requested, to the addresses of Investor and Borrower first set forth above or to such other address as any party subsequently may designate in writing.
- (d) <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, legal representatives, successors and assigns. This Agreement may not be assigned by Borrower without the prior written consent of Investor. This Agreement may be assigned by Investor in connection with an assignment of the Loan without any required notice to Borrower.
- (e) <u>Costs; Fee; Further Assurances</u>. Borrower agrees to pay Investor's out-of-pocket expenses in connection with the preparation of this Agreement and any related expenses, including without limitation, reasonable attorneys' fees. In addition, Borrower agrees to execute such other instruments as may be reasonably required by Investor to evidence or facilitate the agreements set forth herein.

BORROWER SIGNATURE PAGE

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the day and year first above written.

BORROWER:

STRATUS PROPERTIES INC.,

a Delaware corporation
By: /s/ Erin D. Pickens
Name: Erin D. Pickens
Title: Senior Vice President

INVESTOR SIGNATURE PAGE

N WITNESS WHEREOF, the parties have executed this Agreement as of the date first about	ove written.
	INVESTOR:
	AMERICAN STRATEGIC INCOME PORTFOLIO INC.— III, a Minnesota corporation
	By: Name: Its:
9	

2007 SLA \$3.5M Loan

LOAN MODIFICATION AGREEMENT

THIS LOAN MODIFICATION AGREEMENT (this "Agreement") is made as of March ___, 2010, by and between STRATUS PROPERTIES INC., a Delaware corporation having an address of 98 San Jacinto Boulevard, Suite 220, Austin, TX 78791 ("Borrower"), and AMERICAN SELECT PORTFOLIO INC., a Minnesota corporation having an address at c/o FAF Advisors, Inc., 800 Nicollet Mall, Suite 500, BC-MN-H05W, Minneapolis, MN 55402 ("Investor").

RECITALS:

WHEREAS, Investor holds the lender's interest in that certain "Loan" and other "Loan Documents" as defined and described in that certain Loan Agreement dated as of June 1, 2007 between Borrower and Holliday Fenoglio Fowler, L.P., a Texas limited partnership ("Lender") (the "Loan Agreement").

WHEREAS, Borrower is liable for the payment and performance of all of Borrower's obligations under the "**Note**" (as defined in the Loan Agreement) in the original principal amount of Three Million Five Hundred Thousand Dollars (\$3,500,000) and the other Loan Documents.

WHEREAS, Borrower and Investor desire to modify certain terms contained in the Note and Loan Agreement, and to reaffirm the Loan, as modified by this Agreement.

AGREEMENT:

In consideration of the foregoing premises and the mutual covenants set forth in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Investor and Borrower agree as follows.

ARTICLE I

1.1 <u>Definitions.</u> Capitalized terms not otherwise defined herein have the meanings ascribed to them in the Loan Agreement.

ARTICLE II

MODIFICATION OF THE LOAN AGREEMENT

- 2 . 1 <u>Financial Reporting of Total Stockholder Equity.</u> Section 6.A. (2) is hereby revised to add a reporting requirement with respect to Total Shareholder Equity; accordingly said section is amended and restated in its entirety to read as follows:
 - On or before the first Business Day which occurs forty-five (45) calendar days after the close of each Accounting Period, (i) Borrower's balance sheet as of the close of such Accounting Period and its income statement for that portion of the then current fiscal year through the end of such Accounting Period prepared in accordance with GAAP and certified as being complete, correct, and fairly representing its financial condition and results of operations by the chief financial officer of Borrower, subject to the absence of footnotes and year-end adjustments, (ii) a statement of changes in equity and cash flows for the period ended on such date, certified by the chief financial officer of Borrower, (iii) the calculation of the Debt Service Coverage Ratio and the calculation of Total Stockholder Equity demonstrating that Borrower is in compliance with Subsection 8.G. of this Agreement, together with any supporting calculations used to arrive at such calculations, certified by the chief financial officer of Borrower, and (iv) a completed Borrower's Officer's Compliance Certificate;
- 2 . 2 Redemption. Section 7.F. (2) is hereby revised to delete the limited right of Borrower to redeem certain shares of common stock of Borrower, accordingly said section is amended and restated in its entirety to read as follows:
 - (2) Borrower will not, except as allowed below, directly or indirectly redeem, retire, purchase, or otherwise acquire beneficially any shares of any class of its own stock now or hereafter outstanding or set apart any sum for any such purpose.
- 2.3 <u>Senior Management.</u> Section 7.H. (a)(iii) is hereby revised to replace John E. Baker with Erin D. Pickens as one of the three individuals currently comprising senior management; accordingly said section is amended and restated in its entirety to read as follows:
 - (iii) a majority of those three individuals currently comprising senior management, William H. Armstrong, President, Erin D. Pickens, Senior Vice President, and Kenneth N. Jones, General Counsel, cease to serve in their current positions; or
- 2.4 <u>Debt Service Default Exception.</u> Section 8.G. is hereby revised to add an exception to the Debt Service Coverage Ration default provision, upon the Borrower's satisfaction of certain other financial conditions; accordingly said section is amended and restated in its entirety to read as follows

- G. The Debt Service Coverage Ratio measured on a quarterly basis for the previous twelve (12) months shall be less than (1) (a) 5.0 minus (b) the product of 5.0 multiplied by the Cash Collateral Factor, to (2) 1.0, unless Borrower has Total Stockholder Equity equal to or greater than One Hundred Twenty Million Dollars (\$120,000,000).
- 2.5 New Definitions. Schedule 1 to Loan Agreement is hereby revised to add the following new definitions:

"Beal Guaranty" means that certain Amended and Restated Guaranty Agreement dated as of October 21, 2009 executed by Borrower and Canyon-Johnson Urban Fund II, L.P., as guarantors, for the benefit of Beal Bank Nevada in connection with a \$120,000,000.00 construction financing facility from Beal Bank Nevada to CJUF II Stratus Block 21, LLC evidenced by an Amended and Restated Construction Loan Agreement of even date therewith.

"Ford Guaranty" means that certain Guaranty Agreement dated as of the date hereof executed by Borrower, as guarantor, for the benefit of Hunter's Glen/Ford Investments 1 LLC ("Ford Lender") in connection with a Thirty Million Dollar (\$30,000,000) mezzanine construction financing facility from Ford Lender to CJUF II Stratus Block 21, LLC evidenced by a Loan Agreement of even date therewith.

"Total Stockholder Equity" means the amount shown on the line item entry set forth in Borrower's balance sheet (to be submitted to Lender in accordance with Section 6.A.(2) of the Agreement) for "Total Stratus stockholders' equity" (for illustrative purposes only, Total Stockholder Equity as of September 30, 2009 is listed on the Borrower's balance sheet as Total Stratus stockholders' equity in the amount of \$144,598,000); provided, if in the future Borrower deconsolidates financial statements of joint ventures from Borrower's financial statements, then the reference to "Total Stratus stockholders' equity" above will be deemed to refer to "Total stockholders' equity." Total Stockholder Equity shall be determined in accordance with GAAP, as in effect as of the date hereof. In the event the applicable accounting standard or any other aspect of GAAP are materially revised so as to alter the determination of Total Stockholder Equity, Total Stockholder Equity shall nevertheless be calculated in accordance with GAAP in effect as of the date hereof (so that the alternative minimum Total Stockholder Equity requirement set forth in Section 8.G. remains a fixed benchmark) and Borrower shall submit and certify the appropriate calculations separate and apart from the balance sheet it submits to Lender in accordance with Section 6.A.(2).

2 . 6 Revised Definitions. Schedule 1 to Loan Agreement is hereby revised to amend certain definitions; accordingly the following definitions are amended and restated in their entirety to read as follows:

"Comerica Loan Agreement" means that certain Loan Agreement dated as of September 30, 2005, among Borrower and certain Affiliates of Borrower and Comerica Bank-Texas- as modified by Modification and Extension Agreement dated as of May 30, 2006, Second Modification and Extension Agreement dated as of May 30, 2008, and Fourth Modification and Extension Agreement dated as of the date hereof.

"Maturity Date" means the Maturity Date set forth in the Note.

"Permitted Debt" means (i) the Loan and other Indebtedness to Lender or Related Lenders, (ii) the Comerica Debt (as of the date hereof), (iii) any other Indebtedness of Borrower for fair value received that is secured by assets owned by Borrower having an appraised value equal to or greater than the indebtedness secured thereby (and which assets do not secure other indebtedness), (iv) debt outstanding as of the date of the Loan Agreement, (v) unsecured trade, utility or non-extraordinary accounts payable in the ordinary course of business and other unsecured debt of Borrower at any one time not to exceed Five Hundred Thousand Dollars (\$500,000), and (vi) the Beal Guaranty (as of the date hereof), the Ford Guaranty (as of the date hereof), and guaranties of Borrower guaranteeing project development and/or construction costs and related costs, provided that Borrower has a direct or indirect interest in such projects and that the aggregate amount, at any one time, of such guaranties does not exceed the sum of Fifteen Million Dollars (\$15,000,000).

ARTICLE III

MODIFICATION OF THE NOTE

- 3.1 Extended Maturity Date. The Maturity Date (as defined in Section 5.(a) of the Note) shall be extended; accordingly the stated Maturity Date in the Note of December 31, 2011 is hereby amended to so that the Maturity Date for the Note shall be December 31, 2012.
- 3.2 <u>Interest Rate</u>. The Regular Rate (as defined in **Section 2 of the Note**) shall be increased; accordingly the stated Regular Rate in the Note of 6.915% is hereby amended to so that the Regular Rate of interest shall be 8.75%.
- 3.3 <u>Prepayment Adjustments.</u> Sections 8.(a) and (b) of the Note is hereby revised to adjust the applicable time frames and corresponding amounts of any Reinvestment Charge due under the terms of the Loan Documents; accordingly said sections are amended and restated in their entirety to read as follows:
 - (a) Prior to December 31, 2010, this Note may not be prepaid, except as may be expressly provided for in the Loan Agreement.

(b) Beginning on January 1, 2011, the Borrower may prepay this Note in whole or in part, subject to subsection 8(c) below, provided such prepayment is accompanied by a reinvestment charge (hereinafter referred to as the "Reinvestment Charge"). The Reinvestment Charge shall be an amount equal to (i) one percent (1.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2011 through December 31, 2011, and (ii) zero percent (0%) of the principal amount prepaid with respect to prepayments made after December 31, 2011.

ARTICLE IV

MISCELLANEOUS

- 4.1 **No Other Amendments Intended.** Except as specifically provided herein, no other amendment of the Loan Documents is intended and all other terms and conditions of the Note, the Loan Agreement and any other Loan Documents shall remain in full force and effect and shall not be modified or released in any way by this Agreement. This Agreement amends the Note and is not in payment or substitution thereof. Borrower hereby ratifies and reaffirms all of Borrower's obligations under the Note, the Loan Agreement and all of the other Loan Documents as amended hereby.
- 4.2 <u>No Impairment of Lien.</u> Nothing in this Agreement shall affect the lien of any of the Loan Documents or the priority of any such liens, nor release or change the liability of any party who may now be or after the date of this Agreement, become liable, primarily or secondarily, under the Loan Documents.
 - 4.3 Representations and Warranties of Borrower.
- (a) Representations and Warranties in Loan Documents. The representations and warranties of Borrower contained in the other Loan Documents, as amended hereby, are true and correct in all material respects as of the date first written above (as if such representations and warranties were made effective as of the date first written above).
- (b) <u>Power to Perform</u>. Borrower has the power, under its organizational documents, to enter into this Agreement and to perform the obligations required to be performed by Borrower under the terms hereunder.
- (c) <u>Due Authorization</u>. The execution, delivery and performance by Borrower of this Agreement have been duly authorized by all necessary action on the part of Borrower. This Agreement has been duly executed and delivered by Borrower and, assuming the due execution and delivery of this Agreement by Investor, constitute the legal, valid and binding obligations of Borrower enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles affecting the enforcement of creditors' rights generally. No registration with, or consent or approval of, or notice to, or other action by, (i) any trustee or holder of any

indebtedness or obligation of Borrower or (ii) any other person for the execution, delivery and performance of this Agreement by Borrower is required or, if required, such registration has been made, such consent, approval or notice given or such other appropriate action taken.

4.4 Miscellaneous.

- (a) Jurisdiction. This Agreement shall be construed according to and governed by the laws of the state of Minnesota.
- (b) <u>Severability; Counterparts.</u> If any provision of this Agreement is adjudicated to be invalid, illegal or enforceable, in whole or in part, it will be deemed omitted to that extent and all other provisions of this Agreement will remain in full force and effect. This Agreement may be executed in one or more counterparts and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed to be an original; such counterparts, together, shall constitute one and the same agreement.
- (c) <u>Notices.</u> All notices given pursuant to this Agreement must be in writing and will be effectively given if personally delivered or, if mailed, postage prepaid, certified or registered mail, return receipt requested, to the addresses of Investor and Borrower first set forth above or to such other address as any party subsequently may designate in writing.
- (d) Assignment. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, legal representatives, successors and assigns. This Agreement may not be assigned by Borrower without the prior written consent of Investor. This Agreement may be assigned by Investor in connection with an assignment of the Loan without any required notice to Borrower.
- (e) <u>Costs; Fee; Further Assurances</u>. Borrower agrees to pay Investor's out-of-pocket expenses in connection with the preparation of this Agreement and any related expenses, including without limitation, reasonable attorneys' fees. In addition, Borrower agrees to execute such other instruments as may be reasonably required by Investor to evidence or facilitate the agreements set forth herein.

[The remainder of this page was intentionally left blank.]

BORROWER SIGNATURE PAGE

N WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed	as of the day and year first above written.	
	BORROWER:	
	STRATUS PROPERTIES INC., a Delaware corporation	
	B y : /s/ Erin Pickens Name: Erin D. Pickens Title: Senior Vice President	D.
7		

INVESTOR SIGNATURE PAGE

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above written.			
	INVESTOR:		
	AMERICAN SELECT PORTFOLIO INC., a Minnesota corporation		
	By:		
	- Name:		
	Its:		
	8		

2006 SLA \$5M Loan

LOAN MODIFICATION AGREEMENT

THIS LOAN MODIFICATION AGREEMENT (this "Agreement") is made as of March ___, 2010, by and between STRATUS PROPERTIES INC., a Delaware corporation having an address of 98 San Jacinto Boulevard, Suite 220, Austin, TX 78791 ("Borrower"), and AMERICAN SELECT PORTFOLIO INC., a Minnesota corporation having an address at c/o FAF Advisors, Inc., 800 Nicollet Mall, Suite 500, BC-MN-H05W, Minneapolis, MN 55402 ("Investor").

RECITALS:

WHEREAS, Investor holds the lender's interest in that certain "Loan" and other "Loan Documents" as defined and described in that certain Amended and Restated Loan Agreement dated as of December 12, 2006 between Borrower and Holliday Fenoglio Fowler, L.P., a Texas limited partnership ("Lender") (the "Loan Agreement").

WHEREAS, Borrower is liable for the payment and performance of all of Borrower's obligations under the "Note" (as defined in the Loan Agreement) in the original principal amount of Five Million Dollars (\$5,000,000) and the other Loan Documents.

WHEREAS, Borrower and Investor desire to modify certain terms contained in the Note and Loan Agreement, and to reaffirm the Loan, as modified by this Agreement.

AGREEMENT:

In consideration of the foregoing premises and the mutual covenants set forth in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Investor and Borrower agree as follows.

ARTICLE I

1.1 <u>Definitions.</u> Capitalized terms not otherwise defined herein have the meanings ascribed to them in the Loan Agreement.

ARTICLE II

MODIFICATION OF THE LOAN AGREEMENT

- 2 . 1 <u>Financial Reporting of Total Stockholder Equity.</u> Section 6.A. (2) is hereby revised to add a reporting requirement with respect to Total Shareholder Equity; accordingly said section is amended and restated in its entirety to read as follows:
 - On or before the first Business Day which occurs forty-five (45) calendar days after the close of each Accounting Period, (i) Borrower's balance sheet as of the close of such Accounting Period and its income statement for that portion of the then current fiscal year through the end of such Accounting Period prepared in accordance with GAAP and certified as being complete, correct, and fairly representing its financial condition and results of operations by the chief financial officer of Borrower, subject to the absence of footnotes and year-end adjustments, (ii) a statement of changes in equity and cash flows for the period ended on such date, certified by the chief financial officer of Borrower, (iii) the calculation of the Debt Service Coverage Ratio and the calculation of Total Stockholder Equity demonstrating that Borrower is in compliance with Subsection 8.G. of this Agreement, together with any supporting calculations used to arrive at such calculations, certified by the chief financial officer of Borrower, and (iv) a completed Borrower's Officer's Compliance Certificate:
- 2 . 2 <u>Redemption.</u> Section 7.F. (2) is hereby revised to delete the limited right of Borrower to redeem certain shares of common stock of Borrower; accordingly said section is amended and restated in its entirety to read as follows:
 - (2) Borrower will not, except as allowed below, directly or indirectly redeem, retire, purchase, or otherwise acquire beneficially any shares of any class of its own stock now or hereafter outstanding or set apart any sum for any such purpose.
- 2.3 <u>Senior Management.</u> Section 7.H. (a)(iii) is hereby revised to replace John E. Baker with Erin D. Pickens as one of the three individuals currently comprising senior management; accordingly said section is amended and restated in its entirety to read as follows:
 - (iii) a majority of those three individuals currently comprising senior management, William H. Armstrong, President, Erin D. Pickens, Senior Vice President, and Kenneth N. Jones, General Counsel, cease to serve in their current positions; or

- 2.4 <u>Debt Service Default Exception</u>. Section 8.G. is hereby revised to add an exception to the Debt Service Coverage Ration default provision, upon the Borrower's satisfaction of certain other financial conditions; accordingly said section is amended and restated in its entirety to read as follows:
 - G. The Debt Service Coverage Ratio measured on a quarterly basis for the previous twelve (12) months shall be less than (1) (a) 5.0 minus (b) the product of 5.0 multiplied by the Cash Collateral Factor, to (2) 1.0, unless Borrower has Total Stockholder Equity equal to or greater than One Hundred Twenty Million Dollars (\$120,000,000).
- 2.5 New Definitions. Schedule 1 to Loan Agreement is hereby revised to add the following new definitions:

"Beal Guaranty" means that certain Amended and Restated Guaranty Agreement dated as of October 21, 2009 executed by Borrower and Canyon-Johnson Urban Fund II, L.P., as guarantors, for the benefit of Beal Bank Nevada in connection with a \$120,000,000.00 construction financing facility from Beal Bank Nevada to CJUF II Stratus Block 21, LLC evidenced by an Amended and Restated Construction Loan Agreement of even date therewith.

"Ford Guaranty" means that certain Guaranty Agreement dated as of the date hereof executed by Borrower, as guarantor, for the benefit of Hunter's Glen/Ford Investments 1 LLC ("Ford Lender") in connection with a Thirty Million Dollar (\$30,000,000) mezzanine construction financing facility from Ford Lender to CJUF II Stratus Block 21, LLC evidenced by a Loan Agreement of even date therewith.

"Total Stockholder Equity" means the amount shown on the line item entry set forth in Borrower's balance sheet (to be submitted to Lender in accordance with Section 6.A.(2) of the Agreement) for "Total Stratus stockholders' equity" (for illustrative purposes only, Total Stockholder Equity as of September 30, 2009 is listed on the Borrower's balance sheet as Total Stratus stockholders' equity in the amount of \$144,598,000); provided, if in the future Borrower deconsolidates financial statements of joint ventures from Borrower's financial statements, then the reference to "Total Stratus stockholders' equity" above will be deemed to refer to "Total stockholders' equity." Total Stockholder Equity shall be determined in accordance with GAAP, as in effect as of the date hereof. In the event the applicable accounting standard or any other aspect of GAAP are materially revised so as to alter the determination of Total Stockholder Equity, Total Stockholder Equity shall nevertheless be calculated in accordance with GAAP in effect as of the date hereof (so that the alternative minimum Total Stockholder Equity requirement set forth in Section 8.G. remains a fixed benchmark) and Borrower shall submit and certify the appropriate calculations separate and apart from the balance sheet it submits to Lender in accordance with Section 6.A.(2).

2 . 6 Revised Definitions. Schedule 1 to Loan Agreement is hereby revised to amend certain definitions; accordingly the following definitions are amended and restated in their entirety to read as follows:

"Comerica Loan Agreement" means that certain Loan Agreement dated as of September 30, 2005, among Borrower and certain Affiliates of Borrower and Comerica Bank-Texas- as modified by Modification and Extension Agreement dated as of May 30, 2006, Second Modification and Extension Agreement dated as of May 30, 2008, and Fourth Modification and Extension Agreement as of the date hereof.

"Maturity Date" means the Maturity Date set forth in the Note.

"Permitted Debt" means (i) the Loan and other Indebtedness to Lender or Related Lenders, (ii) the Comerica Debt (as of the date hereof), (iii) any other Indebtedness of Borrower for fair value received that is secured by assets owned by Borrower having an appraised value equal to or greater than the indebtedness secured thereby (and which assets do not secure other indebtedness), (iv) debt outstanding as of the date of the Loan Agreement, (v) unsecured trade, utility or non-extraordinary accounts payable in the ordinary course of business and other unsecured debt of Borrower at any one time not to exceed Five Hundred Thousand Dollars (\$500,000), and (vi) the Beal Guaranty (as of the date hereof), the Ford Guaranty (as of the date hereof), and guaranties of Borrower guaranteeing project development and/or construction costs and related costs, provided that Borrower has a direct or indirect interest in such projects and that the aggregate amount, at any one time, of such guaranties does not exceed the sum of Fifteen Million Dollars (\$15,000,000).

ARTICLE III

MODIFICATION OF THE NOTE

- 3.1 Extended Maturity Date. The Maturity Date (as defined in Section 5.(a) of the Note) shall be extended; accordingly the stated Maturity Date in the Note of December 31, 2011 is hereby amended to so that the Maturity Date for the Note shall be December 31, 2014.
- 3.2 <u>Interest Rate</u>. The Regular Rate (as defined in **Section 2 of the Note**) shall be increased; accordingly the stated Regular Rate in the Note of 6.560% is hereby amended to so that the Regular Rate of interest shall be 8.75%.
- 3.3 <u>Prepayment Adjustments</u>. Sections 8.(a) and (b) of the Note are hereby revised to adjust the applicable time frames and corresponding amounts of any Reinvestment Charge due

under the terms of the Loan Documents; accordingly said sections are amended and restated in their entirety to read as follows.

- (a) Prior to December 31, 2011, this Note may not be prepaid, except as may be expressly provided for in the Loan Agreement.
- (b) Beginning on January 1, 2012, the Borrower may prepay this Note in whole or in part, subject to subsection 8(c) below, provided such prepayment is accompanied by a reinvestment charge (hereinafter referred to as the "Reinvestment Charge"). The Reinvestment Charge shall be an amount equal to (i) two percent (2.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2012 through December 31, 2012, (ii) one percent (1.0%) of the principal amount prepaid with respect to prepayments made between January 1, 2013 through December 31, 2013, and (iii) zero percent (0%) of the principal amount prepaid with respect to prepayments made after December 31, 2013.

ARTICLE IV

MISCELLANEOUS

- 4.1 No Other Amendments Intended. Except as specifically provided herein, no other amendment of the Loan Documents is intended and all other terms and conditions of the Note, the Loan Agreement and any other Loan Documents shall remain in full force and effect and shall not be modified or released in any way by this Agreement. This Agreement amends the Note and is not in payment or substitution thereof. Borrower hereby ratifies and reaffirms all of Borrower's obligations under the Note, the Loan Agreement and all of the other Loan Documents as amended hereby.
- 4.2 **No Impairment of Lien.** Nothing in this Agreement shall affect the lien of any of the Loan Documents or the priority of any such liens, nor release or change the liability of any party who may now be or after the date of this Agreement, become liable, primarily or secondarily, under the Loan Documents.
 - 4.3 Representations and Warranties of Borrower.
- (a) <u>Representations and Warranties in Loan Documents</u>. The representations and warranties of Borrower contained in the other Loan Documents, as amended hereby, are true and correct in all material respects as of the date first written above (as if such representations and warranties were made effective as of the date first written above).
- (b) <u>Power to Perform.</u> Borrower has the power, under its organizational documents, to enter into this Agreement and to perform the obligations required to be performed by Borrower under the terms hereunder.

Due Authorization. The execution, delivery and performance by Borrower of this Agreement have been duly authorized by all necessary action on the part of Borrower. This Agreement has been duly executed and delivered by Borrower and, assuming the due execution and delivery of this Agreement by Investor, constitute the legal, valid and binding obligations of Borrower enforceable against Borrower in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium or similar laws or equitable principles affecting the enforcement of creditors' rights generally. No registration with, or consent or approval of, or notice to, or other action by, (i) any trustee or holder of any indebtedness or obligation of Borrower or (ii) any other person for the execution, delivery and performance of this Agreement by Borrower is required or, if required, such registration has been made, such consent, approval or notice given or such other appropriate action taken.

4.4 Miscellaneous.

- (a) Jurisdiction. This Agreement shall be construed according to and governed by the laws of the state of Minnesota.
- (b) <u>Severability: Counterparts.</u> If any provision of this Agreement is adjudicated to be invalid, illegal or enforceable, in whole or in part, it will be deemed omitted to that extent and all other provisions of this Agreement will remain in full force and effect. This Agreement may be executed in one or more counterparts and by the different parties hereto on separate counterparts, each of which, when so executed, shall be deemed to be an original; such counterparts, together, shall constitute one and the same agreement.
- (c) <u>Notices.</u> All notices given pursuant to this Agreement must be in writing and will be effectively given if personally delivered or, if mailed, postage prepaid, certified or registered mail, return receipt requested, to the addresses of Investor and Borrower first set forth above or to such other address as any party subsequently may designate in writing.
- (d) <u>Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, legal representatives, successors and assigns. This Agreement may not be assigned by Borrower without the prior written consent of Investor. This Agreement may be assigned by Investor in connection with an assignment of the Loan without any required notice to Borrower.
- (e) <u>Costs; Fee; Further Assurances.</u> Borrower agrees to pay Investor's out-of-pocket expenses in connection with the preparation of this Agreement and any related expenses, including without limitation, reasonable attorneys' fees. In addition, Borrower agrees to execute such other instruments as may be reasonably required by Investor to evidence or facilitate the agreements set forth herein.

BORROWER SIGNATURE PAGE

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed as of the day and year first above written.

BORROWER:

STRATUS PROPERTIES INC.,

a Delaware corporation

By: <u>/s/ Erin D. Pickens</u>
Name: Erin D. Pickens
Title: Senior Vice President

INVESTOR SIGNATURE PAGE

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first above written.			
	INVESTOR:		
	AMERICAN SELECT PORTFOLIO INC., a Minnesota corporation		
	By: Name: Its:		
9	9		

Certification

I, William H. Armstrong III, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Stratus Properties Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: May 17, 2010

/s/ William H. Armstrong III
William H. Armstrong III
Chairman of the Board,
President & Chief Executive Officer

Certification

I, Erin D. Pickens, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Stratus Properties Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: May 17, 2010

/s/ Erin D. Pickens Erin D. Pickens Senior Vice President & Chief Financial Officer

Certification Pursuant to 18 U.S.C. Section 1350 (Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002)

In connection with the Quarterly Report on Form 10-Q of Stratus Properties Inc. (the "Company") for the quarter ending March 31, 2010, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), William H. Armstrong III, as Chairman of the Board, President & Chief Executive Officer of the Company, hereby certifies, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to the best of his knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: May 17, 2010

/s/ William H. Armstrong III
William H. Armstrong III
Chairman of the Board,
President & Chief Executive Officer

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

This certification shall not be deemed filed by the Company for purposes of § 18 of the Securities Exchange Act of 1934, as amended.

Certification Pursuant to 18 U.S.C. Section 1350 (Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002)

In connection with the Quarterly Report on Form 10-Q of Stratus Properties Inc. (the "Company") for the quarter ending March 31, 2010, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Erin D. Pickens, as Senior Vice President & Chief Financial Officer of the Company, hereby certifies, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that, to the best of her knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Dated: May 17, 2010

/s/ Erin D. Pickens
Erin D. Pickens
Senior Vice President &
Chief Financial Officer

A signed original of this written statement required by Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.

This certification shall not be deemed filed by the Company for purposes of § 18 of the Securities Exchange Act of 1934, as amended.